Youshan Tao

Jianjun Paul Tian

Mathematical Oncology

SPIN Springer's internal project number, if known

Free Boundary Problems of Tumor Growth

February, 2015

Springer

2 Berlin Heidelberg NewYork HongKong London Milan Paris Tokyo To our mentor Professor Avner Friedman

and to our parents and families

Preface

(under revision) This book is devoted to the topics of tumor growth and treatment, chemotactic driven cell movement, and cancer invasion. These topics were the focuses of the authors' research work in past decade. Cancers appear with multiscale features: genes, cells, and biological tissues. From the view point of scales, there are basically three types of mathematical models of cancer: microscopic models (at the molecular and the cellular scales), macroscopic models (at the tissue scale), multiscale model (the cancer model is viewed as a system of subsystems with specific scales). However, this book focuses only on macroscopic models which are generally based on partial differential equations.

Chapters 1-5 deal with the mathematical modeling of tumor growth under various therapies, whereas Chapters 6-8 are concerned with typical mathematical problems arising from cancer biology. To explore tumor dynamics and design possible optimal protocols of treatment, the models developed in chapters 1-5 are confined to a spherical geometry.

VIII Preface

Chapter 1 deals with a model describing the growth of a prostate tumor under hormone therapy. The model considers the mutation by which androgen-dependent tumor cells mutate into androgen-independent ones. Interestingly, explicit formulae of tumor growth in an androgen-deprived environment are found. The relapse of tumor is a crucial problem in hormonal therapy of prostate cancer. The androgen-independent cells are considered to be responsible for such a recurrence. These cells are not sensitive to androgen suppression but rather apt to proliferate even in an androgen-poor environment. Some experimental and clinical studies suggested that intermittent androgen suppression (IAS) may delay or prevent the relapse when compared with continuous androgen suppression. A mathematical model of prostate tumor growth under the IAS therapy is presented in Chapter 2, and the model suggests an optimal protocol of the IAS therapy.

Replication-competent viruses have been used an alternative therapeutic approach for cancer treatment. However, new clinical data revealed an innate immune response to virus may mitigate the effects of treatment. Chapter 3 is concerned with the competitive dynamics between tumor cells, a replication-competent virus and an immune response. It finds an explicit threshold of the intensity of the immune response for controlling a tumor. Chapter 4 deals with a mathematical model of combined therapy which requires not only injection of viruses but also administration of radioiodide. The combination of virotherapy with radiotherapy has recently been experimentally and clinically shown to be significantly more effective than treatment with virotherapy alone. The mathematical model not only verifies the above observation but also can be used to numerically study an optimal timing for radio-iodine administration and an optimal dose for the radioactive iodide.

Glioblastoma multiforme, a type of glioma, is the most aggressive of brain tumors. The standard treatment for newly diagnosed glioblastoma multiforme is surgical resection followed by radiotherapy and chemotherapy. Chapter 5 gives a mathematical model which could predict the survival time of patients who undergo resection, radiation, and chemotherapy with different protocols.

Chapter 6 qualitatively studies a mathematical model describing the cell cycle dynamics and chemotactic driven cell movement in multicellular tumor spheroids. The model is a free boundary problem for a system of partial differential equations with novel free boundary conditions due to different velocities of cells.

Chapter 7 deals with a chemotaxis-haptotaxis model of cancer invasion. The global existence and uniform-in-time boundedness of solutions to the system is studied. This chapter develops some new a priori estimate techniques for chemotaxis-haptotaxis systems. Chapter 8 is concerned with a density-dependent chemotaxis-haptotaxis model of cancer invasion. The equation for cell density includes two bounded nonlinear density-dependent chemotactic and haptotactic sensitivity functions, which exclude the possibility of blow-up of solutions to the model.

X Preface

Our work, and this book, would not have been possible without the guidance from Avner Friedman, who introduced the authors to the research field of cancer models.

We would like to express our gratitude to our research collaborators, particularly Kazuyaki Aihara, Qian Guo, Michael Winkler, ... for their helpful and inspiring cooperation in the past decade. Finally, Tao acknowledges the supports of the National Natural Science Foundation of China (NSFC10571023) and the Natural Science Foundation of Shanghai (09ZR1401200); Tian acknowledges the supports of the National Science Foundation of US (DMS-1446139) and the Natural Science Foundation of China (NNSF-11371048).

Shanghai

Youshan Tao

Las Cruces

Jianjun Paul Tian

February, 2015

Contents

1	Inti	roduction	1
2	Modeling Prostate Tumor Growth under Androgen		
	Sup	pression	3
	2.1	Introduction	3
	2.2	Mathematical Model for Continuous Androgen Suppression	6
	2.3	Transformation of Continuous Androgen Suppression Model	11
	2.4	Formulae of Prostate Tumor Dynamics under Continuous	
		Androgen Suppression	12
	2.5	Mathematical Model for Intermittent Androgen Suppression	21
	2.6	Transformation of Intermittent Androgen Suppression Model	30
	2.7	Numerical Study of Intermittent Androgen Suppression Model	31
	2.8	Summary	35
3	Mo	deling the Competition between Tumor Cells,	
	One	colytic Viruses and Immune Response	37

XII	С	ontents
	3.1	Introduction
	3.2	Mathematical Model 40
	3.3	Transformation 46
	3.4	Threshold of the Immune Response 50
	3.5	A Time-Delay Model 58
	3.6	Glioma Virotherapy 61
		3.6.1 Mathematical model and parameter values
		3.6.2 Comparing with data
		3.6.3 Comparing studies
		3.6.4 Burst size of viruses
		3.6.5 CPA treatments
		3.6.6 The tumor radius
		3.6.7 Conclusion of glioma virotherapy
	3.7	Summary
4	Mo	deling Cancer Radiovirotherapy
	4.1	Introduction
	4.2	Mathematical Model
	4.3	Model transformation
	4.4	Constant Equilibrium Solutions
	4.5	Stability of Equilibrium Solutions
	4.6	Possible Optimal Protocols
	4.7	Summary and Discussion

5	Modeling of Resection, Radiation, and Chemotherapy in			
	Glioblastoma			
	5.1	Introduction		
	5.2	Mathematical Model and Clinical Data		
	5.3	Parameter Estimation		
	5.4	Mathematical Protocols109		
	5.5	Explicit solutions		
	5.6	Summary and Discussion		
6	Tur	nor Modeling with Different Cell Velocities125		
	6.1	Introduction		
	6.2	Mathematical Model		
	6.3	Transformation and Main Results		
	6.4	A Priori Bounds		
	6.5	Main Ideas143		
	6.6	Problem with Nonlinear Boundary Condition145		
	6.7	Local Existence and Uniqueness		
	6.8	Global Existence		
	6.9	Summary		
7	Che	emotaxis-Haptotaxis Modeling for Cancer Invasion $\dots 157$		
	7.1	Introduction		
	7.2	Mathematical Model		
	7.3	Local Existence and Uniqueness		

XIV Contents

	7.4	Global Existence and Boundedness in Dimension 3 171	
	7.5	Global Existence and Boundedness in dimension 2	
8	Der	nsity-Dependent Chemotaxis-Haptotaxis Model of	
	Car	acer Invasion	
	8.1	Introduction	
	8.2	Mathematical Model	
	8.3	Local Existence and Uniqueness	
	8.4	A Priori Estimates and Global Existence	
	8.5	Boundedness for A Volume-Filling Model	
	Refe	erences	
Index			

Introduction

Modeling Prostate Tumor Growth under

Androgen Suppression

2.1 Introduction

The prostate cancer is the secondly most common cancer in males after the lung cancer [53]. More than 670,000 men are diagnosed with prostate cancers every year in the world, and that accounts for one in nine of all new cancers in men. The United States and Sweden have the highest incidence rate of prostate cancers, while China and India have the lowest incidence rate [87]. The chestnut-shaped prostate gland under the bladder is a male sexual organ which produces seminal fluid. The proliferation and apoptosis of prostate cells are regulated by androgens. Prostate cancers are characterized as an abnormal and uncontrolled growth of prostate cancer cells. At present, the cause of prostate cancers is not fully understood, although it was found that many factors such as gene mutations, aging, family history, race, and diet may influence the development of prostate cancers [102, 135].

The growth of prostate tumors depends on androgens. Androgens are secreted by the testicles and adrenal glands. In 1941, Huggin and Hodges 4 2 Modeling Prostate Tumor Growth under Androgen Suppression

[88] pioneered (ADT) for prostate cancers. Since then, ADT has been an important treatment for prostate cancers. Androgen deprivation currently can be achieved by medical castration [83]. ADT stops and rogen production from the testes. (TAB) which further combines anti-androgens with ADT is also widely used. However, both ADT and TAB seldom succeed in removing all prostate tumor cells, and relapses of prostate tumors often occur. The so-called androgen-independent (AI) cells are considered to be responsible for tumor relapses. These cells are not sensitive to androgen suppression but rather apt to proliferate even in an androgen-poor environment [16, 90]. Since androgen dependent (AD) cells cannot proliferation under the androgen deprivation condition, a prostate tumor relapse would imply some increase of AI cells. Several mechanisms have been identified for progression of AD cells to AI cells. These include and rogen receptor (AR) gene amplification, AR mutation, and bypass of androgenic activation of AR or of AR signaling itself [83, 102, 113]. Hence, the continuous and rogen suppression (CAS) therapy with ADT and TAB often results in an AI relapse.

To our knowledge, Jackson [95, 96] developed the first partial differential equation model for the CAS therapy of prostate cancers. The model considers a prostate tumor as a heterogeneous mixture of AD and AI cells, and it is assumed that a relapse of a prostate tumor can possibly result from decreasing the apoptotic rate of the AI cells by androgen-deprivation. Jackson's model well agrees with experimental observations. This model predicts that the CAS therapy is successful only for some small ranges of biological parameters. The model also suggests that an androgen-independent relapse is associated with decreasing of apoptosis while without increasing of proliferation [95, 96].

In this chapter, we will first extend Jackson's model. Our model will incorporate mutation inhibitors [154]. By the mutation we mean that androgen-dependent (AD) prostate tumor cells mutate into androgen-independent (AI) prostate tumor cells, and by mutation inhibitors we mean inhibition effects to the mutation rates. As afore-mentioned, the relapse of tumors is a crucial problem in hormonal therapy of prostate cancers. Currently we know that there are two possible mechanisms that a prostate tumor can recur [91]. One is mutation or adaptation where AD cells mutate into AI cells under the androgen deprived condition. The other is selection or competition where AI cells are minor but exist from the beginning of the therapy and they will dominate under androgen suppression condition. These two recurrence mechanisms have *different* effects on the prostate tumor relapse, and they may work together. However, we will first consider the mutation mechanism of prostate tumor relapse under continuous androgen suppression.

Qualitative analysis suggests that a tumor relapse cannot be avoided under androgen-deprived therapy. This implication may support a possible strategy of intermittent androgen suppression (IAS). Intermittent androgen suppression is a type of androgen ablative therapy delivered intermittently with off-treatment periods [17]. Actually, there are several mathematical models that study prostate tumor growth under IAS therapy. We will also 6 2 Modeling Prostate Tumor Growth under Androgen Suppression review a mathematical model that considers the mutation mechanism of prostate tumor relapse under intermittent androgen suppression.

2.2 Mathematical Model for Continuous Androgen

Suppression

Following [95, 96], the prostate tumor is viewed as a densely packed and radially-symmetric sphere. We denote R(t) as the radius of the tumor sphere. The tumor contains both AD and AI cells, and their number densities are denoted by $x_1(r,t)$ and $x_2(r,t)$, respectively. Since the level of androgens within the tissue under consideration can be regulated by medical methods [17, 18, 95, 96], we neglect spatial heterogeneity of androgens within the prostate tumor tissue. The tumor is assumed to be incompressible fluids with a velocity field \mathbf{v} ($v := \mathbf{v} \cdot \mathbf{r}/|\mathbf{r}|$). The velocity field is caused by cell proliferation and death. The cell proliferation depends on the androgen level, a(t). In solid tumor growth modeling, it is usually assumed that cell movement has two components: 1) motion due to the velocity $\mathbf{v}(\mathbf{r}, t)$ [93, 94, 168], and 2) random motion [20, 66, 73, 74]. We exploit the spherical symmetry of the problem by assuming henceforth that the variables x_1, x_2 , and **v** depend only on (r, t)where r is the radial distance from the center of the tumor, and t is time. As an extension of the model [95, 96], the model we consider consists of the following equations:

2.2 Mathematical Model for Continuous Androgen Suppression

 $\overline{7}$

$$\frac{\partial x_1}{\partial t}(r,t) + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 v(r,t) x_1(r,t) \right)$$

$$= \frac{D_1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial x_1}{\partial r}(r,t) \right)$$

$$+ p_1(a(t)) x_1(r,t) - q_1(a(t)) x_1(r,t)$$

$$- (1-I)m(a(t)) x_1(r,t), \qquad (2.1)$$

$$\frac{\partial x_2}{\partial t}(r,t) + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 v(r,t) x_2(r,t) \right)$$

$$= \frac{D_2}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial x_2}{\partial r}(r,t) \right) + (1-I)m(a(t)) x_1(r,t)$$

$$+p_2(a(t))x_2(r,t) - q_2(a(t))x_2(r,t), \qquad (2.2)$$

where D_1 and D_2 are the random motility coefficients of the AD and AI cells; $p_1(a(t))$ and $p_2(a(t))$ are proliferation rates of the AD and AI cells; $q_1(a(t))$, and $q_2(a(t))$ are their apoptosis rates; m(a(t)) is the mutation rate by which AD cells mutate to AI ones. The intensity of the inhibitors that reduces the mutation rate is represented by parameter I, which varies from zero to one. I = 0 corresponds to no inhibition to the mutation, while I = 1 corresponds to the full inhibition to the mutation. So, the rate at which cells mutate from the AD type to AI type is given by (1 - I)m(a(t)). The cell proliferation rate, apoptotic rate, and mutation rate are assumed to be dependent on the local concentration of androgens, a(t).

It is a fact that within a tumor the total number of cells per unit volume is constant [95, 96, 168]. Hence,

$$x_1 + x_2 = k \equiv \text{constant.} \tag{2.3}$$

Equations (2.1) and (2.2) with assumption (2.3) yield

2 Modeling Prostate Tumor Growth under Androgen Suppression

8

$$\frac{k}{r^2} \frac{\partial}{\partial r} \left(r^2 v(r,t) \right) = (D_1 - D_2) \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial x_1}{\partial r}(r,t) \right) + p_1(a(t)) x_1(r,t) + p_2(a(t))(k - x_1(r,t)) - q_1(a(t)) x_1(r,t) - q_2(a(t))(k - x_1(r,t)).$$
(2.4)

By the radial symmetry assumption of the problem, we have

$$\frac{\partial x_1}{\partial r}(0,t) = \frac{\partial x_2}{\partial r}(0,t) = v(0,t) = 0.$$
(2.5)

To close the system of the equations, we need to impose boundary and initial conditions.

Boundary Conditions: We assume that there is no-flux of cancer cells across the outer boundary of the tumor. This assumption gives the following equations,

$$\begin{cases} \left[x_1(r,t)\frac{dR(t)}{dt} - \left(x_1(r,t)v(r,t) - D_1\frac{\partial x_1}{\partial r}(r,t)\right)\right]_{r=R(t)} = 0, \\ \left[x_2(r,t)\frac{dR(t)}{dt} - \left(x_2(r,t)v(r,t) - D_2\frac{\partial x_2}{\partial r}(r,t)\right)\right]_{r=R(t)} = 0, \end{cases}$$
(2.6)

which, together with (2.3), yield the free boundary conditions:

$$\begin{cases} \frac{dR(t)}{dt} - v(R(t), t) = 0, \\ \frac{\partial x_1}{\partial r}(R(t), t) = 0. \end{cases}$$
(2.7)

Initial Conditions: We prescribe the initial data as follows,

$$R(0) = R_0, \quad x_1(r,0) = x_{10}(r) \quad \text{for } 0 \le r \le R_0.$$
(2.8)

Remark 2.1. The no-flux boundary condition (2.6) is obtained by considering the relative velocities of cells on the outer boundary of the growing tumor. These types of no-flux boundary conditions for diffusion-advection equations in a moving domain $\{r \leq R(t)\}$ were firstly clarified by Tao [150].

9

Since $x_1 + x_2 = k$, it is not necessary to impose any additional initial condition for x_2 in Eq. (2.8). We also note that Eq. (1.2) is a consequence of Eqs. (2.1), (2.3) and (2.4), so in what follows we simply drop this equation.

After adopting the non-dimensional variables and some parameter values given in [95], the model (2.1)-(2.8) can be rewritten as follows:

$$\frac{\partial x_1}{\partial t}(r,t) + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 v(r,t) x_1(r,t) \right)$$

$$= \frac{\epsilon_1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial x_1}{\partial r}(r,t) \right) + p_1(a(t)) x_1(r,t) - q_1(a(t)) x_1(r,t)$$

$$- (1-I)m(a(t)) x_1(r,t), \qquad (2.9)$$

$$\frac{1}{r^2} \frac{\partial r}{\partial r} \left(r \, v(r,t) x_1(r,t) \right)$$

= $p_1(a(t)) x_1(r,t) + 1 - x_1(r,t) - q_1(a(t)) x_1(r,t)$

$$-q_2(a(t))(1-x_1(r,t)), (2.10)$$

$$\frac{dR}{dt} = v(R(t), t), \tag{2.11}$$

$$R(0) = 1, \quad x_1(r,0) = x_{10}(r),$$
 (2.12)

$$\frac{\partial x_1}{\partial r}(0,t) = 0, \quad v(0,t) = 0, \tag{2.13}$$

$$\frac{\partial x_1}{\partial r}(R(t),t) = 0, \qquad (2.14)$$

where $0 < \varepsilon_1 \ll 1$ is some constant. The functions a(t), $p_1(a(t))$, $q_1(a(t))$, $q_2(a(t))$, and m(a(t)) take the following specific forms (see [95] for details):

10 2 Modeling Prostate Tumor Growth under Androgen Suppression

$$a(t) = e^{-bt} + a_s, \quad t \ge 0,$$
 (2.15)

$$p_1(a(t)) = \theta_1 + (1 - \theta_1) \frac{a(t)}{a(t) + K},$$
(2.16)

$$q_1(a(t)) = \delta_1 \Big[\omega_1 + (1 - \omega_1) \frac{a(t)}{a(t) + K} \Big], \tag{2.17}$$

$$q_2(a(t)) = \delta_2 \Big[\omega_2 + (1 - \omega_2) \frac{a(t)}{a(t) + K} \Big],$$
(2.18)

$$m(a(t)) = m_1 \left(1 - \frac{a(t)}{1 + a_s} \right), \tag{2.19}$$

where $b, K, m_1, \delta_1, \delta_2, \omega_1$, and ω_2 are some positive constants with the following assumed conditions:

$$0 \le a_s < 1, \quad 0 \le \theta_1 < 1, \quad \delta_1 < \delta_2, \quad \omega_1 > 1, \quad \text{and} \quad \omega_2 < 1.$$
 (2.20)

In Eq. (2.15) we assume that the hormonal treatment is initiated at the time t = 0. The parameter $a_s > 0$ corresponds to ADT, while $a_s = 0$ corresponds to TAB. The parameter θ_1 represents the proliferation rate of AD cells in the androgen-deprived state. $0 \le \theta_1 < 1$ means that the deprivation of androgens will decrease the proliferation rate of the AD cells. The assumption of $\delta_1 < \delta_2$ is due to the fact that the AD cells are dominant in androgen-rich conditions [52, 95, 96]. The assumption of $\omega_1 > 1$ and $\omega_2 < 1$ mean that the deprivation of androgens will increase the apoptosis rate of the AD cells but reduce that of AI cells [95]. Equation (2.19) assumes that the mutation rate m(a(t)) is decreasing with increasing the local concentration of androgens a(t).

2.3 Transformation of Continuous Androgen Suppression Model

It is difficult to study free boundary problems, in general. We usually choose various transformations to change free boundary problems to fixed boundary problems. For the model we introduced above, we adopt, as shown in [145], the change of variables $(r, t, x_1, v, R) \mapsto (\rho, t, \tilde{x}_1, \tilde{v}, R)$ as follows:

$$\rho = r/R(t), \quad t = t, \quad R(t) = R(t),$$
$$\tilde{x}_1(\rho, t) = x_1(\rho R(t), t), \quad \tilde{v}(\rho, t) = v(\rho R(t), t)/R(t).$$
(2.21)

In terms of the new variables and after dropping the tildes of $\tilde{x}_1(\rho, t)$ and $\tilde{v}(\rho, t)$ for notational convenience, the system (2.9)-(2.14) takes the following form in $\{0 < \rho < 1, t > 0\}$:

$$\frac{\partial x_1}{\partial t}(\rho,t) + \left[v(\rho,t) - \rho v(1,t)\right] \frac{\partial x_1}{\partial \rho}(\rho,t) - \frac{\varepsilon_1}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial x_1}{\partial \rho}(\rho,t)\right) \\
= \left[p_1(a(t)) + q_2(a(t)) - 1 - q_1(a(t))\right] x_1(\rho,t) (1 - x_1(\rho,t)) \\
- (1 - I)m(a(t)) x_1(\rho,t),$$
(2.22)

$$x_1(\rho, 0) = x_{10}(\rho), \tag{2.23}$$

$$\frac{\partial x_1}{\partial \rho}(0,t) = \frac{\partial x_1}{\partial \rho}(1,t) = 0, \qquad (2.24)$$

$$v(\rho,t) = \frac{1}{\rho^2} \int_0^{\rho} \left[p_1(a(t)) x_1(s,t) + 1 - x_1(s,t) - q_1(a(t)) x_1(s,t) - q_2(a(t)) (1 - x_1(s,t)) \right] s^2 ds,$$
(2.25)

$$\frac{dR(t)}{dt} = R(t)v(1,t),$$
(2.26)

$$R(0) = 1, (2.27)$$

12 2 Modeling Prostate Tumor Growth under Androgen Suppression where we have used the fact that v(0,t) = 0 in deriving Eq. (2.25).

The global existence and uniqueness of solutions to the system (2.22)-(2.27) were prove by Tao et al. [154]. We do not give the results and proofs here, instead, in next section, we will review the dynamic behavior of tumor growth in the androgen-deprived environment.

2.4 Formulae of Prostate Tumor Dynamics under

Continuous Androgen Suppression

For the general case where $a = a(t) \neq 0$ given in (2.15), the dynamics of prostate tumor growth can be numerically studied (see Tao et al. [154] for details). In this section, we shall focus on the dynamical behavior of prostate tumor growth in androgen-deprived environment. That is, we consider the case:

$$a = 0, \tag{2.28}$$

and therefore

$$p_1(a) = \theta_1, \ q_1(a) = \delta_1 \omega_1, \ q_2(a) = \delta_2 \omega_2, \ \text{and} \ m(a) = \beta_1.$$
 (2.29)

Under the assumption (2.28), Eqs. (2.22)-(2.27) can be rewritten as follows:

2.4 Formulae of Prostate Tumor Dynamics under Continuous Androgen Suppression

$$\frac{\partial x_1}{\partial t}(\rho,t) + \left[v(\rho,t) - \rho v(1,t)\right] \frac{\partial x_1}{\partial \rho}(\rho,t) - \frac{\varepsilon_1}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial x_1}{\partial \rho}(\rho,t)\right)$$

$$= \left(\theta_1 + \delta_2 \omega_2 - 1 - \delta_1 \omega_1\right) x_1(\rho,t) (1 - x_1(\rho,t))$$

$$- (1 - I)m_1 x_1(\rho,t),$$
(2.30)

$$x_1(\rho, 0) = x_{10}(\rho), \tag{2.31}$$

$$\frac{\partial x_1}{\partial \rho}(0,t) = \frac{\partial x_1}{\partial \rho}(1,t) = 0, \qquad (2.32)$$

$$v(\rho,t) = \frac{1}{\rho^2} \int_0^{\rho} \left[\theta_1 x_1(s,t) + 1 - x_1(s,t) - \delta_1 \omega_1 x_1(s,t) - \delta_2 \omega_2 (1 - x_1(s,t)) \right] s^2 ds,$$
(2.33)

$$\frac{dR(t)}{dt} = R(t)v(1,t),$$
(2.34)

$$R(0) = 1. (2.35)$$

Set $x_2(\rho, t) := 1 - x_1(\rho, t)$, and define

$$V_1(t) := 4\pi R^3(t) \int_0^1 x_1(\rho, t) \rho^2 d\rho, \quad V_2(t) := 4\pi R^3(t) \int_0^1 x_2(\rho, t) \rho^2 d\rho,$$

where $V_1(t)$ and $V_2(t)$ are the volumes occupied by AD cells and AI cells at time t, respectively. Then

$$V(t) := V_1(t) + V_2(t) \equiv \frac{4}{3}\pi R^3(t)$$
(2.36)

is the tumor volume at time t. In this section we will derive formulae of the tumor volume (or the tumor radius) at time t. The following results were obtained by Tao et al. [154].

Theorem 2.1. Under various conditions about combined parameters $\theta_1 - \delta_1 \omega_1 - (1-I)m_1$ and $1 - \delta_2 \omega_2$, the formulae for prostate tumor volume growth

13

14 2 Modeling Prostate Tumor Growth under Androgen Suppression over time can be explicitly expressed. Specifically, the tumor volume growth formulae are given under the following 5 different conditions.

(1). If the condition 1, $\theta_1 - \delta_1 \omega_1 - (1 - I)m_1 = 0$ and $1 - \delta_2 \omega_2 = 0$, holds, then we have

$$V(t) = \frac{4\pi}{3} + V_1(0)(1-I)m_1t.$$
(2.37)

(2). If the condition 2, $\theta_1 - \delta_1 \omega_1 - (1 - I)m_1 = 0$ and $1 - \delta_2 \omega_2 \neq 0$, holds, then we have

$$V(t) = V_1(0) + V_2(0)e^{(1-\delta_2\omega_2)t} + \frac{(1-I)m_1V_1(0)}{1-\delta_2\omega_2} \Big[e^{(1-\delta_2\omega_2)t} - 1\Big].$$
(2.38)

(3). If the condition 3, $\theta_1 - \delta_1 \omega_1 - (1 - I)m_1 \neq 0$ and $1 - \delta_2 \omega_2 = 0$, holds,

then we have

$$V(t) = V_2(0) + V_1(0)e^{[\theta_1 - \delta_1 \omega_1 - (1-I)m_1]t} + \frac{(1-I)m_1V_1(0)}{\theta_1 - \delta_1 \omega_1 - (1-I)m_1} \Big\{ e^{[\theta_1 - \delta_1 \omega_1 - (1-I)m_1]t} - 1 \Big\}.$$
 (2.39)

(4). If the condition 4, $\theta_1 - \delta_1 \omega_1 - (1 - I)m_1 \neq 0$, $1 - \delta_2 \omega_2 \neq 0$, and $\theta_1 - \delta_2 \omega_2 \neq 0$.

 $\delta_1\omega_1 - (1-I)m_1 = 1 - \delta_2\omega_2$, holds, then we have

$$V(t) = \left[V_2(0) + (1-I)m_1V_1(0)t\right]e^{(1-\delta_2\omega_2)t} + V_1(0)e^{[\theta_1 - \delta_1\omega_1 - (1-I)m_1]t}.$$
(2.40)

(5). If the condition 5, $\theta_1 - \delta_1 \omega_1 - (1 - I)m_1 \neq 0$, $1 - \delta_2 \omega_2 \neq 0$, and $\theta_1 - \delta_2 \omega_2 \neq 0$.

 $\delta_1\omega_1 - (1-I)m_1 \neq 1 - \delta_2\omega_2$, holds, then we have

$$V(t) = V_1(0)e^{[\theta_1 - \delta_1\omega_1 - (1-I)m_1]t} + V_2(0)e^{(1-\delta_2\omega_2)t}$$

$$+ \frac{(1-I)m_1V_1(0)e^{(1-\delta_2\omega_2)t}}{\theta_1 - \delta_1\omega_1 - (1-I)m_1 - 1 + \delta_2\omega_2} \Big\{ e^{[\theta_1 - \delta_1\omega_1 - (1-I)m_1 - 1 + \delta_2\omega_2]t} - 1 \Big\}.$$
(2.41)

$$\begin{split} \frac{1}{4\pi} \dot{V}_{1}(t) &= 3R^{2}(t)\dot{R}(t) \int_{0}^{1} x_{1}\rho^{2}d\rho + R^{3}(t) \int_{0}^{1} \frac{\partial x_{1}}{\partial t}\rho^{2}d\rho \\ &= 3R^{3}(t)v(1,t) \int_{0}^{1} x_{1}\rho^{2}d\rho + R^{3}(t) \int_{0}^{1} \frac{\partial x_{1}}{\partial t}\rho^{2}d\rho \\ &= 3R^{3}(t)v(1,t) \int_{0}^{1} x_{1}\rho^{2}d\rho \\ &+ R^{3}(t) \int_{0}^{1} \left[(\theta_{1} + \delta_{2}\omega_{2} - 1 - \delta_{1}\omega_{1})x_{1}(1 - x_{1}) - (1 - I)m_{1}x_{1} \right]\rho^{2}d\rho \\ &+ \varepsilon_{1}R(t) \int_{0}^{1} \frac{\partial}{\partial\rho} \left(\rho^{2}\frac{\partial x_{1}}{\partial\rho}\right)d\rho \\ &- R^{3}(t) \int_{0}^{1} \left[\rho^{2}v(\rho, t) - \rho^{3}v(1, t) \right] \frac{\partial x_{1}}{\partial\rho}d\rho \\ &= 3R^{3}(t)v(1, t) \int_{0}^{1} x_{1}\rho^{2}d\rho \\ &+ R^{3}(t) \int_{0}^{1} \left[(\theta_{1} + \delta_{2}\omega_{2} - 1 - \delta_{1}\omega_{1})x_{1}(1 - x_{1}) - (1 - I)m_{1}x_{1} \right]\rho^{2}d\rho \\ &+ \varepsilon_{1}R(t) \left(\rho^{2}\frac{\partial x_{1}}{\partial\rho}\right) \Big|_{\rho=0}^{1} \\ &- R^{3}(t) \left\{ \left[\rho^{2}v(\rho, t) - \rho^{3}v(1, t) \right]x_{1}(\rho, t) \right\} \Big|_{\sigma=0}^{1} \end{split}$$

$$\begin{aligned} &-R^{3}(t)\left\{\left[\rho^{2}v(\rho,t)-\rho^{9}v(1,t)\right]x_{1}(\rho,t)\right\}\right|_{\rho=0} \\ &+R^{3}(t)\int_{0}^{1}x_{1}\frac{\partial}{\partial\rho}\left[\rho^{2}v(\rho,t)\right]d\rho-3R^{3}(t)v(1,t)\int_{0}^{1}x_{1}\rho^{2}d\rho \\ &=R^{3}(t)\int_{0}^{1}\left[(\theta_{1}+\delta_{2}\omega_{2}-1-\delta_{1}\omega_{1})x_{1}(1-x_{1})-(1-I)m_{1}x_{1}\right]\rho^{2}d\rho \\ &+R^{3}(t)\int_{0}^{1}x_{1}\left[\theta_{1}x_{1}+1-x_{1}-\delta_{1}\omega_{1}x_{1}-\delta_{2}\omega_{2}(1-x_{1})\right]\rho^{2}d\rho \\ &=R^{3}(t)\int_{0}^{1}\left[\theta_{1}-\delta_{1}\omega_{1}-(1-I)m_{1}\right]x_{1}\rho^{2}d\rho \\ &=\frac{1}{4\pi}\left[\theta_{1}-\delta_{1}\omega_{1}-(1-I)m_{1}\right]V_{1}(t). \end{aligned}$$

We obtain an ODE for $V_1(t)$,

$$\dot{V}_1(t) = [\theta_1 - \delta_1 \omega_1 - (1 - I)m_1]V_1(t).$$
(2.42)

15

16 2 Modeling Prostate Tumor Growth under Androgen Suppression

This equation further yields

$$V_1(t) = V_1(0)e^{[\theta_1 - \delta_1 \omega_1 - (1-I)m_1]t}.$$
(2.43)

Similarly, we combine Eqs. (2.36), (2.33), (2.34), and (2.42), and calculate

$$\begin{split} \dot{V}_2(t) &= 4\pi R^2(t)\dot{R}(t) - \dot{V}_1(t) \\ &= 4\pi R^3(t)V(1,t) - [\theta_1 - \delta_1\omega_1 - (1-I)m_1]V_1(t) \\ &= 4\pi R^3(t)\int_0^1 \left[\theta_1 x_1 + 1 - x_1 - \delta_1\omega_1 x_1 - \delta_2\omega_2(1-x_1)\right]\rho^2 d\rho \\ &- 4\pi R^3(t)\int_0^1 \left[\theta_1 - \delta_1\omega_1 - (1-I)m_1\right]x_1\rho^2 d\rho \\ &= 4\pi R^3(t)\int_0^1 (1 - \delta_2\omega_2)x_2\rho^2 d\rho \\ &+ 4\pi R^3(t)\int_0^1 (1-I)m_1x_1\rho^2 d\rho \\ &= (1 - \delta_2\omega_2)V_2(t) + (1 - I)m_1V_1(t). \end{split}$$

This gives us an ODE equation for $V_2(t)$:

$$\dot{V}_2(t) = (1 - \delta_2 \omega_2) V_2(t) + (1 - I) m_1 V_1(t).$$
(2.44)

Now combining the initial condition (2.35), the tumor volume (2.36), and the expressions (2.43) and (2.44), we easily obtain (2.37)-(2.41).

Remark 2.2. A necessary condition for successful treatments of prostate tumors was derived in [95]. Our Theorem 2.1 gives the *explicit formulae* of the tumor volume growth over time t in androgen-deprived environment. Therefore, the dynamics of the tumor growth in androgen-deprived environment can be predicted by these formulae. Although the formulae were derived under 2.4 Formulae of Prostate Tumor Dynamics under Continuous Androgen Suppression the assumption of radial symmetry, it may be useful to predict the long-term behavior of tumor growth qualitatively.

From Theorem 2.1, we can derive some interesting analytical results about androgen deprivation therapy. To demonstrate, we look at several situations in the following.

Case 2.1. In Jackson's article [95], parameter values are given as

$$\delta_1 = \frac{0.3812}{0.4621}, \ \delta_2 = \frac{0.4765}{0.4621}, \ \theta_1 = 0.8, \ \omega_1 = 1.35, \ \omega_2 = 0.25.$$

For this set of typical paprameter values, we can easily verify that

$$\theta_1 - \delta_1 \omega_1 < 0 \quad \text{and} \quad 1 - \delta_2 \omega_2 > 0.$$
 (2.45)

That means that the "*net*" growth rate of AD cells is negative whereas the "*net*" growth rate of AI cells is positive in androgen-deprived environment. It follows from (2.45) that

$$\theta_1 - \delta_1 \omega_1 - 1 + \delta_2 \omega_2 < 0. \tag{2.46}$$

This is a general case in Theorem 2.1. To explain biological significance, we now rewrite the formula (2.41) as follows:

$$V(t) = \left(V_1(0) + \frac{(1-I)m_1V_1(0)}{\theta_1 - \delta_1\omega_1 - (1-I)m_1 - 1 + \delta_2\omega_2}\right)e^{[\theta_1 - \delta_1\omega_1 - (1-I)m_1]t} \\ + \left(V_2(0) - \frac{(1-I)m_1V_1(0)}{\theta_1 - \delta_1\omega_1 - (1-I)m_1 - 1 + \delta_2\omega_2}\right)e^{(1-\delta_2\omega_2)t} \\ := A(t) + B(t).$$
(2.47)

If we further assume that

17

18 2 Modeling Prostate Tumor Growth under Androgen Suppression

$$V_2(0) > 0,$$

namely, the initial volume occupied by AI cells is non-zero, then from (2.45)and (2.46) we have

$$A(t) \to 0$$
 and $B(t) \to +\infty$, as $t \to +\infty$.

Therefore, from (2.47) we obtain

$$V(t) \to +\infty, \quad \text{as} \quad t \to +\infty.$$
 (2.48)

This result implies a prostate tumor relapse under androgen-deprived therapy. We should be aware of that (2.48) holds in a sense of mathematics, and it may not be biologically plausible. In fact, the host will die when the prostate tumor is large enough. However, (2.48) predicts eventual failure of the androgen-deprived therapy.

Case 2.2. Because

$$A(t) \to 0$$
 and $B(t) \to +\infty$, as $t \to +\infty$,

the term B(t) will dominate the growth of a prostate tumor. We also notice that the factor of B(t),

$$-\frac{(1-I)m_1}{\theta_1 - \delta_1\omega_1 - (1-I)m_1 - 1 + \delta_2\omega_2} \equiv 1 - \frac{(\delta_1\omega_1 - \theta_1) + (1 - \delta_2\omega_2)}{(1-I)m_1 + (\delta_1\omega_1 - \theta_1) + (1 - \delta_2\omega_2)},$$

which is *decreasing* with increasing of $I \in [0, 1]$ under conditions (2.45). This fact suggests that controlling the mutation intensity may delay the relapse of prostate tumors under the androgen-deprived therapy. 2.4 Formulae of Prostate Tumor Dynamics under Continuous Androgen Suppression 19 Case 2.3. We now explain some biological significance of combined parameters $\theta_1 - \delta_1 \omega_1 - (1 - I)m_1$ and $1 - \delta_2 \omega_2$ in Theorem 2.1. Let's denote

the "net" growth rate of AD cells

= the proliferation rate of AD cells - the death rate of AD cells - the mutation rate

$$=\theta_1-\delta_1\omega_1-(1-I)m_1,$$

the "natural net" growth rate of AI cells

= the proliferation rate of AI cells - the death rate of AI cells

 $= 1 - \delta_2 \omega_2,$

the "net" growth rate of AI cells

= the mutation rate + the "natural net" growth rate of AI cells

$$= (1 - I)m_1 + (1 - \delta_2 \omega_2),$$

where the mutation rate $(1 - I)m_1$ is defined for AD cells to mutate to AI cells.

The condition 1 means that the "net" growth rate of AD cells and the "natural net" growth rate of AI cells are both equal to zero. Theorem 2.1 says that under the condition 1 the prostate tumor relapse can not be avoided due to the mutation of AD cells which results in increasing of AI cells as shown in (2.37). However, (2.37) also suggests that controlling the mutation (i.e. increasing the value of I) may delay (for 0 < I < 1) or prevent (for I = 1) the relapse. In fact, from (2.37), as $t \to +\infty$, we see

$$V(t) \rightarrow +\infty$$
 if $0 < I < 1$, but $V(t) \equiv \frac{4}{3}\pi$ if $I = 1$.

20 2 Modeling Prostate Tumor Growth under Androgen Suppression

The condition 2 means that the "*net*" growth rate of AD cells is equal to zero and the "*natural net*" growth rate of AI cells is positive (we assume that $1 - \delta_2 \omega_2 > 0$ for the typical parameter values as given in [95]). Theorem 2.1 says that under the condition 2 the AI cells will dominate the prostate tumor growth as shown in (2.38) and the tumor relapse can not be avoided. In fact, from (2.38) and $1 - \delta_2 \omega_2 > 0$, we see that

$$V(t) \to +\infty$$
 as $t \to +\infty$.

The condition 3 means that the "net" growth rate of AD cells is non-zero (we assume that it is negative for typical parameter values as given in [95]) and the "natural net" growth rate of AI cells is zero. Theorem 2.1 says that under the condition 3 the prostate tumor growth could be controlled as shown in (2.39). In fact, from $\theta_1 - \delta_1 \omega_1 - (1 - I)m_1 < 0$ and (2.39), we have

$$V(t) \to V_2(0) + \frac{(1-I)m_1V_1(0)}{(1-I)m_1 - (\theta_1 - \delta_1\omega_1)}$$
 as $t \to +\infty$.

The condition 4 means that the "*net*" growth rate of AD cells is equal to the "*natural net*" growth rate of AI cells and they are positive. Theorem 2.1 states that under the condition 4 both AD cells and AI cells contribute to the prostate tumor growth as shown in (2.40) and the tumor relapse can not be avoided. In fact, from (2.40) and $\theta_1 - \delta_1 \omega_1 - (1 - I)m_1 = 1 - \delta_2 \omega_2 > 0$, we see

$$V(t) \to +\infty$$
 as $t \to +\infty$.

The condition 5 means that the "*net*" growth rate of AD cells and the "*natural net*" growth rate of AI cells are both non-zero and they are not equal.

2.5 Mathematical Model for Intermittent Androgen Suppression 21 For typical parameter values given in [95], we assume that the "*net*" growth rate of AD cells is negative and the "*natural net*" growth rate of AI cells latter is positive (as in (2.45)). Under these assumptions the AI cells will dominate the prostate tumor growth as shown in (2.41) and the tumor relapse can not be avoided as shown in (2.48).

2.5 Mathematical Model for Intermittent Androgen Suppression

The prostate gland is a male sexual organ which produces and secretes seminal fluid. Activities of prostate cells, such as proliferation and apoptosis, are regulated by androgens. Causes of prostate cancer are not fully understood although genes, aging, race, family history, and lifestyle-related factors are regarded as influential factors [135]. The screening, detection, and staging of prostate cancer is currently conducted by using the serum prostate-specific antigen (PSA) test [140].

A prostate tumor as well as the prostate gland itself is influenced by androgens. Androgens are secreted by the testicles and the adrenal glands. Androgens circulate in the blood, diffuse into the prostate tissue, and stimulate the prostate tumor to grow. The androgen suppression therapy for prostate cancers was initially proposed by Huggins and Hodges in 1941 [88]. Androgen suppression can now be realized easily by chemical castration [90]. However, the relapse of a prostate tumor remains a crucial problem for androgen 22 2 Modeling Prostate Tumor Growth under Androgen Suppression suppression therapy. The growth of androgen-independent (AI) cells is considered as one of the causes for this relapse. These AI cells are not sensitive to androgen suppression. As a result AI cells can still grow in androgen-deprived environments where androgen-dependent (AD) cells cannot proliferate [16, 17, 18, 90]. Thus continuous androgen suppression (CAS) therapy often results in a tumor relapse due to an emergence of such AI cancer cells, which has already been studied in previous sections.

Recent clinical studies [12, 16, 17, 18] suggest that intermittent androgen suppression (IAS) therapy may prolong or possibly prevent the relapse. In IAS therapies, androgen suppression is stopped when the monitored serum PSA concentration decreases to less than a lower threshold, while androgen suppression is resumed when the PSA concentration exceeds an upper threshold. The clinical results of IAS therapies are reviewed in the article [12]. However, a very important question is how to administer IAS therapy in order to have a maximum efficacy because IAS is dynamical therapy. To answer this question, experimental results about the prostate tumor relapse combining mathematical models for IAS therapy is needed.

There are intensive experimental studies on prostate tumor recurrence [83, 98, 101, 102, 113]. Two possible mechanisms of prostate tumor recurrence [91] are recognized currently. One is mutation or adaptation, where AI cells may emerge from androgen-dependent (AD) cells under the androgen deprived condition. The other is selection or competition, where AI cells are minor but exist from the beginning of the therapy and will be winners of competition 2.5 Mathematical Model for Intermittent Androgen Suppression 23 between AD cells and AI cells under the hormonal therapy. Since these two mechanisms may work together and the real situation can be much more complex due to polyclonality of AD cells and AI cells [52, 111, 132], different mathematical models are needed for these mechanisms.

Ideta et al. [89] proposed an ordinary differential equation (ODE) model for prostate cancer IAS therapy, which studies the *mutation mechanism* of prostate tumor relapse. This model simplifies biological reality since spatial heterogeneity is neglected. Incorporating spatial motion of cells, Guo et al. [76] extended this ODE model to a partial differential equation (PDE) model. The main difference between the ODE model [89] and the PDE model [76] is that the subpopulation of AI (or AD) cells in the PDE model is *nonlinearly* related to the subpopulations of AD (or AI) cells due to the spatial movement of tumor cells, while the subpopulations of AD (or AI) cells. Furthermore, the numerical study of the PDE model finds an optimal lower threshold r_0 (when the upper threshold r_1 is fixed), which suggests an optimal protocol for IAS therapy.

As mentioned, mutation and competition are two different mechanisms of the relapse of a prostate tumor. Shimada and Aihara [134] proposed another ODE model for IAS therapy, which studies the *competition mechanism* of a tumor relapse. Tao et al. [152] extended this ODE model to a PDE model by incorporating the spatial motion of cells within the model.

24 2 Modeling Prostate Tumor Growth under Androgen Suppression

Let's assume that $\Omega \in \mathcal{R}^3$ is the domain of the solid prostate tumor under consideration. To make the model tractable, Ω is assumed to be spherical. Following [95, 96], the tumor is considered as a densely packed, radially-symmetric sphere of radius R(t). We assume that tumor cells move as a result of a convective velocity field \mathbf{v} caused by cell proliferation and death driving local volume changes [95, 168]. We also assume that cells move through random processes such as diffusion processes [73, 74]. Since the growth of a prostate tumor is mainly dependent on androgens [12, 95], we also assume that the growth of a solid prostate tumor is stimulated by androgens (here we neglect other nutrients or inhibitors). Androgen deprivation now can be easily performed by chemical castration [12, 90], for instance, by administration of pharmacological agents such as luteinizing hormone releasing hormone (LHRH) analogs that inhibit the production of androgens from its primary source, the testes. The remaining androgens that are produced by the adrenal glands can also be eliminated by additional treatment with androgen-receptor antagonists (anti-androgens). Thus, the level of androgens within the tissue under consideration can be controlled by medical means, and we can neglect spatial heterogeneity of androgens within the tissue. As mentioned before, prostate cancer cells are divided into two groups: and rogen-dependent (AD) cells and and rogen-independent (AI) cells. AI cells are not sensitive to androgen suppression but rather apt to increase even in an androgen deprivation condition in which AD cells cannot proliferate. Therefore, we assume that their proliferation rate
2.5 Mathematical Model for Intermittent Androgen Suppression 25 and apoptosis rate only depend on the androgen concentration within the tissue. The serum PSA is secreted by prostate cancer cells, and its level can be observed by medical blood examination. We can consider the PSA concentration is a function of time only. The PSA is a good bio-marker for estimating the progression for prostate cancer [16, 140], and IAS therapy is based on monitoring of the concentration of PSA.

We introduce the following physical variables:

- the androgen concentration a
- the number density of the AD cells x_1
- the number density of the AI cells x_2
- the serum PSA concentration y
- the velocity field within the tumor v.

Since IAS therapy is a dynamical therapy, we use a binary variable u(t) to describe IAS therapy of a prostate tumor: the medication is alternatively either present (u = 1) or absent (u = 0). The switch on/off of the medication is based on the monitored level y(t) of the serum PSA. Hence, the androgen concentration level a(t) (nmol/l) satisfies the following equation [89]:

$$\frac{da(t)}{dt} = -\gamma \left(a(t) - a_0 \right) - \gamma a_0 u(t), \qquad (2.49)$$

where a_0 (nmol/l) denotes the steady-state value of the normal androgen concentration, which takes a value between 15 and 30 for usual adult males. The rate of the recovery and decay of the androgen concentration is governed by the parameter γ .

26 2 Modeling Prostate Tumor Growth under Androgen Suppression

Under the assumptions mentioned above, the mass balance equations for the AD cell population and the AI cell population are given:

$$\begin{aligned} \frac{\partial x_1}{\partial t}(r,t) &+ \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 v(r,t) x_1(r,t) \right) \\ &= \frac{D}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial x_1}{\partial r}(r,t) \right) \\ &+ \left[\alpha_1 p_1(a(t)) - \beta_1 q_1(a(t)) - m(a(t)) \right] x_1(r,t), \end{aligned} \tag{2.50} \\ \\ \frac{\partial x_2}{\partial t}(r,t) &+ \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 v(r,t) x_2(r,t) \right) \\ &= \frac{D}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial x_2}{\partial r}(r,t) \right) + m(a(t)) x_1(r,t) \\ &+ \left[\alpha_2 p_2(a(t)) - \beta_2 q_2(a(t)) \right] x_2(r,t), \end{aligned} \tag{2.51}$$

where D is the random motility coefficient of the AD and AI cells, $\alpha_1 p_1(a(t))$ and $\beta_1 q_1(a(t))$ are proliferation rate and apoptosis rate of AD cells respectively, $\alpha_2 p_2(a(t))$ and $\beta_2 q_2(a(t))$ are proliferation rate and apoptosis rate of IA cells, m(a(t)) is the mutation rate by which AD cells mutate to AI cells. The cell proliferation rate, apoptotic rate, and mutation rate are assumed to be dependent on the androgen concentration a(t) as assumed. These androgen-dependent functions are given as follows [89]:

$$p_1(a(t)) = k_1 + (1 - k_1) \frac{a(t)}{a(t) + k_2},$$
(2.52)

$$q_1(a(t)) = k_3 + (1 - k_3) \frac{a(t)}{a(t) + k_4},$$
(2.53)

2.5 Mathematical Model for Intermittent Androgen Suppression 27

$$p_2(a(t)) = \begin{cases} (i) & 1, \\ (ii) & 1 - \left(1 - \frac{\beta_2}{\alpha_2}\right) \frac{a(t)}{a_0}, \\ (iii) & 1 - \frac{a(t)}{a_0}, \end{cases}$$
(2.54)
$$q_2(a(t)) = 1,$$
(2.55)

$$m(a(t)) = m_1 \left(1 - \frac{a(t)}{a_0} \right).$$
(2.56)

As shown in Eqs. (2.52)-(2.53), the proliferation rate of AD cells is an increasing function of the androgen concentration a while the apoptosis rate of AD cells is a decreasing function of the androgen concentration a, where $0 \le k_1 \le 1$ and $k_3 > 1$. The proliferation rate and apoptosis rate of AI cells also depends on the androgen concentration [100], although details of their androgen dependence remain unclear. We assume three possibilities on the net growth rate of the AI cells, as given in Eqs. (2.54) and (2.55), which are characterized by whether the net growth rate is positive, zero, or negative in androgen-rich environments.

The fact that the number density of cells is constant within a solid tumor [95, 168] gives,

$$x_1(r,t) + x_2(r,t) \equiv 1. \tag{2.57}$$

Further, we assume that the serum PSA concentration y(t) is monitored as a bio-marker for the prostate tumor growth. Since a large amount of PSA is secreted by cancer cells, the PSA concentration is assumed to be linearly dependent on the densities of the AD and AI cell populations as follows [89]: 28 2 Modeling Prostate Tumor Growth under Androgen Suppression

$$y(t) = \int_0^{2\pi} d\theta \int_0^{\pi} d\varphi \int_0^{R(t)} \left(c_1 x_1(r,t) + c_2 x_2(r,t) \right) r^2 \sin\varphi \, dr, \qquad (2.58)$$

where c_1 and c_2 are some constants. The PSA concentration y(t) can be used as a basis for intermittent administration in the IAS therapy model.

We switch the medication when the serum PSA marker level y(t) crosses the upper and lower thresholds [89]:

$$u(t) = \begin{cases} 0 \to 1 & \text{when } y(t) = r_1 \text{ and } dy(t)/dt > 0 \\ 1 \to 0 & \text{when } y(t) = r_0 \text{ and } dy(t)/dt < 0 \end{cases},$$
(2.59)

where r_1 and r_0 are the upper and lower thresholds, respectively. However, how to administer IAS therapy, or how to appropriately set adjustable parameters r_1 and r_0 under the condition of $r_1 > r_0 > 0$, is a very important question for clinical practice [12].

By the radial symmetry assumption of the problem, we have

$$\frac{\partial x_1}{\partial r}(0,t) = \frac{\partial x_2}{\partial r}(0,t) = 0, \quad v(0,t) = 0.$$
(2.60)

To complete the system, we impose the following initial and boundary conditions:

$$R(0) = R_0, \quad x_1(r,0) = x_{10}(r) \text{ with } 0 \le x_{10}(r) \le 1,$$
 (2.61)

$$\left[x_1(r,t)\frac{dR(t)}{dt} - \left(x_1(r,t)v(r,t) - D\frac{\partial x_1}{\partial r}(r,t)\right)\right]_{r=R(t)} = 0, \quad (2.62)$$

$$\left[x_2(r,t)\frac{dR(t)}{dt} - \left(x_2(r,t)v(r,t) - D\frac{\partial x_2}{\partial r}(r,t)\right)\right]_{r=R(t)} = 0, \quad (2.63)$$

where Eqs. (2.62) and (2.63) are no-flux boundary conditions for AD and AI cells respectively, as explained in Chapter 2.2.

The system (2.49)-(2.63) is the mathematical problem of prostate tumor growth under intermittent androgen suppression. In order to solve this problem, we need some more conditions or equations.

By adding Eqs. (2.50) and (2.51) and using the fact (2.57), we obtain an equation for the velocity field within the tumor,

$$\frac{1}{r^2} \frac{\partial}{\partial r} \Big(r^2 v(r,t) \Big) = \Big[\alpha_1 p_1(a(t)) - \beta_1 q_1(a(t)) \Big] x_1(r,t) \\ + \Big[\alpha_2 p_2(a(t)) - \beta_2 q_2(a(t)) \Big] (1 - x_1(r,t)). \quad (2.64)$$

By adding Eqs. (2.62) and (2.63) and using (2.57), we obtain an equation for the velocity of the outer boundary of the tumor,

$$\frac{dR(t)}{dt} = v(R(t), t).$$
(2.65)

This equation, together with Eq. (2.62), yields the following boundary condition for AD cells

$$\frac{\partial x_1}{\partial r}(R(t), t) = 0. \tag{2.66}$$

We notice that Eq. (2.51) is a consequence of Eqs. (2.50), (2.57), and the expression (2.64). Therefore, in what follows we may drop this equation and replace x_2 by $1 - x_1$ in Eqs. (2.50) and (2.58).

We also notice that the no-flux boundary conditions (2.62)-(2.63) are equivalent to the boundary conditions (2.65)-(2.66) under the condition (2.57). So shall replace Eqs. (2.62)-(2.63) with Eqs. (2.65)-(2.66).

In next section we will transform the problem (2.49)-(2.66) in the moving domain $\{r \leq R(t)\}$ into a new system in a fixed domain. 30 2 Modeling Prostate Tumor Growth under Androgen Suppression

2.6 Transformation of Intermittent Androgen

Suppression Model

To transform the moving domain $\{r \leq R(t)\}$ into a fixed domain, as done in Chapter 2.2, we introduce a change of variables $(r, t, a, x_1, v, R, y, u) \mapsto$ $(\rho, t, a, \tilde{x}_1, \tilde{v}, R, y, u)$ as follows:

$$\rho = r/R(t), \quad \tilde{x}_1(\rho, t) = x_1(\rho R(t), t), \quad \tilde{v}(\rho, t) = v(\rho R(t), t)/R(t). \quad (2.67)$$

For notational convenience, we drop the tildes of $\tilde{x}_1(\rho, t)$ and $\tilde{v}(\rho, t)$, in terms of the new variables, the system (2.49)-(2.66) takes the following form in $\{0 < \rho < 1, \ 0 < t < T\}$:

$$\frac{da(t)}{dt} = -\gamma \left(a(t) - a_0 \right) - \gamma a_0 u(t), \quad a(0) = a_0,$$

$$\frac{\partial x_1}{\partial t} (\rho, t) + \left[v(\rho, t) - \rho v(1, t) \right] \frac{\partial x_1}{\partial \rho} (\rho, t) - \frac{D}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial x_1}{\partial \rho} (\rho, t) \right)$$

$$= \left[\alpha_1 p_1(a(t)) - \beta_1 q_1(a(t)) - \alpha_2 p_2(a(t)) + \beta_2 q_2(a(t)) \right] x_1(\rho, t) (1 - x_1(\rho, t))$$

$$= m(a(t)) n (a, t)$$
(2.68)

$$-m(a(t))x_1(\rho, t),$$
 (2.09)

$$x_1(\rho, 0) = x_{10}(\rho), \tag{2.70}$$

$$\frac{\partial x_1}{\partial \rho}(0,t) = \frac{\partial x_1}{\partial \rho}(1,t) = 0, \qquad (2.71)$$

$$v(\rho,t) = \frac{1}{\rho^2} \int_0^{\rho} \left\{ \left[\alpha_1 p_1(a(t)) - \beta_1 q_1(a(t)) \right] x_1(s,t) + \left[\alpha_2 p_2(a(t)) - \beta_2 q_2(a(t)) \right] (1 - x_1(s,t)) \right\} s^2 ds, \quad (2.72)$$

$$\frac{dR(t)}{dt} = R(t)v(1,t), \quad R(0) = R_0,$$
(2.73)

2.7 Numerical Study of Intermittent Androgen Suppression Model 31

$$y(t) = c_0 R^3(t), (2.74)$$

$$u(t) = \begin{cases} 0 \to 1 & \text{when } y(t) = r_1 \text{ and } dy(t)/dt > 0 \\ 1 \to 0 & \text{when } y(t) = r_0 \text{ and } dy(t)/dt < 0 \end{cases},$$
(2.75)

where we have used the fact that v(0, t) = 0 in deriving Eq. (2.72), and we have used the assumptions $c_1 = c_2$ and (2.57) in deriving Eq. (2.74) ($c_0 := 4\pi c_1/3$). The model dynamics will be examined using numerical simulation in next section.

2.7 Numerical Study of Intermittent Androgen

Suppression Model

In this section we will perform numerical simulations of the model (2.68)-(2.75) to study the effects of IAS therapy in the three cases (i)-(iii) of the proliferation rate of the AI cells in (2.54). The model describes IAS therapy for $0 < r_0 < r_1$, while the model can be viewed to describe CAS therapy if $r_0 = 0$. The typical parameter values and the initial conditions for the numerical simulations are given[76, 89, 95]:

$$k_1 = 0, \quad k_2 = 2, \quad k_3 = 8, \quad k_4 = 0.5, \quad \alpha_1 = 0.0204, \quad \alpha_2 = 0.0242,$$

 $\beta_1 = 0.0076, \quad \beta_2 = 0.0168, \quad \gamma = 0.08, \quad a_0 = 30, \quad m_1 = 0.00005$
 $a(0) = 30, \quad u(0) = 1, \quad D = 0 \quad \text{and} \quad R(0) = 2 \text{ mm.}$

In the simulations, the initial condition for x_{10} is taken as follows:

$$x_{10} = 0.95.$$

32 2 Modeling Prostate Tumor Growth under Androgen Suppression

This initial condition implies that AI cells are minor at the beginning of the therapy. The parameters r_0 and r_1 are the adjustable parameters in our model.



Fig. 2.1. The dynamics of the serum PSA concentration y(t) in case (i), where $c_0 = 0.5$.

Figure 2.1 shows the dynamics of the serum PSA concentration y(t) in case (i). It indicates that IAS seems to be more ineffective than CAS in prolonging a relapse, which is similar to the dynamics of the ODE model [89]. We notice that the net growth rate (the proliferation rate – the apoptosis rate) of AI cells is always positive due to $\alpha_2 p_2 - \beta_2 q_2 = \alpha_2 - \beta_2 > 0$ in case (i), and therefore AI cells will be rapidly dominant within a prostate tumor. On the other hand, Figure 2.1 also shows that a relapse occurs within approximately 2.7 Numerical Study of Intermittent Androgen Suppression Model 33 three years in all the trials with different values of r_0 similarly with the CAS. Therefore, the IAS therapy may be worthy to be adopted in the sense that the periods of off-administration introduced by this therapy can reduce side effects of androgen deprivation and improve the quality of life for patients.



Fig. 2.2. The dynamics of the serum PSA concentration y(t) in case (ii), where $c_0 = 1.5$. $r_0 = 7$ is a nearly *optimal* lower threshold for IAS therapy if we fix the upper threshold at $r_1 = 15$.

The dynamics of the serum PSA concentration y(t) in case (ii) of Eq. (2.54) is presented in Figure 2.2. Unlike the previous case, all the trials with IAS therapy prolong the relapse time when compared with the CAS therapy. Figure 2.2 shows that the relapse time is *not* monotonously decreasing in the lower threshold $r_0 \in [1, 14]$ if $c_0 = 1.5$. Then, to achieve better effects of 34 2 Modeling Prostate Tumor Growth under Androgen Suppression

IAS therapy, the lower threshold r_0 should be carefully selected. In fact, to maximally delay the relapse time, a nearly *optimal* lower threshold $r_0 = 7$ is numerically found if the upper threshold is fixed at $r_1 = 15$ as shown in Figure 2.2. It is impossible, however, to avoid an eventually relapse in case (ii) due to $\alpha_2 p_2(a) - \beta_2 q_2(a) > 0$ for $0 \le a < a_0$ and $\alpha_2 p_2(a_0) - \beta_2 q_2(a_0) = 0$. This implies that the population of AI cells increases in androgen-poor environments, and does not decrease even in environments that have the normal androgen concentration a_0 .



Fig. 2.3. The dynamics of the serum PSA concentration y(t) in case (iii), where $c_0 = 0.8$ and $r_1 = 15.0$. y(t) will be periodically oscillatory for $0.31 \le r_0 < r_1$, where $r_0 = 0.31$ is the *critical* lower threshold for the existence of a stable periodic solution which corresponds to successful IAS.

Figure 2.3 shows the dynamics of the serum PSA concentration y(t) in case (iii) with $c_0 = 0.8$ and $r_1 = 15.0$. The CAS therapy leads to a relapse due to $\alpha_2 p_2(0) - \beta_2 q_2(0) = \alpha_2 - \beta_2 > 0$. This implies that the population of AI cells continues to increase in an androgen-deprived state. However, IAS therapy can realize cyclic growth and regression of the tumor without a relapse due to $\alpha_2 p_2(a_0) - \beta_2 q_2(a_0) = -\beta_2 < 0$, which implies that the population of AI cells decreases in an androgen-rich environment. In fact, the biological evidence that proliferation of AI cancer cells is repressed by androgen [102].

2.8 Summary

This chapter reviews mathematical models for prostate tumor growth under hormone therapy with mutation inhibitors. The prostate tumors have two types of cancerous cells, androgen-dependent (AD) cells and androgen-independent (AI) cells. AI cells are undetectable prior to treatments, but they grow even in androgen-poor conditions during continuous hormone therapy. The models are formulated as free boundary problems of nonlinear parabolic systems, and describe the evolution of cancerous cell populations within a prostate tumor and the dynamics of the tumor radius (volume).

We found explicit formulae of the tumor volume at any time t in continuously androgen-deprived environment. The long-term behavior of tumor growth can be predicted by these formulae. Qualitative analysis suggests that a tumor relapse cannot be avoided under androgen-deprived therapy. This implication 36 2 Modeling Prostate Tumor Growth under Androgen Suppression may support a possible strategy of intermittent androgen suppression (IAS). Intermittent androgen suppression is a type of androgen ablative therapy delivered intermittently with off-treatment periods [17].

In fact, a crucial problem of the CAS therapy on prostate cancer is the relapse of a prostate tumor, which has been shown by a number of experimental and clinical studies [12, 16, 17, 18, 83, 90, 95, 102]. This forces us to look for alternative method to treat prostate tumor, which is intermittent androgen suppression. We then reviews the PDE model [76] for IAS therapy. The numerical results show that the IAS therapy may be more effective than the CAS therapy in delaying a relapse in some cases (see Figs. 2.2 and 2.3). To achieve a better effect of the IAS therapy, we need to optimally administer the intermittent treatment. That is, we need to optimally choose the lower threshold r_0 (see Fig. 2.2), which depends on the parameters r_1 and c_0 .

We should mention that the nonlinear competitive effects between AD and AI cancer cells has been recently considered by Shimada and Aihara [134]. In the same way as presented in this chapter, Tao et al. [152] extended this ODE model to a PDE model and performed some qualitative and numerical analysis of the extended PDE model.

Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response

3.1 Introduction

In this chapter, we will review mathematical models for tumor virotherapy [144, 170, 65]. Unlike the model of Byrne and Chaplain [19] which introduces a genetic inhibitor, the present chapter considers a tumor therapy by oncolytic viruses, called virotherapy. Virotherapy has been and is continues to be actively tested in clinical trials for variety of malignant cancers [14, 34, 38, 77, 128, 181].

One of the obstacles in developing efficient gene therapies for cancers is in the delivery process. The macromolecules used as gene delivery carriers are too large to be transported into, and diffuse within, the tumor [92, 141]. A recent approach aimed at bypassing this problem involves the use of viruses. Viruses are engineered to selectively bind to receptors on the tumor cell surface (but not to the surface of normal healthy cells). The virus particles then gain entry by endocytosis and proceed to reproduce within the tumor cell, eventually causing death (lysis). On lysis of an infected cell, a swarm of new 38 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response viruses burst out of the dead cell, and then infect neighboring tumor cells.
This process continue until all cancer cells are destroyed.

However, most virus species are unable to eradicate the majority of tumor modules. There are increasing evidences that the host immune response to active viral infections. Indeed, it has been manifested that the innate immune system destroys infected cells as well as free virus particles, thus enabling the tumor to grow [33].

A mathematical model that describes the evolution of a solid tumor under viral injection was initially developed by Wu, Byrne, Kirn and Wein [178]. They computed and compared the evolution of the tumor under different initial conditions using the 'simplified' version of their model. Friedman and Tao [61] presented a somewhat different model, and made a rigorous mathematical analysis on their model. The main difference between the FT model and the WBKW model is a PDE for viruses. The WBKW model does not include the diffusion term and their mathematical system is not well posed.

Later on, Wein, Wu and Kirn [170] incorporated the immune response into their earlier model [178]. They used some preclinical and clinical data to validate their model and estimate several key parameter values. They also discussed some design of oncolytic viruses. The viruses should be designed for rapid intratumoral spread and immune avoidance, in addition to tumor-selectively and safety. Tao and Guo [144] made a rigorous mathematical analysis of the WWK model. A very interesting point of the analysis is to theoretically find an explicit threshold of the intensity of the immune response for controlling a tumor, which is the focus of the present chapter.

We also review a mathematical model proposed by Friedman and Tian et al [65]. This model is formulated for spherical glioma under virotherapy. The model studies administration of oncolytic virus hrR3 into the tumor center. HrR3 is a mutant of herpes simplex virus (HSV) [34]. Importantly, this model includes the effects of the immunosuppression drug cyclophosphamide. We used this mathematical model to determine how different protocols of cyclophosphamide treatment and how the burst size of mutated viruses will affect the growth of gliomas. It is showed that with the current burst size of oncolytic virus hrR3 the glioma tumor cannot be eradicated even with administration of cyclophosphamide. If, however, the viruses can be further altered to yield a burst size $b \geq 150$, then the glioma tumor will shrink to very small size even with no cyclophosphamide treatment.

In this chapter, we will mainly review the model [170] initially proposed by Wein et al. and present some analytical mathematical results [144] on this model. Section 3.2 gives the model. Section 3.3 re-formulates the model. Section 3.4 states the analytical results. Section 3.5 numerically studies a mathematical model with a time-delay of the immune response. Section 3.6 review the mathematical model for glioma virotherapy. Finally, this chapter is closed with a summary section. 40 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response3.2 Mathematical Model

Consider a radially symmetrical tumor, and denote its boundary by R(t). We use symbol r for the distance from a point to the center of the tumor, and rwill be a variable. We also introduce the following physical variables

> the number density of uninfected tumor cells : $\hat{x}(r,t)$ the number density of infected tumor cells : $\hat{y}(r,t)$ the number density of dead cells : $\hat{n}(r,t)$ the number density of free virus particles : $\hat{v}(r,t)$ the number density of the immune molecules : $\hat{z}(r,t)$ the velocity field within the tumor : u(r,t).

As that in Chapter 2.2, the proliferation and removal of cells cause movements of cells with the radial velocity field u(r,t), where u(0,t) = 0.

There is some biological evidences that many types of tumor cells secrete a variety of growth-promoting factors, for example, epidermal growth factor (EGF) and transforming growth factor- α (TGF- α) [182]. So, we usually consider that tumor cells proliferate. Following Wu *et al* [178], we assume that the uninfected cells proliferate exponentially at rate λ . Viruses infect tumor cells by binding to receptors on cell surfaces and gaining entry by endocytosis. Therefore, as in [178] we assume that the infection rate of cells centered at r equals a constant β times the density of uninfected cells at r times $\bar{v}(r, t)$, which is the spatially-weighted average of virus density on the surface of a spherical cell with radius r_c that is centered at r . In [178], Wu *et al* showed that

$$\bar{v}(r,t) = \frac{1}{2r_c} \int_{r-r_c}^{r+r_c} \hat{v}(s,t) ds.$$
(3.1)

Since the diffusion coefficient of tumor cells is much smaller than the diffusion coefficient of the motile virus particles, we assume that the tumor cells are subjected to the convection with velocity field u(r,t), and the diffusion of tumor cells is neglected. for a similar assumption, we refer to articles [175, 178, 179]. By the conservation law of mass for the uninfected tumor cells,

$$\frac{D\hat{x}}{Dt} \equiv \frac{\partial\hat{x}(r,t)}{\partial t} + \frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2u(r,t)\hat{x}(r,t)\right)$$

$$= \lambda\hat{x}(r,t) - \beta\hat{x}(r,t)\bar{v}(r,t).$$
(3.2)

As in papers [178, 179], we assume that the infected cells do not proliferate and that all infected cells undergo lysis at rate δ , where δ^{-1} represents the mean infected cell lifetime. Clinical results suggest that the immune response is cytokine-mediated. The expression of viruses in tumor cells sensitizes cells to lysis by the TNF (tumor necrosis factor) cytokine [130]. The binding of TNF to death receptors on the tumor cell surface preferentially induces apoptosis of viral-infected tumor cells, whereas the uninfected tumor cells are generally resistant to TNF-induced killing. Based on the ability of TNF to selectively kill viral-infected cells, we postulate that the immune response kills only the infected tumor cells. The immune response is assumed to kill the infected tumor cells at a rate (with proportionality constant k) proportional to the product of the densities of the infected tumor cells and the immune response. 42 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response We assumed that these killed cells do not release any free viruses. By the conservation law of mass for infected cells,

$$\frac{D\hat{y}}{Dt} \equiv \frac{\partial\hat{y}(r,t)}{\partial t} + \frac{1}{r^2}\frac{\partial}{\partial r}\Big(r^2u(r,t)\hat{y}(r,t)\Big)
= \beta\hat{x}(r,t)\bar{v}(r,t) - \delta\hat{y}(r,t) - k\hat{y}(r,t)\hat{z}(r,t).$$
(3.3)

At the time of lysis, an infected cell becomes necrotic. We also regard the infected cells which have been killed by the immune response as necrotic cells. The necrotic debris is removed from the tumor at rate μ . By the conservation law of mass for the necrotic cells,

$$\frac{D\hat{n}}{Dt} \equiv \frac{\partial\hat{n}(r,t)}{\partial t} + \frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2u(r,t)\hat{n}(r,t)\right)$$

$$= \delta\hat{y}(r,t) + k\hat{y}(r,t)\hat{z}(r,t) - \mu\hat{n}(r,t).$$
(3.4)

Since the free virus particles are very small relative to the cells, we assume that they undergo diffusion within the tumor tissue. The viruses addressed in this chapter play a role similar to an anti-tumor drug. The major difference between viruses and anti-tumor drugs is that viruses have replicating ability once they infect tumor cells and newly replicated viruses can infect neighboring tumor cell when infected tumor cell lysis. The drugs usually undergo diffusion within the tumor tissue [168]. There are some experimental evidences that the diffusion coefficient of drugs is lowest near the center of the tumor, and it is increasing to maximal levels at the tumor periphery [92]. However, for simplicity, one usually assume that the diffusion coefficient of a drug is constant, see [168]. In this chapter we shall also confine ourselves to consider the situation in which the virus diffusion coefficient (denoted by D_1) is constant. When an infected tumor cell dies, we assume that N new virus particles are released, and this number is call the burst size of the viruses. The combined parameter $N\delta$ is the release rate of free virus particles per unit time per infected cell. Because virus particles are present throughout a sphere of radius r_c (where r_c is the tumor cell radius), the rate of virus release at location r is $N\delta$ times $\bar{y}(r,t)$, which is the spatially-averaged infected cell density throughout a sphere of radius r_c that is centered at r. In [178], Wu *et al* derived that

$$\bar{y}(r,t) = \frac{1}{4r_c^3} \int_{r-r_c}^{r+r_c} \hat{y}(s,t) [r_c^2 - (s-r)^2] ds.$$
(3.5)

As in [178, 179], we assume that the infected cells killed by the immune cells do not release any free virus. The precise mechanism by which the viruses are cleared remains unknown. We assume that it is cleared at a constant rate γ , where $1/\gamma$ is the mean lifetime of free viruses. We shall also neglect the effect of the velocity field on the virus, as in [178, 179]. By combining the conservation law of mass with the effect of diffusion, we obtain

$$\frac{\partial \hat{v}(r,t)}{\partial t} = N\delta \bar{y}(r,t) - \gamma \hat{v}(r,t) + D_1 \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial \hat{v}(r,t)}{\partial r} \right), \quad \frac{\partial \hat{v}(0,t)}{\partial r} = 0, \quad (3.6)$$

where the last equation in (3.6) is a consequence of the radial symmetry.

Since the immune response (several innate immune cell populations) under consideration in this chapter mainly consists of molecules with very small size, we assume that it undergoes diffusion. Also, we will confine ourselves to consider the situation in which the diffusion coefficient of immune response 44 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response (denoted by D_2) is constant. Typically, $D_2 = 10^{-2} \ cm^2 day^{-1} \gg D_c = 1.3 \times 10^{-6} \ cm^2 day^{-1}$ (the diffusion coefficient of tumor cells) [27]. We assume that the immune response is produced at an unsaturated rate that is proportional (with constant l) to the product of the density of infected tumor cells and the immune response, and incurs second-order clearance with rate constant ω . We assume second-order clearance because first-order clearance was inconsistent with the clinical data [170]. By the conservation law of mass, we have

$$\frac{\partial \hat{z}(r,t)}{\partial t} = l\hat{z}(r,t)\hat{y}(r,t) - \omega[\hat{z}(r,t)]^2 + D_2 \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial \hat{z}(r,t)}{\partial r}\right), \quad \frac{\partial \hat{z}(0,t)}{\partial r} = 0,$$
(3.7)

where the last equation in (3.7) is a consequence of the radial symmetry.

We finally assume that all tumor cells have the same size, and that they are uniformly distributed in the tumor [178], so that

$$\hat{x} + \hat{y} + \hat{n} \equiv \text{const.} \equiv \theta.$$
 (3.8)

Adding equations (3.2)-(3.4) and using (3.8), we get the equation for velocity field,

$$\frac{\theta}{r^2}\frac{\partial}{\partial r}\Big(r^2u(r,t)\Big) = \lambda\hat{x}(r,t) - \mu\hat{n}(r,t).$$
(3.9)

The boundary conditions, at the outer boundary of the tumor, are

$$\frac{\partial}{\partial r}\hat{v}(R(t),t) = 0, \qquad (3.10)$$

$$\frac{\partial}{\partial r}\hat{z}(R(t),t) = 0, \qquad (3.11)$$

$$\frac{dR(t)}{dt} = u(R(t), t).$$
(3.12)

Remark 3.1. We assume that there is no-flux of tumor cells, virus particles and the immune response across the outer boundary of the tumor, that is, they remain within the tumor. Then, we have boundary conditions,

$$\left[\hat{x}(r,t)\frac{dR(t)}{dt} - \hat{x}(r,t)u(r,t)\right]_{r=R(t)} = 0,$$
(3.13)

$$\left[\hat{y}(r,t)\frac{dR(t)}{dt} - \hat{y}(r,t)u(r,t)\right]_{r=R(t)} = 0, \qquad (3.14)$$

$$\left[\hat{n}(r,t)\frac{dR(t)}{dt} - \hat{n}(r,t)u(r,t)\right]_{r=R(t)} = 0, \qquad (3.15)$$

$$\left[\hat{v}(r,t)\frac{dR(t)}{dt} - \left(\hat{v}(r,t)u(r,t) - D_1\frac{\partial\hat{v}}{\partial r}(r,t)\right)\right]_{r=R(t)} = 0, \quad (3.16)$$

$$\left[\hat{z}(r,t)\frac{dR(t)}{dt} - \left(\hat{z}(r,t)u(r,t) - D_2\frac{\partial\hat{z}}{\partial r}(r,t)\right)\right]_{r=R(t)} = 0.$$
(3.17)

Adding Eqs. (3.13)-(3.15) and using Eq. (3.8), we derived the free boundary condition (3.12). This, along with Eqs. (3.16) and (3.17), further yields the boundary conditions (3.10) and (3.11). In Eqs. (3.6) and (3.7) we neglected the effects of the velocity field on the virus and the immune response. However, we need to consider the velocity field in deriving appropriate and tractable boundary conditions for the viruses and immune response at the moving outer boundary of the tumor. We further notice that, as addressed in Remark 2.1, Eqs. (3.13)-(3.17) are obtained by considering the *relative* velocity of cells on the outer boundary of the growing tumor.

We notice that Equation (3.4) is a consequence of Eqs. (3.2), (3.3), (3.8) and (3.9). Therefore, in what follows we simply drop this equation and replace \hat{n} by $\theta - \hat{x} - \hat{y}$ in Eq. (3.9). 46 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response

3.3 Transformation

For the typical parameters $r_c = 0.01 \ mm$, $R(0) = 2 \ mm$ given in [178], we have $r_c \ll R(0)$. By (3.1) and (3.5) we get

$$\bar{v}(r,t) \approx \hat{v}(r,t), \quad \bar{y}(r,t) \approx \hat{y}(r,t).$$
 (3.18)

We introduce the new variables

$$\tilde{x}=\frac{\hat{x}}{\theta},\quad \tilde{y}=\frac{\hat{y}}{\theta},\quad \tilde{v}=\frac{\hat{v}}{\theta N},\quad \tilde{z}=\hat{z},\quad \tilde{u}=u$$

and the quantity

$$p_0 = \frac{\beta N\theta}{\gamma}.$$

The parameter p_0 is called the *basic reproductive ratio* in the epidemic study. It represents the mean number of virus particles released by one virus.

In terms of the new variables, the system (3.1)-(3.12) takes the following form:

$$\frac{\partial \tilde{x}(r,t)}{\partial t} = \lambda \tilde{x} - p_0 \gamma \tilde{x} \tilde{v} - \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \tilde{u} \tilde{x} \right), \tag{3.19}$$

$$\frac{\partial \tilde{y}(r,t)}{\partial t} = p_0 \gamma \tilde{x} \tilde{v} - \delta \tilde{y} - k \tilde{y} \tilde{z} - \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \tilde{u} \tilde{y} \right), \qquad (3.20)$$

$$\frac{\partial \tilde{v}(r,t)}{\partial t} = \delta \tilde{y} - \gamma \tilde{v} + D_1 \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial \tilde{v}}{\partial r} \right), \quad \frac{\partial \tilde{v}(0,t)}{\partial r} = 0, \quad (3.21)$$

$$\frac{\partial \tilde{z}(r,t)}{\partial t} = l\theta \tilde{z}\tilde{y} - \omega \tilde{z}^2 + D_2 \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial \tilde{z}}{\partial r} \right), \quad \frac{\partial \tilde{z}(0,t)}{\partial r} = 0, \quad (3.22)$$

$$\frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2\tilde{u}(r,t)\right) = \lambda\tilde{x} - \mu(1 - \tilde{x} - \tilde{y})$$
(3.23)

in the moving domain $\{r < R(t), t > 0\},\$

$$\frac{\partial \tilde{v}}{\partial r}(R(t),t) = 0, \quad t > 0, \tag{3.24}$$

3.3 Transformation 47

$$\frac{\partial \tilde{z}}{\partial r}(R(t),t) = 0, \quad t > 0, \tag{3.25}$$

$$\frac{dR(t)}{dt} = \tilde{u}(R(t), t), \quad t > 0,$$
(3.26)

$$\tilde{u}(0,t) = 0, \quad t > 0$$
 (3.27)

with initial conditions

R(0) is prescribed,

$$\tilde{x}(r,0) = \tilde{x}_{0}(r), \tilde{y}(r,0) = \tilde{y}_{0}(r), \tilde{v}(r,0) = \tilde{v}_{0}(r), \tilde{z}(r,0) = \tilde{z}_{0}(r)$$
where $\tilde{z}_{0}(r) > 0$ and $\tilde{x}_{0}(r), \tilde{y}_{0}(r), \tilde{v}_{0}(r)$ are nonnegative
$$(3.28)$$

functions with $x_0(r) + y_0(r) \le 1$, for $0 \le r \le R(0)$.

It is convenient to transform the moving region $\{0 \le r \le R(t)\}$ to the fixed region $\{0 \le \rho \le 1\}$. So, we use Landau transformation

$$\rho = \rho(r,t) = \frac{r}{R(t)},$$

and setting

$$\begin{aligned} x(\rho,t) &= \tilde{x}(r,t), \quad y(\rho,t) = \tilde{y}(r,t), \quad z(\rho,t) = \tilde{z}(r,t), \\ v(\rho,t) &= \tilde{v}(r,t), \quad u(\rho,t) = \tilde{u}(r,t)/R(t), \end{aligned}$$

Eqs. (3.19)-(3.23) combined with (3.27) take the following form in $\{0 < \rho < 1, t > 0\}$:

$$\frac{\partial x}{\partial t} + \left[u(\rho, t) - \rho u(1, t)\right] \frac{\partial x}{\partial \rho}
= \lambda x - p_0 \gamma x v - \left[-\mu + (\lambda + \mu)x + \mu y\right] x,$$
(3.29)

48 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response

$$\frac{\partial y}{\partial t} + \left[u(\rho, t) - \rho u(1, t)\right] \frac{\partial y}{\partial \rho}
= p_0 \gamma x v - \delta y - k y z - \left[-\mu + (\lambda + \mu) x + \mu y\right] y, \quad (3.30)$$

$$\frac{\partial v}{\partial t} - \rho u(1,t) \frac{\partial v}{\partial \rho} = \delta y - \gamma v + \frac{D_1}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial v}{\partial \rho} \right), \quad \frac{\partial v}{\partial \rho}(0,t) = 0, \quad (3.31)$$

$$\frac{\partial z}{\partial t} - \rho u(1,t) \frac{\partial z}{\partial \rho} = l\theta z y - \omega z^2 + \frac{D_2}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial z}{\partial \rho}\right), \quad \frac{\partial z}{\partial \rho}(0,t) = 0, \quad (3.32)$$
$$u(\rho,t) = \frac{1}{\rho^2} \int_0^\rho s^2 \left[-\mu + (\lambda + \mu)x(s,t) + \mu y(s,t)\right] ds. \quad (3.33)$$

The boundary and initial conditions (3.24)-(3.3) become

$$\dot{R}(t) = R(t)u(1,t), \quad R(0) \text{ is given},$$
(3.34)

$$\frac{\partial v}{\partial \rho}(1,t) = 0, \tag{3.35}$$

$$\frac{\partial z}{\partial \rho}(1,t) = 0, \tag{3.36}$$

$$x(\rho,0) = x_0(\rho), y(\rho,0) = y_0(\rho), v(\rho,0) = v_0(\rho), z(\rho,0) = z_0(\rho) \quad (3.37)$$

$$x_0(\rho) \ge 0, y_0(\rho) \ge 0, \quad x_0(\rho) + y_0(\rho) \le 1, \quad z_0(\rho) > 0.$$
 (3.38)

We also assume that

$$x_0(\rho), y_0(\rho), v_0(\rho) \text{ and } z_0(\rho) \text{ belong to } C^1[0,1], \text{ and } \frac{\partial v_0}{\partial \rho}(1) = \frac{\partial z_0}{\partial \rho}(1) = 0.$$

(3.39)

We now state a result on the global existence and uniqueness of solutions to the system (3.29)-(3.39) (see [144]).

Theorem 3.1. The system (3.29)-(3.39) has a unique solution $(x(\rho, t), y(\rho, t), v(\rho, t), z(\rho, t), u(\rho, t), R(t))$ with $x, \partial x/\partial \rho, y, \partial y/\partial \rho, v, \partial v/\partial \rho, z, \partial z/\partial \rho$ and $u, \partial u/\partial \rho$ in $C[0 \le \rho \le 1, 0 < t < \infty]$, R(t) in $C^1[0, \infty)$, and

3.3 Transformation 49

$$x(\rho, t) \ge 0, \quad y(\rho, t) \ge 0, \quad x(\rho, t) + y(\rho, t) \le 1.$$
 (3.40)

$$R(0)e^{-\beta t} < R(t) < R(0)e^{\beta t}$$
 for some $\beta > 0.$ (3.41)

Furthermore, if $z_0(\rho) = const. = z_0$, then

$$\frac{z_0}{1+z_0\omega t} \le z(\rho,t) \le \frac{1}{\frac{\omega}{l\theta} + \left(\frac{1}{z_0} - \frac{\omega}{l\theta}\right)e^{-l\theta t}} \le \max\left(z_0, \frac{l\theta}{\omega}\right).$$
(3.42)

We conclude this section by recalling, from [170], estimates on the size of parameters that appear in the system (3.29)-(3.32).

The parameter λ is typically small. For example, $\lambda = 3.2 \times 10^{-4} hr^{-1}$, which corresponds to a tumor doubling time of three months, typical for head and neck tumors [118]. The parameters δ and μ are one order of magnitude larger than λ . Laboratory results from Onyx suggest that the mean infected cell lifetime is about two days, and so we set $\delta = \frac{1}{48} hr^{-1}$. Typical necrotic removal rate $\mu = \frac{1}{72} hr^{-1}$. The tumor cell density $\theta = 10^6 cells/mm^3$. The parameter γ is at least one order of magnitude larger than δ and μ , and some laboratory data suggests that γ is about 1 hr^{-1} [178]. Typical value of the clearance rate ω of the immune response is 1.6 ml/ng-hr. The three parameters, the immune stimulation rate l, the immune killing rate k, and the basic reproductive ration p_0 , are more difficult to estimate. In [170], the authors use new clinical data to estimate that $p_0 = 3.73$, k =15.3 mm^3/ng -hr, $l = 0.048 mm^3/cells$ -hr.

The diffusion coefficients of several drugs given in [168] lay in the range $10^{-6} \sim 10^{-5} \ cm^2/s = 0.0864 \sim 0.864 \ cm^2/day$. The drug diffusion coefficient given in [93] is 1.7 $\ cm^2/day$. For clarity and definiteness, we will take $D_1 =$

50 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response 1 cm^2/day in (3.31). The immune response diffusion coefficient $D_2 = 7.0 \times 10^{-5} \sim 10^{-2} \ cm^2/day$ in (3.32), see [27] and the references therein.

3.4 Threshold of the Immune Response

In this section we study the effect of the immune response to cancer treatment. We will explore the threshold of the intensity of the immune response for controlling the tumor. We begin with the constant equilibrium solutions (x_s, y_s, v_s, z_s) to equations (3.29)-(3.32). We easily find there are precisely following six solutions of this type:

 $E_1 = (0, 0, 0, 0),$

$$\begin{split} E_2 &= (1, \ 0, \ 0, \ 0), \\ E_3 &= \left(0, \ 1 - \frac{\delta}{\mu}, \ \frac{\delta}{\gamma} (1 - \frac{\delta}{\mu}), \ 0\right), \\ E_4 &= \left(\frac{p_0 \delta^2 - \mu (p_0 \delta - \delta - \lambda)}{p_0 \delta (p_0 \delta - \lambda)}, \ \frac{(\lambda + \mu) (p_0 \delta - \delta - \lambda)}{p_0 \delta (p_0 \delta - \lambda)}, \\ &\qquad \frac{\delta}{\gamma} \cdot \frac{(\lambda + \mu) (p_0 \delta - \delta - \lambda)}{p_0 \delta (p_0 \delta - \lambda)}, \ 0\right), \\ E_5 &= \left(0, \ \frac{\omega (\mu - \delta)}{k l \theta + \mu \omega}, \ \frac{\delta}{\gamma} \cdot \frac{\omega (\mu - \delta)}{k l \theta + \mu \omega}, \ \frac{l \theta (\mu - \delta)}{k l \theta + \mu \omega}\right), \\ E_6 &= \left(\frac{k l \theta (\lambda + \mu) + \omega [p_0 \delta^2 - \mu (p_0 \delta - \delta - \lambda)]}{k l \theta (\lambda + \mu) + p_0 \delta \omega (p_0 \delta - \lambda)}, \ \frac{\delta}{k l \theta (\lambda + \mu) + p_0 \delta \omega (p_0 \delta - \lambda)}, \ \frac{l \theta (\lambda + \mu) (p_0 \delta - \delta - \lambda)}{k l \theta (\lambda + \mu) + p_0 \delta \omega (p_0 \delta - \lambda)}, \end{split}$$

From (3.42) in Theorem 3.1 we have that z(t) is nonnegative for all t > 0 if $z_0(\rho) = const. = z_0 > 0$. As a result, we are only interested in the stability of the equilibrium solutions in which $z_s > 0$. Hence, the equilibrium solutions E_1 , E_2 , E_3 and E_4 can be discarded. In the process of our analysis, we also

assume

$$\delta > \mu, \tag{3.43}$$

$$p_0 > 1 + \frac{\lambda}{\mu} + \frac{\lambda}{\delta} \tag{3.44}$$

which hold for the typical parameter values given in previous section. Since $\mu < \delta$ by (3.43), we see that $z_s < 0$ in E_5 , and hence this equilibrium solution can also be discarded. So, we are only interested in the equilibrium solution E_6 . In the following we shall prove that if the initial densities are approximately equal to E_6 , then the evolution of the tumor radius R(t) will strongly depend on the intensity $(m = \frac{lk\theta}{\omega})$ of the immune response.

Let $(x_s, y_s, v_s, z_s) = E_6$. We look at the system (3.29)-(3.39) with (x_0, y_0, v_0, z_0) near (x_s, y_s, v_s, z_s) . It is convenient to introduce new variables:

$$X(\rho, t) = x(\rho, t) - x_s, Y(\rho, t) = y(\rho, t) - y_s,$$

$$V(\rho, t) = v(\rho, t) - v_s, Z(\rho, t) = z(\rho, t) - z_s,$$

$$U(\rho, t) = u(\rho, t).$$

(3.45)

Then the system (3.29)-(3.39) takes the following form in $\{0 < \rho < 1, t > 0\}$:

$$\begin{aligned} \frac{\partial X}{\partial t} &+ \left[U(\rho, t) - \rho U(1, t) \right] \frac{\partial X}{\partial \rho} \\ &= \left[(\lambda + \mu) - (p_0 \delta + \mu) y_s - 2(\lambda + \mu) x_s \right] X \\ &- \mu x_s Y - p_0 \gamma x_s V - \left[(\lambda + \mu) X + \mu Y + p_0 \gamma V \right] X \\ &+ x_s [\lambda + \mu - (p_0 \delta + \mu) y_s - (\lambda + \mu) x_s], \end{aligned}$$
(3.46)

52 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response

$$\frac{\partial Y}{\partial t} + \left[U(\rho, t) - \rho U(1, t) \right] \frac{\partial Y}{\partial \rho}
= \left[p_0 \gamma v_s - (\lambda + \mu) y_s \right] X - \left[(\delta - \mu) + (\lambda + \mu) x_s + 2\mu y_s + k z_s \right] Y
+ p_0 \gamma x_s V - k y_s Z + \left[p_0 \gamma X V - (\lambda + \mu) X Y - \mu Y^2 - k Y Z \right]
+ y_s \left[\mu - \delta - \mu y_s + (p_0 \delta - \mu - \lambda) x_s - k z_s \right],$$
(3.47)

$$X(\rho, 0) = X_0(\rho), \quad Y(\rho, 0) = Y_0(\rho), \tag{3.48}$$

$$\frac{\partial V}{\partial t} - \rho U(1,t) \frac{\partial V}{\partial \rho} - \frac{D_1}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial V}{\partial \rho}\right) = \delta Y - \gamma V, \qquad (3.49)$$

$$V(\rho,0) = V_0(\rho), \quad \frac{\partial V}{\partial \rho}\Big|_{\rho=0} = \frac{\partial V}{\partial \rho}\Big|_{\rho=1} = 0, \quad (3.50)$$

$$\frac{\partial Z}{\partial t} - \rho U(1,t) \frac{\partial Z}{\partial \rho} - \frac{D_2}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial Z}{\partial \rho}\right) = (l\theta y_s - 2\omega z_s) Z - \omega Z^2 + l\theta ZY + l\theta z_s Y + z_s (l\theta y_s - \omega z_s),$$
(3.51)

$$Z(\rho,0) = Z_0(\rho), \quad \frac{\partial z}{\partial \rho}\Big|_{\rho=0} = \frac{\partial Z}{\partial \rho}\Big|_{\rho=1} = 0, \quad (3.52)$$

$$U(\rho,t) = \frac{1}{\rho^2} \int_0^{\rho} s^2 \left[Q + (\lambda + \mu) X(s,t) + \mu Y(s,t) \right] ds, \qquad (3.53)$$

$$\dot{R}(t) = R(t)U(1,t), \quad R(0) \quad \text{is given}, \tag{3.54}$$

where $X_0(\rho) = x_0(\rho) - x_s, Y_0(\rho) = y_0(\rho) - y_s, V_0(\rho) = v_0(\rho) - v_s, Z_0(\rho) = z_0(\rho) - z_s,$

$$Q \equiv -\mu + (\lambda + \mu)x_s + \mu y_s$$
$$= \frac{m\lambda(\lambda + \mu) - p_0\delta[\mu(p_0\delta - \delta - \lambda) - \lambda\delta]}{m(\lambda + \mu) + p_0\delta(p_0\delta - \lambda)}$$
(where $m = \frac{lk\theta}{\omega}$)

and $\frac{\partial V_0(\rho)}{\partial \rho}|_{\rho=1} = \frac{\partial Z_0(\rho)}{\partial \rho}|_{\rho=1} = 0$. Since (x_s, y_s, v_s, z_s) is an equilibrium solution, the last terms on the right-hand sides of (3.46), (3.47) and (3.51) vanish.

As in [178] we combine the immune response parameters l, k and ω into the single parameter $m = \frac{lk\theta}{\omega}$, which indicates the intensity of the immune response against the virus. That is, the immune response is strong when the value of l (the immune stimulation rate) and k (the immune killing rate) is large relative to ω (the immune response clearance rate).

In the following we shall assume

$$Q \neq 0. \tag{3.55}$$

Let us first consider the two eigenvalues (denoted by λ_1 and λ_2) of the linear coefficient matrix A of the hyperbolic system (3.46)-(3.47). Here

$$A = \begin{pmatrix} (\lambda + \mu) - (p_0 \delta + \mu)y_s - 2(\lambda + \mu)x_s & -\mu x_s \\ (p_0 \delta - \lambda - \mu)y_s & -(\delta - \mu) - (\lambda + \mu)x_s - 2\mu y_s - kz_s \end{pmatrix}$$
(3.56)

Notice that

$$l\theta y_s = \omega z_s, \quad \lambda + \mu - (p_0 \delta + \mu) y_s - (\lambda + \mu) x_s = 0,$$

then the matrix A can be rewritten as

$$A = \begin{pmatrix} -(\lambda + \mu)x_s & -\mu x_s \\ (p_0 \delta - \lambda - \mu)y_s & -(\delta - \mu) - (\lambda + \mu)x_s - (2\mu + m)y_s \end{pmatrix}$$

=: $\begin{pmatrix} a_{11} & a_{12} \\ a_{21} & a_{22} \end{pmatrix}$. (3.57)

54 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response By the assumptions (3.43) and (3.44) we easily find that

$$\lambda_1 + \lambda_2 \equiv a_{11} + a_{22} < 0,$$

$$\lambda_1 \cdot \lambda_2 \equiv |A| = a_{11}a_{22} - a_{12}a_{21} > 0.$$

Therefore, λ_1 and λ_2 have negative real parts,

$$Re\lambda_1 < 0, \quad Re\lambda_2 < 0.$$
 (3.58)

Thus, we may expect that the constant equilibrium solution $(x_s, y_s, v_s, z_s) = E_6$ to Eqs. (3.29)-(3.33) is locally asymptomatically stable.

In fact, we have the following results [144].

Theorem 3.2. Denote

$$m=\frac{lk\theta}{\omega}$$

and assume that

$$Q =: \frac{m\lambda(\lambda+\mu) - p_0\delta[\mu(p_0\delta-\delta-\lambda)-\lambda\delta]}{m(\lambda+\mu) + p_0\delta(p_0\delta-\lambda)} \neq 0,$$
(3.59)

$$\delta > \mu \text{ and } p_0 > 1 + \frac{\lambda}{\mu} + \frac{\lambda}{\delta}.$$
 (3.60)

If

$$\| X_0(\rho), Y_0(\rho), V_0(\rho), Z_0(\rho) \|_{C^1[0,1]} \le \varepsilon$$
(3.61)

where ε is sufficiently small, then there exists a unique global solution $(X(\rho, t), Y(\rho, t), V(\rho, t), Z(\rho, t), U(\rho, t), R(t))$ of (3.46)-(3.54) for all t > 0 with $R \in C^{1}[0, \infty)$, (X, Y, V, Z, U) in $C^{1}([0, 1] \times [0, \infty))$ and the following estimates:

$$|X(\xi(t),t)| + |Y(\xi(t),t)| \le Ce^{-\alpha t} || (X_0,Y_0)||_{C[0,1]},$$
(3.62)

3.4 Threshold of the Immune Response 55

$$\left|\frac{\partial X}{\partial \rho}(\xi(t), t)\right| + \left|\frac{\partial Y}{\partial \rho}(\xi(t), t)\right| \le Ce^{-\alpha t} \| (X_0, Y_0)\|_{C^1([0,1])},$$
(3.63)

which hold for all t > 0, where $\xi(t) = \xi(t; \rho_0)$ is any forward characteristic curve of the equations (3.46) and (3.47) satisfying $\rho_0 = \xi(0; \rho_0)$ and C is some positive constants independent of T. Furthermore, if

$$0 \le m < \frac{p_0 \delta[\mu(p_0 \delta - \delta - \lambda) - \lambda \delta]}{\lambda(\lambda + \mu)} =: m_0, \tag{3.64}$$

then

$$\dot{R}(t) < 0 \quad for \ all \quad t > 0, \tag{3.65}$$

$$R(0)e^{-\frac{|Q|}{2}t} \le R(t) \le R(0)e^{-\frac{|Q|}{6}t} \quad for \ all \quad t > 0;$$
(3.66)

if

$$m > m_0, \tag{3.67}$$

then

$$\dot{R}(t) > 0 \quad for \ all \quad t > 0, \tag{3.68}$$

$$R(t) \ge R(0)e^{\frac{|Q|}{6}t}$$
 for all $t > 0.$ (3.69)

Remark 3.2. In order to perform (3.1) and (3.5) mathematically, we need the following assumption

$$R(t) > r_c.$$

In fact, if the tumor radius $R(t) < r_c$ (here r_c is the radius of a tumor cell), then the tumor cannot contain a single live tumor cell and therefore the tumor is controlled. Actually, the models in [170, 178, 179] are valid only for $R(t) > r_c$. From $r_c = 0.01 \text{ mm}$, R(0) = 2 mm [178] we find that the tumor 56 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response nodule with a size of 2 mm in radius contains around 8×10^6 cells. So, the assumption $R(t) > r_c$ is minor, and the approximation (3.18) is reasonable.

Remark 3.3. In the absence of an immune response, the viruses (if uniformly injected throughout the tumor) may be powerful enough to eradicate the tumor. For example, if no immune response occurs (i.e., m = 0), then by Theorem 3.2 and the parameter values, $\lambda = 3.2 \times 10^{-4}$, $\delta = \frac{1}{48}$, $\mu = \frac{1}{72}$, $p_0 = 3.73$, given in Section 3.3, we easily find that

$$R(t) \to 0$$
 monotonously and exponentially as $t \to +\infty$.

Further, we see that without immune response (i.e., m = 0), the viruses can eradicate tumors with

$$\lambda \le \left(\frac{1}{\delta} + \frac{1}{\mu}\right)^{-1} (p_0 - 1) = 0.0225 \ hr^{-1}. \tag{3.70}$$

This means that any tumor with a doubling time more than 31 hours will be eradicated.

However, the strength of the immune-mediated clearance displayed in the human clinical data changes the entire picture. The viruses cannot control the tumor in the presence of the effective immune response. For example, by the clinical data given in [170], we have

$$m = \frac{lk\theta}{\omega} = 459 \ hr^{-1},$$

$$m_0 = \frac{p_0\delta[\mu(p_0\delta - \delta - \lambda) - \lambda\delta]}{\lambda(\lambda + \mu)} \approx 13.81 \ hr^{-1}.$$

Therefore

So, from Theorem 3.2 we get

 $R(t) \to +\infty$ monotonously and exponentially as $t \to +\infty$.

Because of the relative low of p_0 we see that only the slowest-growing tumors might be eradicated by the viruses in presence of immune-mediated clearance. By Theorem 3.2, in order to control tumor, we need the following condition:

$$m < m_0.$$

that is,

$$\lambda < \frac{p_0(p_0 - 1)\delta^2}{m} \approx 0.0963 \times 10^{-7} \ hr^{-1}$$
, approximately. (3.71)

Remark 3.4. The critical immune response intensity m_0 found in Theorem 3.2 is biologically interesting. In order to improve the efficacy of the oncolytic adenovirus, the immune-mediated viral clearance must be suppressed. One possible strategy to combat the effect of the immune response is to administer an immune suppressor, which would decrease the stimulation rate l in our model. For example, if

$$l < 1.4441 \times 10^{-3} \ mm^3/cells - hr,$$
 (3.72)

then, the tumor can be controlled from Theorem 3.2 for the typical parameter values given in Section 3.3. In fact, Friedman et al [65] proposed a mathematical model considering the effects of an immune suppressor. 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response *Remark 3.5.* Currently, biopharmaceutical companies and academic investigators working in the area of cancer viral therapies have made hundreds of viruses that are potential candidates for cancer therapy (see [170], for instance).

3.5 A Time-Delay Model

The data in [170] suggests that the immune system responds to the virus infection with a time delay of about 42 hours.

Using the quasi-steady-state approximation, $V = \frac{\delta}{\gamma} Y$, a simplified version of the system (3.46)-(3.54) with a time delay τ for the immune response takes the following form:

$$\frac{\partial X}{\partial t} + \left[U(\rho, t) - \rho U(1, t) \right] \frac{\partial X}{\partial \rho}$$

= $\left[(\lambda + \mu) - (p_0 \delta + \mu) y_s - 2(\lambda + \mu) x_s \right] X$
 $-\mu x_s Y - p_0 \gamma x_s V - \left[(\lambda + \mu) X + \mu Y + p_0 \gamma V \right] X,$ (3.73)

$$\frac{\partial Y}{\partial t} + \left[U(\rho, t) - \rho U(1, t) \right] \frac{\partial Y}{\partial \rho}$$

= $\left[p_0 \gamma v_s - (\lambda + \mu) y_s \right] X - \left[(\delta - \mu) + (\lambda + \mu) x_s + 2\mu y_s + k z_s \right] Y$
+ $p_0 \gamma x_s V - k y_s Z + \left[p_0 \gamma X V - (\lambda + \mu) X Y - \mu Y^2 - k Y Z \right],$ (3.74)

$$X(\rho, 0) = X_0(\rho), \quad Y(\rho, 0) = Y_0(\rho), \tag{3.75}$$

$$\frac{\partial Z}{\partial t} - \rho U(1,t) \frac{\partial Z}{\partial \rho} - \frac{D_2}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial Z}{\partial \rho}\right) = (l\theta y_s - 2\omega z_s) Z - \omega Z^2 + l\theta Z Y(\cdot, t - \tau) + l\theta z_s Y(\cdot, t - \tau), \qquad (3.76)$$

3.5 A Time-Delay Model 59

$$Z(\rho,0) = Z_0(\rho), \quad \frac{\partial Z}{\partial \rho}\Big|_{\rho=0} = \frac{\partial Z}{\partial \rho}\Big|_{\rho=1} = 0, \quad (3.77)$$

$$U(\rho,t) = \frac{1}{\rho^2} \int_0^{\rho} s^2 \left[Q + (\lambda + \mu) X(s,t) + \mu Y(s,t) \right] ds, \qquad (3.78)$$

 $\dot{R}(t) = R(t)U(1,t), \quad R(0)$ is given. (3.79)



Fig. 3.1. The spatio-temporal dynamics of the density $y(\rho, t)$ of the infected cells for $D_2 = 0.015$, $l = 1.8 \times 10^{-3}$, $\tau = 50$, $(x_s, y_s, z_s) = (0.97933, 0.00321, 0.00361)$. Take $(x_0(\rho), y_0(\rho), z_0(\rho)) \sim (x_s, y_s, z_s)$.

Figure 3.1 shows the spatio-temporal dynamics of the infected cells. Clearly, if the immune system responds to the virus with a time delay $\tau > \tau_0$ (where $\tau_0 \approx 28.6$ is the critical value for the stability of the steady state E_6 , see [144]), then the density of the infected cells will have an oscillation. This oscillation may become strong. Figure 3.1 also indicates the spatially heterogeneous distribution of the infected cells.



Fig. 3.2. The effects of the diffusivity D_2 of the immune response on tumor growth for $l = 1.8 \times 10^{-3}$, $\tau = 50$, $(x_s, y_s, z_s) = (0.97933, 0.00321, 0.00361)$. Take $(x_0(\rho), y_0(\rho), z_0(\rho)) \sim (x_s, y_s, z_s)$.

Figure 3.2 shows that the diffusivity D_2 of the immune response has little effects on the tumor growth when the time $t \leq 1100$ (hours) and the immune system responds to the virus with a time-delay $\tau > \tau_0$. However, the tumor growth will heavily influenced by the diffusivity D_2 when the time is large $(t > 1100 \text{ hours} \approx 46 \text{ days})$. Furthermore, we find that the tumor growth is monotonously increasing if the time t is sufficiently large. This phenomenon may be biologically explained as follows. The faster the immune response diffuses, the more quickly the viruses may be killed, which may mitigate the effects of the treatment.
3.6 Glioma Virotherapy

In this section, we review a mathematical model about glioma virotherapy proposed by Friedman and Tian et al [65]. This model is similar to a model introduced by Wu et al [178, 179]. However, there are several important differences. First, this model has included the effects of immunosuppressive agent Cyclophosphamide (CPA), and we used our model to determine the effect of administering CPA under different protocols on the progress of glioma. A second difference is that our model includes the presence of innate immune cells in the tumor tissue, whereas [178] includes only the immune response (tumor necrotic factor TNF) which consists of molecules with negligible volume. This difference is important, since the immune cells make up to 50% of the total number of cells at some stages of the tumor progression. We have also added a term which accounts for the destruction of virus particles by the immune cells [5]. Finally, in the mathematical model of [178], the parameters are estimated by using data from head and neck cancer. In our model, the parameters are estimated so as to conform with experimental results for glioma in article by Fulci G et al, Cyclophospharnide enhances glioma virotherapy by inhibiting innate immune response [68]. There is a substantial difference in the values of some of the parameters due to the fact that glioma is much more aggressive cancer.

62 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response

3.6.1 Mathematical model and parameter values

Our model includes infected tumor cells (y), uninfected tumor cells (x), necrotic cells (n), immune cells (z) and free virus particles (v). The model is given as follows.

$$\frac{\partial x(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} (r^2 u(r,t) x(r,t)) = \lambda x(r,t) - \beta x(r,t) v(r,t), \qquad (3.80)$$

$$\frac{\partial y(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} (r^2 u(r,t) y(r,t)) = \beta x(r,t) v(r,t) - k y(r,t) z(r,t) - \delta y(r,t),$$
(3.81)

$$\frac{\partial n(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} (r^2 n(r,t) u(r,t)) = ky(r,t)z(r,t) + \delta y(r,t) - \mu n(r,t), \quad (3.82)$$

$$\frac{\partial z(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} (r^2 z(r,t) u(r,t)) = sy(r,t)z(r,t) - c(z(r,t))z(r,t) - P(r,t)z(r,t) \quad (3.83)$$

$$\frac{\partial v(r,t)}{\partial t} - D\frac{1}{r^2}\frac{\partial}{\partial r}(r^2\frac{\partial v}{\partial r} = b\delta y(r,t) - k_0 v(r,t)z(r,t) - \gamma v(r,t), \qquad (3.84)$$

We assume that all the cells have the same size and that they are uniformly distributed in the tumor. Then

$$x(r,t) + y(r,t) + n(r,t) + z(r,t) = \text{const.} = \theta$$
 (3.85)

and, by [122], $\theta \approx 10^6 \text{cells}/mm^3$.

Combining the equations (3.80)-(3.83) and using (3.85), we obtain

$$\frac{\theta}{r^2} \frac{\partial}{\partial r} (r^2 u(r,t))$$

$$= \lambda x(r,t) - \mu n(r,t) + sy(r,t)z(r,t) - c(z(r,t))z(r,t) - P(r,t)z(r,t).$$
(3.86)

We also have

$$\frac{\partial v}{\partial r}(0,t) = 0, \qquad \text{for}t > 0.$$
 (3.87)

Since the free viruses remain in the tumor,

$$\frac{\partial v}{\partial r}(R(t),t) = 0, \text{ for } t > 0.$$
(3.88)

Finally, the free boundary is subject to the kinematic condition

$$\frac{dR(t)}{dt} = u(R(t), t).$$
(3.89)

In our simulation, we will take R(0)=2 mm and

$$x(r,0) = 0.84 \times 10^{6}, y(r,0) = 0.1 \times 10^{6},$$
$$z(r,0) = 0.06 \times 10^{6}, v(r,0) = Ae^{-\frac{r^{2}}{2^{2}}}, 0 \le r \le 2$$
$$5.2\pi \times 10^{8} \le 4\pi \int_{0}^{2} Ae^{-\frac{r^{2}}{2^{2}}} r dr \le 11.2\pi \times 10^{8}.$$

The way immune cells are cleared is typically by membrane protein Fas and FasL which, when combined on the cell surface, signal to caspase protein to execute the cell. The percentage of immune cells in the brain is typically 1-2%. When stimulated by infected cells in glioma, this percentage arises sharply. As the number of infected cells drops, the need for immune cells diminishes, so they undergo apoptosis, either by killing themselves (using their own Fas and FasL to activate caspase) or by killing each other (Fas from one cell ligands to FasL from another cell)[5] [121]. The first process occurs when z is small, and yields a linear clearance; the second process occurs when z is large, and it yields a quadratic clearance. Hence, 64 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response

$$c(z)z = \begin{cases} cz & \text{if } z < z_0, \\\\ \frac{c}{z_0}z^2 & \text{if } z \ge z_0. \end{cases}$$

From [121] one can infer the value of z_0 , and combining this with the linear clearance rate, we arrive at the number $\omega = \frac{c}{z_0} \approx 20 \times 10^{-8} mm^3 / hour \cdot cell$. For simplicity we will assume only quadratic clearance with a rate ω as above.

After CPA is administered to the rats, the level of CPA arises and reaches a plateau after 2 days. This level is maintained for approximately 3 days and then begins to drop off to zero in the next two days. We simulate the CPA level in the tumor by a piecewise linear function

$$P(t) = \begin{cases} 8.5 \times 10^{-2} & \text{if } 0 \le t \le 72, \\ \\ \frac{8.5 \times 10^{-2}}{48} (120 - t) & \text{if } 72 \le t \le 120, \end{cases}$$

and P(t) = 0 if $t \ge 120$, where the unit of P(t) is 1/hour.

The model parameters are based on experimental results of Fulci et al ??. In these experiments D74/HveC Rat Glioma cell lines were implanted into the brain of rats. After 7 days, the tumor reached the size of 4mm in diameter, and then the oncolytic virus hrR3 (which is a mutant of Herpes Simplex virus 1 (HSV)) was injected into the center of the tumor. This mutant attacks tumor cells, but does not attack healthy normal cells. Six hours after injection, some rats were sacrificed, the tumor was stained and the JPEG pictures were taken, and then the infected area of the tumor was measured. This procedure was repeated after 72-76 hours from the time of virus injection, and then once more, one week after virus injection. Five different stains were used: one for each of four innate immune cells, and one for the infected cells. The immune cells are CD11b (dendritic cells), CD68+ (main monocytes, also called ED1 in rats), CD163+ (also called ED2 in rats) and CD169 (also called ED3 in rats). Different immune cells may participate in the inflammatory response at different time points and may be cleared at different time points. Rapid up-regulation was observed for certain immune cells, and depletion of some macrophages alone has also been produced. However, the CPA effects on the different immune cells were only partially reproduced. For this reason, we shall take in our model the average response of all the immune cells, rather than the response of each population of immune cells. The immunosuppression drug CPA was pre-administered to rats on the fifth day after tumor cell implantation, that is, 2 days before virus injection. Most of the parameter values we obtained from experimental results or literature. We provide them here: Proliferation rate of tumor cells $\lambda = 2 \times 10^{-2}$ per hour, Infected cell lysis rate $\delta = \frac{1}{18}$ per hour, Removal rate of necrotic cells $\mu = \frac{1}{48}$ per hour, Burst size of infected cells b = 50 viruses per cell, Density of tumor cells $\theta = 10^6$ cells per mm³, Diffusion coefficient of viruses $D = 3.6 \times 10^{-2} \text{ mm}^2$ per hour, Infection rate $\beta = \frac{7}{10} \times 10^{-9} \text{ mm}^3$ per hour per virus, Immune killing rate $k = 2 \times 10^{-8} mm^3$ per hour per immune cell, Take-up rate of viruses $k_0 = 10^{-8} mm^3$ per hour immune cell, Stimulation rate by infected cells $s = 5.6 \times 10^{-7} mm^3$ per hour per infected cell, Clearance rate of immune cells $\omega = 20 \times 10^{-8} mm^3$ per hour per cell, Clearance rate of viruses $\gamma = 2.5 \times 10^{-2}$ per hour.

3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response3.6.2 Comparing with data

In order to compare our numerical simulation with the experimental results of article [68] we take, as in article [68], the initial radius of the tumor to be 2mm, and the number of particle forming units (pfu) of virus injected at the center to be 10^{8} - 10^{9} . The initial time is the seventh day after tumor implantation in the rat's brain.



Fig. 3.3. Experimental data vs simulation results. Figure 1A is for infected tumor cells without CPA; Figure 1B is for immune cells without CPA; Figure 1C is for innate immune cells with CPA. In each picture, data group 1 is after 6 hours, data group 2 is after 72 hours and data group 3 is before the rat dies.

Fig.3.3 1A compares the experimental measurements with our numerical simulation of the percentage of infected tumor cells (relative to the total number of all cells) without pretreatment of CPA. Fig.3.3 1B compares the experimental measurement data with the numerical simulation of the percentage of the innate immune cells without pretreatment of CPA. Fig.3.3 1C compares the experimental measurements of immune cells, when CPA was administered, with our numerical simulation. The discrepancy between measurements and simulation develops only after a relatively long time both in Fig.3.3 1B and 1C.

Fig.3.3 shows that our model fits quite well with experiments. We next proceed to compare the cancer progression with and without CPA.

3.6.3 Comparing studies

Fig.3.4 shows simulation results based on our model. Fig.3.4 A shows the



Fig. 3.4. Profiles of the percentages of cell populations within the tumor with and without CPA. Figure A is for innate immune cells, B is for infected tumor cells, and C is for uninfected tumor cells.

profiles of the averages over space, of immune cell densities with and without CPA pre-treatments. Without CPA, the immune cells reach the maximum 52% at 26th hour after virus injection; with CPA, the immune cells reach the maximum 34% at 24th hour after virus injection. Thus, CPA suppresses the maximum level of innate immune cells and shortens the time that the immune system reaches its peak. Since the effect of CPA disappears after 120 hours, the percentage of the immune cells climbs up after 120 hours, thus forming a bimodal profile. 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response Fig.3.4 B shows the profiles of the percentage of infected tumor cell with and without CPA. Clearly with CPA we expect to have more infected cells.
As the simulation shows, without CPA, the infected cells reach the maximum 46% at 5th hour; with CPA, the infected cells reach the maximum 50% at 7th hour approximately.

Fig.3.4 C shows the profiles of the percentage of uninfected tumor cells with and without CPA. The first thing to notice is that there is a time delay in the effect of CPA; the immune suppression does not begin right away; in fact, there is a 3 days delay. The effect of CPA becomes negligible after approximately 17 days. However, during the intermediate period, it is significant.

3.6.4 Burst size of viruses

Suppose we inject into the tumor OV which replicates at a faster rate than hrR3. A large burst size b of virus will increase the stimulation of the immune system, which will then attack the infected cells and the free viruses. As a result, the population of viruses will decrease and this will be followed by a decrease in the population of immune cells. The population of virus and infected cells will then be able to increase, and it will follow by a re-stimulation of the immune system, etc. Thus, we may expect a "feedback mechanism" which will cause an oscillatory behavior of the percentages of infected cells, immune cells, and uninfected cells, with slight time-shift of the corresponding maxima. This is indeed demonstrated in Fig.3.5 for burst size b = 400 and

1000. For smaller values of b such as b = 200, oscillations occur only in the first 20-30 days. For b = 50 (not shown here), we do not see any oscillation.



Fig. 3.5. Profiles of cell populations within the tumor with different burst sizes. Figure A is for infected tumor cells, B is for uninfected tumor cells, and C is for innate immune cells.

3.6.5 CPA treatments

We compare two different protocols for administering CPA. The first protocol is to administer a "normal" amount of CPA weekly, and the second protocol is to administer twice the normal amount every two weeks. The simulation results of the percentage of uninfected tumor cells for burst sizes b = 100,200and 400 is shown in Fig.3.6.

These figures show that there is little difference between the weekly and bi-weekly treatments, especially when we consider weekly averages.

It is instructive to compare these treatments with the "traditional" treatment, where we pre-administer normal amount CPA just once. The weekly treatment reduces the weekly averages of the uninfected cells, but due to oscillations

70 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response



Fig. 3.6. Profiles of uninfected tumor cell populations with different treatments and different burst sizes. Figure A is for burst size 100, B is for burst size 200, and C is for burst size 400.

there are periods of time when this traditional treatment yields a small percentage of uninfected cells.

3.6.6 The tumor radius

All the preceding numerical results are based on solving the partial differential equations (PDEs) of the model given in the Appendix and then taking averages over the tumor region. We have obtained approximately the same numerical results using the much simpler ordinary differential equations (ODEs) obtained by neglecting spatial variations.

There is however one important quantity that we have not yet taken into account, namely, the radius of the tumor, and in order to compute it we need to work directly with the PDE system.

As the simulations presented above show, uninfected tumor cells will persist even with large burst sizes and with repeated CPA treatments. However, if the radius of the tumor can be kept small enough then long term survival of the animals can be assured, unless cell invasion and metastasis will occur as a result of cell shedding and migration.

Fig.3.7 simulates the growth of the tumor radius for b = 50 without CPA and with one pre-treatment of CPA.



Fig. 3.7. Tumor radius with different burst sizes and CPA treatments.

According to Fulci et al [68], without CPA the rats die after 8-10 days after the injection of viruses, and with one CPA pretreatment the rats die after 11-13 days after the injection of viruses; at death, the radius of the tumor is approximately 6 mm. These experimental results roughly fit with the simulations in Fig.3.7a.

Fig.3.7b simulates the growth of the tumor's radius without CPA and with weekly CPA treatment for different burst sizes.

We see that with weekly CPA treatment, the radius of the tumor will decrease as long as the burst size is bigger than 100; without any CPA treatment and with burst size b already as high as 150, the radius of the tumor will decrease to 1 mm. If the burst size could be increased to 400, the 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response tumor will shrink to an extremely small size. Finally, the weekly treatment by CPA helps in decreasing the radius of the tumor, as can by seen by comparing the profiles in Fig.3.7b for burst size b = 130 and b = 150: For b = 130 with CPA we achieve a smaller radius than for b = 150 without CPA.

3.6.7 Conclusion of glioma virotherapy

We have shown that with the current burst size of oncolytic virus hrR3 the tumor cannot be eradicated with CPA treatment. In fact, its radius will grow, and the rats will die within several weeks. If, however, the virus can be further altered to yield a burst size $b \ge 150$, then the tumor will shrink to a very small size, even with no CPA treatment. It is well known that tumor cells in glioma may shed and migrate into other areas in the brain. We did consider this infiltration/invasion problem. Thus even when the tumor size can be kept very small, there is still a chance of developing a secondary tumor. In this respect, a repeated treatment of the tumor by CPA is important, for it decreases the percentage of uninfected tumor cells, and thus reduces the risk of secondary tumors.

As was shown, by our model, that there is little difference between the weekly and bi-weekly CPA treatments. The protocol of choice should therefore depend on the side effects to this chemotherapy.

Finally, our model considers only spherical gliomas with oncolytic virus injection at the center. But we expect that conclusions here will hold also for non-spherical gliomas. Based on this mathematical model, there are several studies on particular aspect of the glioma virotherapy. For example, an analysis [157] shows the replicability of oncolytic virus is a critical parameter, [158, 159] show rich and complex dynamical behavior involved in virotherapy.

3.7 Summary

Replication-selective viruses as a novel therapeutic approach for cancer treatment have now been used in clinical trials. The immune response to anti-tumor virus can cause early elimination of viruses but also possibly be beneficial because it causes the immune-mediated killing of tumor cells. So, the competitive dynamics between tumor cells, a replication virus and the immune response is complex. To figure out the complex dynamics and help to design optimal protocols of tumor treatment, mathematical models may be needed. Wein, Wu and Kirn [170] have formulated a model describing the complex interplay between tumor cells, a replication-competent virus that kill tumor cells, and an immune response that kills the virus-infected tumor cells (and hence the virus itself). The mathematical model is a free boundary problem for a nonlinear system of partial differential equations. The variables are the radius of the tumor r = R(t), the evolving densities x(r, t), y(r, t), n(r, t), v(r, t) and z(r, t)of the uninfected cells, the infected cells, the dead cells, the free viruses and the immune response, and the velocity field u(r, t). 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response Importantly, an explicit and theoretical threshold m_0 of the intensity m

of the immune response for controlling the tumor was found by Tao and Guo

- [144] in the following sense:
 - (i) If $0 \le m < m_0$, then for the initial densities

$$x(r,0) \sim \text{const.} = x_s, \ y(r,0) \sim \text{const.} = y_s,$$

 $v(r,0) \sim \text{const.} = v_s, \ z(r,0) \sim \text{const.} = z_s,$

the radius R(t) will *decrease* monotonically and exponentially to zero as t increase to ∞ ;

(ii) If m > m₀, then for the same initial densities (x(r,0), y(r,0), v(r,0),
z(0)) ~ (x_s, y_s, v_s, z_s) as in (i), the radius R(t) will *increase* monotonically and exponentially to ∞ as t increase to ∞.

The above result suggests that the efficacy of the replication-competent viruses for treatment of tumors *strongly* depends on the intensity of an innate immune response against virus-infected tumor cells. Therefore, the viruses should be designed to immune avoidance, in addition to tumor-selectivity and safety. One possible strategy to combat the effect of the immune response is to co-administer an immune suppressor, which have been considered by Friedman et al. [65]. It is reported that hundreds of viruses have now been tested preclinically, and at least ten have already initiated testing in humans [65, 179].

We have found that for one pair of initial densities

$$x(r,0) \sim \text{const.} = x_s, \ y(r,0) \sim \text{const.} = y_s, z(r,0) \sim \text{const.} = z_s$$

the following is true:

Given any initial radius R(0), if the intensity m of the immune response against the virus is weaker than the threshold m_0 , by injecting virus particles of density

$$v(r,0) \sim \text{const.} = v_s \quad \text{where } v_s = \frac{\delta}{\gamma} y_s,$$

the radius R(t) will decrease monotonically and exponentially to zero as t increase to ∞ .

This example suggests that one may reduce tumors with doses of viral densities that are not necessarily large, which is quite different from the traditional chemotherapy in the sense that the doses of drugs of chemotherapy are usually large [92]. The above difference may be due to the replicating ability of virus within the tumor tissue.

We also review a similar mathematical model for glioma virotherapy in Section 3.6. This model includes the effects of immunosuppressive agent Cyclophosphamide (CPA). This model has a new term which accounts for the destruction of virus particles by the immune cells. This model explicitly includes the biological parameter virus burst size b. As a concrete study, the parameter values are estimated so as to conform with experimental results for glioma in article [68]. We used this model to determine the effect of administering CPA under different protocols on the progress of glioma. We showed that with the current burst size of oncolytic virus hrR3 the tumor cannot be eradicated with CPA treatment. In fact, the gliomas will grow, and 3 Modeling the Competition between Tumor Cells, Oncolytic Viruses and Immune Response the rats will die within several weeks. But, if the virus can be further altered to yield a burst size $b \ge 150$, then the tumor will shrink to a very small size, even with no CPA treatment. There is a difference between CPA pre-treatment and repeated treatment, although there is little difference between the weekly and bi-weekly CPA treatments.

This chapter also reviews the spatio-temporal heterogeneity of cancer cells caused by the time-delay of the immune response. In fact, spatial heterogeneity is an important topic in mathematical biology; see [28, 41], for instance.

4.1 Introduction

Traditional therapy for tumors is chemotherapy. Mathematicians have developed some mathematical models to study and improve chemotherapy, for example, models presented in articles [93, 168] and rigorously analysis presented in articles [143, 145]. However, one of the obstacles in developing efficient chemotherapy to cancers is in the delivery process. The macromolecules used as drug delivery carriers are too large to be transported into, and diffuse within, the tumor [92]. Recently, replication-competent viruses have been proposed as an approach to bypass the delivery problem. The advantage of replicating viruses for cancer therapy is the establishment of the persistent infection with ongoing viral proliferation. The virus is engineered to selectively bind to receptors on the tumor cell surface (but not to the surface of normal healthy cells). The virus particles then gain entry by endocytosis and proceed to proliferate exponentially within the tumor cell, eventually causing death (lysis). Thereupon the newly reproduced virus particles are released and then

proceed to infect adjacent cancer cells. Mathematical models of virotherapy have recently been developed [178] and studied [61]. However, a major factor influencing the efficacy of virus agents is immune response. New clinical data [117, 118] revealed an innate immune response to viruses that may mitigate the effects of treatments. The interaction between tumor cells, a replication-competent virus and the potential immune response is a complex biological process [170, 175, 176, 179].

The study in [175] suggests that a fast growth rate of the tumor decreases the efficacy of treatments with viruses, the success of therapies can be promoted by using a combination of viral therapy and conventional chemotherapy or radiotherapy. These suggestions are supported by experimental data [56, 77, 129]. The studies in [49, 50, 51] show that while some cell lines are sensitive in vitro to the oncolytic effect of the virus, tumor xenografts in animal models can persist despite repeated doses of the virus. In order to circumvent this problem, Dingli *et al* [49, 50] engineered the virus to induce expression of the human sodium iodide symporter that allows infected tumor cells to concentrate on iodide isotopes. This virus retains the natural oncolytic activity of the parent virus but has the advantage that it can eliminate tumors resistant to the virus when it is combined with radioiodide [49]. The ODE model developed by Dingli *et al* in [51], is the first mathematical model for radiovirotherapy.

Radiovirotherapy requires not only injection of viruses but also administration of radioiodide. Radioiodide is in a continuous state of flux between the tumor and the remaining part and continuously lost from the system due to both physical decay as well as losses to the environment due to excretion [49]. The experimental data in [49, 50] were obtained for immunocompromised mice and these mice do not mount an immune response to either the virus or tumor. On the other hand, the model in [51] focuses on multiple myeloma and patients with multiple myeloma, the disease for which the virus was designed, have profound defects in the immune system. So, the model in [51] does not explicitly include the immune system.

Tao and Guo [146] developed a PDE model for cancer radiovirotherapy, which is a generalization of Dingli *et al*'s ODE model [51]. The ODE model in [51] provides a simplified description of cancer radiovirotherapy as homogeneity assumption is imposed. However, more effective therapy requires a deep understanding of non homogeneity of tumor spatial structures. In this chapter, we review the PDE model for radiovirotherapy proposed by Tao and Guo (2007). Modeling method in this article is a combination of that in [51, 61] and [178]. The tumor volume is modeled as an incompressible fluid, through which cells move via a convective field. Local changes in cell numbers lead to variations of the internal pressure, which in turn induce motion of tumor cells.

((The structure of this chapter is as follows. Section 4.2 presents the model. Section 4.3 scales the system of equations by non-dimensionalization and reduces the free boundary problem into a problem in a fixed region. Section 4.4 discusses constant equilibrium solutions. Section 4.5 is devoted to study the stability of equilibrium solutions and to find an explicit parameter condition

for tumor eradication by this novel therapy. Section 4.6 numerically studies the model to explore possible optimal therapy strategies. Finally, this chapter is closed with a summary and discussion section.))

4.2 Mathematical Model

In order to model the effects of radiation on tumor cells, we introduced a population of cells that is irreparably damaged by radiation, in addition to populations of virus particles, uninfected cells, infected cells and necrotic cells. These damaged cells do not proliferate, and are destined to die. But they still occupy space and compete for nutrients, and therefore should contribute to the self-regulation of tumor growth. We then introduce the following physical variables,

- $\hat{x} = \text{density of uninfected tumor cells},$
- $\hat{y} = \text{density of infected tumor cells},$
- $\hat{z} = \text{density of tumor cells irreparably damaged by radiation,}$
- $\hat{n} = \text{density of necrotic cells},$
- $\hat{v} = \text{density of free viruses in the extracellular tissues},$
- u = the velocity field within the tumor.

The velocity field is a result of the spatio-temporal variation due to the proliferation of uninfected tumor cells and the removal of necrotic cells. Local changes in cell population lead to variations in the internal pressure, which in turn, induce motion of tumor cells. We assume that the tumor is radially symmetric, so that all unknown variables or functions depend only on (r, t), where r is the distance from the tumor center and t is time. The model was derived by applying the principle of mass conservation to each of the cell population. The model consists of the following system of equations:

$$\frac{D\hat{x}}{Dt} \equiv \frac{\partial \hat{x}}{\partial t}(r,t) + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 u(r,t) \hat{x}(r,t) \right)$$

$$= \lambda \hat{x}(r,t) - \beta \hat{x}(r,t) \hat{v}(r,t) - \kappa D(t) \hat{x}(r,t), \quad (4.1)$$

$$\frac{D\hat{y}}{Dt} \equiv \frac{\partial\hat{y}}{\partial t}(r,t) + \frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2u(r,t)\hat{y}(r,t)\right)
= \beta\hat{x}(r,t)\hat{v}(r,t) - \delta\hat{y}(r,t) - \kappa D(t)\hat{y}(r,t),$$
(4.2)

$$\frac{D\hat{z}}{Dt} \equiv \frac{\partial\hat{z}}{\partial t}(r,t) + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 u(r,t) \hat{z}(r,t) \right)$$

$$= \kappa D(t) \left(\hat{x}(r,t) + \hat{y}(r,t) \right) - \sigma \left(\hat{z}(r,t) \right)^{\nu},$$
(4.3)

$$\frac{D\hat{n}}{Dt} \equiv \frac{\partial\hat{n}}{\partial t}(r,t) + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 u(r,t)\hat{n}(r,t) \right)
= \delta\hat{y}(r,t) + \sigma \left(\hat{z}(r,t) \right)^{\nu} - \mu \hat{n}(r,t),$$
(4.4)

$$\frac{\partial \hat{v}}{\partial t}(r,t) = N\delta\hat{y}(r,t) - \gamma\hat{v}(r,t) + d\frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2\frac{\partial \hat{v}}{\partial r}(r,t)\right), \quad \frac{\partial \hat{v}}{\partial r}(0,t) = 0.$$
(4.5)

Although we do not consider the microenvironment within the tumor, we implicitly assume that tumor cells have ample nutrients, and allow them to experience first-order growth with a rate constant λ in (4.1) [178]. β is the infection rate of the uninfected cells. The rate κ by which tumor cells (both virus infected and uninfected) become irreparably damaged by radiation is

assumed to be proportional to the radiation dose D(t) absorbed by these cells and to their densities [51, 164]. Parameter κ is taken to be the constant for simplicity.

In (4.2), δ is the death rate of the infected cells from lysis. This parameter measures viral cytotoxicity.

In (4.3), the term $\sigma(\hat{z}(r,t))^{\nu}$ represents the effective death rate of damaged cells. Power-law dependence of rates is known to represent general behavior of biological systems [131]. The data in [51] suggests that $0 < \nu \leq 1$.

In (4.4), μ is the removal rate of the necrotic cells.

In (4.5), γ is the removal (or clearance) rate of viruses (1/ γ is the mean lifetime of free viruses), $N\delta$ is the virus release rate (N is the burst size of virus at the death of an infected cell), and d is the diffusion coefficient of viruses.

Radioiodide is in a continuous state of flux between the tumor and the remaining part. Iodide undergoes beta particle decay. The emitted beta particles have a path length of 0.8 mm with a significant effect on tumor cells [49, 50]. In model (4.1)-(4.5), we assume that after injection, radioactive iodine is rapidly distributed within the tumor, and that only the radioactivity at tumor site contributes to the absorbed dose [51]. The absorbed radiation dose D(t) is proportional to the cumulative activity [49, 82].

We finally assume that all tumor cells have the same size, and that they are uniformly distributed in the tumor. Then,

4.2 Mathematical Model 83

$$\hat{x} + \hat{y} + \hat{z} + \hat{n} \equiv \text{const.} \equiv \theta.$$
 (4.6)

Summing equations (4.1)-(4.4) together and using (4.6) yield

$$\frac{\theta}{r^2}\frac{\partial}{\partial r}\Big(r^2u(r,t)\Big) = \lambda\hat{x}(r,t) - \mu\hat{n}(r,t).$$
(4.7)

The boundary conditions, at the outer boundary of the tumor, are

$$\frac{\partial}{\partial r}\hat{v}(R(t),t) = 0, \qquad (4.8)$$

$$\frac{dR(t)}{dt} = u(R(t), t), \tag{4.9}$$

which can be biologically explained as in Chapter 2.

We notice that Eq. (4.4) is a consequence of Equations (4.1)-(4.3), (4.6) and (4.7). So, in what follows we may drop this equation and replace \hat{n} by $\theta - \hat{x} - \hat{y} - \hat{z}$ in (4.7).

We also note that since the velocity field is radially symmetric, we have

$$u(0,t) = 0. (4.10)$$

To completely pose this free boundary problem, we impose the following initial conditions,

$$\begin{aligned} R(0) & \text{ is prescribed,} \\ \hat{x}(r,0) &= \hat{x}_0(r), \hat{y}(r,0) = \hat{y}_0(r), \hat{z}(r,0) = \hat{z}_0(r), \hat{v}(r,0) = \hat{v}_0(r), \\ & \text{ where } \hat{x}_0(r), \hat{y}_0(r), \hat{z}_0(r) \text{ and } \hat{v}_0(r) \text{ and are nonnegative} \end{aligned}$$
(4.11)

functions with $\hat{x}_0(r) + \hat{y}_0(r) + \hat{z}_0(r) \le \theta$, for $0 \le r \le R(0)$.

In next section, we will transform free boundary problem (4.1)-(4.2) into a problem in a fixed region.

4.3 Model transformation

We first introduce the variables

$$\tilde{x} = \frac{\hat{x}}{\theta}, \quad \tilde{y} = \frac{\hat{y}}{\theta}, \quad \tilde{z} = \frac{\hat{z}}{\theta}, \quad \tilde{v} = \frac{\hat{v}}{\theta N}, \quad \tilde{u} = u$$

and the quantity

$$p_0 = \frac{\beta N \theta}{\gamma}, \quad \sigma_0 = \frac{\sigma}{\theta^{1-\nu}}.$$

The parameter p_0 is called the *basic reproductive ratio* in the epidemic models.

It represents the mean number of virus particles released by one virus.

Then, we introduce the change of variables:

$$\begin{split} \rho &= \rho(r,t) = r/R(t), \\ x(\rho,t) &= \tilde{x}(r,t), \quad y(\rho,t) = \tilde{y}(r,t), \quad z(\rho,t) = \tilde{z}(r,t), \\ v(\rho,t) &= \tilde{v}(r,t), \quad u(\rho,t) = \tilde{u}(r,t)/R(t), \end{split}$$

In terms of new variables, Eqs. (4.1)-(4.3), (4.5) and (4.7) take the following form in $\{0 < \rho < 1, t > 0\}$:

$$\begin{aligned} \frac{\partial x}{\partial t} &+ \left[u(\rho, t) - \rho u(1, t) \right] \frac{\partial x}{\partial \rho} \\ &= \lambda x - p_0 \gamma x v - \kappa D(t) x - \left[-\mu + (\lambda + \mu) x + \mu y + \mu z \right] x, \quad (4.12) \\ \frac{\partial y}{\partial t} &+ \left[u(\rho, t) - \rho u(1, t) \right] \frac{\partial y}{\partial \rho} \\ &= p_0 \gamma x v - \delta y - \kappa D(t) y - \left[-\mu + (\lambda + \mu) x + \mu y + \mu z \right] y, \quad (4.13) \\ \frac{\partial z}{\partial t} &+ \left[u(\rho, t) - \rho u(1, t) \right] \frac{\partial z}{\partial \rho} \\ &= \kappa D(t) (x + y) - \sigma_0 z^{\nu} - \left[-\mu + (\lambda + \mu) x + \mu y + \mu z \right] z, \quad (4.14) \end{aligned}$$

4.4 Constant Equilibrium Solutions 85

$$\frac{\partial v}{\partial t} - \frac{d}{R^2(t)} \frac{1}{\rho^2} \frac{\partial}{\partial \rho} \left(\rho^2 \frac{\partial v}{\partial \rho} \right) - \rho u(1,t) \frac{\partial v}{\partial \rho} = \delta y - \gamma v, \qquad (4.15)$$

$$u(\rho,t) = \frac{1}{\rho^2} \int_0^{\rho} s^2 \big[-\mu + (\lambda + \mu)x(s,t) + \mu y(s,t) + \mu z(s,t) \big] ds.$$
(4.16)

The boundary and initial conditions (4.8)-(4.9) and (4.2) become

$$\dot{R}(t) = R(t)u(1,t), \quad R(0) \text{ is given}, \tag{4.17}$$

$$v_{\rho}(0,t) = v_{\rho}(1,t) = 0, \qquad (4.18)$$

$$x(\rho, 0) = x_0(\rho), \ y(\rho, 0) = y_0(\rho), \ z(\rho, 0) = z_0(\rho), \ v(\rho, 0) = v_0(\rho), (4.19)$$

$$x_0(\rho), y_0(\rho), z_0(\rho) \ge 0, \qquad x_0(\rho) + y_0(\rho) + z_0(\rho) \le 1.$$
 (4.20)

We also assume that,

$$x_0(\rho), y_0(\rho), z_0(\rho) \text{ and } v_0(\rho) \text{ belong to } C^1[0,1], \text{ and } \frac{\partial v_0}{\partial \rho}(1) = 0, \quad (4.21)$$

$$D(t) \in C[0,\infty), \ D(t) \ge 0, \ \overline{D} = \lim_{t \to \infty} D(t).$$
 (4.22)

Here the absorbed radiation dose D(t) is proportional to the cumulative activity (so we assume that it is increasing in t), and we assume that it has a maximum value \overline{D} .

The global existence and uniqueness of the solutions to the system (4.12)-(4.22) was proved by Tao and Guo [146]. We do not present the proof here, instead, we will focus our attention on finding an explicit parameter condition for success of therapy (i.e., $R(t) \rightarrow 0$ as $t \rightarrow \infty$).

4.4 Constant Equilibrium Solutions

To find a parameter condition for $R(t) \rightarrow 0$, we first discuss the constant equilibrium solutions of (4.12)-(4.15). These solutions are determined by the

equations,

$$\left[(\lambda + \mu - \kappa \overline{D}) - (\lambda + \mu)x - (p_0 \delta + \mu)y - \mu z \right] x = 0, \qquad (4.23)$$

$$\left[(\delta + \kappa \overline{D} - \mu) + (\lambda + \mu - p_0 \delta) x + \mu y + \mu z \right] y = 0, \qquad (4.24)$$

$$\kappa \overline{D}(x+y) - \sigma_0 z^{\nu} - \left[-\mu + (\lambda+\mu)x + \mu y + \mu z\right]z = 0, \qquad (4.25)$$

$$\delta y - \gamma v = 0. \tag{4.26}$$

Clearly, $(x_s, y_s, z_s, v_s) = (0, 0, 0, 0) := E^0$ is a trivial equilibrium solution. In the following we try to find non-zero equilibrium solutions. Throughout this section, we assume

$$\kappa \overline{D} > \lambda + \mu, \tag{4.27}$$

which is the maximal rate condition for radiation damage.

Theorem 4.1. Under the assumption (4.27), we have

(*i*) *if*

$$\frac{(1-\nu)^{1-\nu}}{(2-\nu)^{2-\nu}} < \frac{\sigma_0}{\mu},\tag{4.28}$$

there is no non-zero equilibrium solutions;

(ii) if

$$\frac{(1-\nu)^{1-\nu}}{(2-\nu)^{2-\nu}} = \frac{\sigma_0}{\mu},\tag{4.29}$$

there exists unique non-zero equilibrium solution $(x_s, y_s, z_s, v_s) = (0, 0, z_s^{(0)}, 0) := E_0$ where $z_s^{(0)} = \frac{1-\nu}{2-\nu}$; (iii) if

$$\frac{(1-\nu)^{1-\nu}}{(2-\nu)^{2-\nu}} > \frac{\sigma_0}{\mu},\tag{4.30}$$

there exist two non-zero equilibrium solutions $(x_s^{(1)}, y_s^{(1)}, z_s^{(1)}, v_s^{(1)}) = (0, 0, z_s^{(1)}, 0) := E_1$ and $(x_s^{(2)}, y_s^{(2)}, z_s^{(2)}, v_s^{(2)}) = (0, 0, z_s^{(2)}, 0) := E_2$ where $0 < z_s^{(1)} < z_s^{(0)} < z_s^{(2)} < 1$.

Proof. The proof was given by Tao and Guo [146]. We distinguish four cases.

Case 1: x = 0, y = 0.

From (4.25), the equilibrium solutions satisfy

$$\sigma_0 z^{\nu} + (-\mu + \mu z)z = 0.$$

Noting $z \neq 0$ in this case for non-zero equilibrium solutions, this equation can be rewritten as

$$\frac{\sigma_0}{\mu} + (-1+z)z^{1-\nu} = 0. \tag{4.31}$$

Define

$$f(z) = \frac{\sigma_0}{\mu} + (-1+z)z^{1-\nu}, \quad 0 \le z \le 1.$$
(4.32)

We easily check that

$$f(0) = f(1) = \frac{\sigma_0}{\mu} > 0,$$

$$f'(z) = (2 - \nu)z^{-\nu} \left(z - \frac{1 - \nu}{2 - \nu}\right),$$

$$f'(z_s^{(0)}) = 0 \quad \text{where } z_s^{(0)} = \frac{1 - \nu}{2 - \nu},$$
(4.33)

$$f'(z) < 0 \quad \text{for } 0 < z < z_s^{(0)},$$
 (4.34)

$$f'(z) > 0 \quad \text{for } z_s^{(0)} < z \le 1,$$
 (4.35)

$$f(z_s^{(0)}) = \min_{0 \le z \le 1} f(z) = \frac{\sigma_0}{\mu} - \frac{(1-\nu)^{1-\nu}}{(2-\nu)^{2-\nu}}.$$

From these facts we easily get the conclusions in Theorem 4.1.

Case 2: $x = 0, y \neq 0$.

From (4.24) we have that the non-zero equilibrium solutions must satisfy

$$(\delta + \kappa \overline{D} - \mu) + \mu y + \mu z = 0,$$

which does not exist non-negative solution (y_s, z_s) by assumption (4.27).

Case 3: $x \neq 0, y = 0$.

This case can be discussed similarly to Case 2.

Case 4: $x \neq 0, y \neq 0$.

From (4.23) and (4.27), we easily check that there do *not* exist non-zero equilibrium solutions in this case. \Box

4.5 Stability of Equilibrium Solutions

For simplicity, throughout this section we assume that $D(t) \equiv \overline{D}$. We denote the right-hand sides of Eqs. (4.12)-(4.14) by $g_1(x, y, z, v)$, $g_2(x, y, z, v)$ and $g_3(x, y, z, v)$, respectively.

We begin with considering the trivial equilibrium solution $E^0 = (x_s, y_s, z_s, v_s)$ = (0, 0, 0, 0). By simple calculation, the linearized coefficient matrix A^0 of the right-hand terms of hyperbolic system (4.12)-(4.14) with $\nu = 1$ at equilibrium point E^0 has the form:

$$A^{0} = \begin{pmatrix} \frac{\partial g_{1}}{\partial x} & \frac{\partial g_{1}}{\partial y} & \frac{\partial g_{1}}{\partial z} \\ \frac{\partial g_{2}}{\partial x} & \frac{\partial g_{2}}{\partial y} & \frac{\partial g_{2}}{\partial z} \\ \frac{\partial g_{3}}{\partial x} & \frac{\partial g_{3}}{\partial y} & \frac{\partial g_{3}}{\partial z} \end{pmatrix}_{E^{0}} = \begin{pmatrix} -(\kappa \overline{D} - \lambda - \mu) & 0 & 0 \\ 0 & -(\kappa \overline{D} + \delta - \mu) & 0 \\ 0 & 0 & -\sigma_{0} + \mu \end{pmatrix}_{E^{0}}$$

We have

Theorem 4.2. In addition to the assumption (4.27), we also assume that

$$\sigma_0 > \mu. \tag{4.36}$$

Then, the trivial equilibrium solution E^0 of the system (4.12)-(4.22) with $\nu =$ 1 is locally stable.

Proof. Under the assumptions (4.27) and (4.36), the three eigenvalues of matrix A^0 are negative. Therefore, the equilibrium solution E^0 to system (4.12)-(4.22) with $\nu = 1$ is *linearly* stable. Furthermore, proceeding as in the proofs of Lemma 6.2, Theorem 7.1 and Theorem 8.1 in [61], we can obtain the locally *nonlinear* stability of equilibrium solution E^0 to the system (4.12)-(4.22) with $\nu = 1$. \Box

Remark 4.1. The stability of the trivial equilibrium solution E^0 of the system (4.12)-(4.22) for general $0 < \nu < 1$ remains open, because the function $\partial g_3/\partial z$ has singularity at E^0 ($\frac{\partial g_3}{\partial z}|_{E^0} = \infty$). The stability of the equilibrium solution E^0 corresponds to success of the therapies. However, the instability of the equilibrium solution E^0 may imply the recurrence of a tumor.

We next turn to study the stability of the non-zero equilibrium solutions $E_i \ (i = 0, 1, 2)$. It is easy to check that

$$\begin{split} &\frac{\partial g_1}{\partial x}\big|_{E_i} = \lambda + \mu - \kappa \overline{D} - \mu z_s^{(i)}, \quad \frac{\partial g_1}{\partial y}\big|_{E_i} = 0, \quad \frac{\partial g_1}{\partial z}\big|_{E_i} = 0, \\ &\frac{\partial g_2}{\partial x}\big|_{E_i} = 0, \quad \frac{\partial g_2}{\partial y}\big|_{E_i} = \mu - \delta - \kappa \overline{D} - \mu z_s^{(i)}, \quad \frac{\partial g_2}{\partial z}\big|_{E_i} = 0, \\ &\frac{\partial g_3}{\partial x}\big|_{E_i} = \kappa \overline{D} - (\lambda + \mu) z_s^{(i)}, \quad \frac{\partial g_3}{\partial y}\big|_{E_i} = \kappa \overline{D} - \mu z_s^{(i)}, \\ &\frac{\partial g_3}{\partial z}\big|_{E_i} = -\sigma_0 \nu \left(z_s^{(i)}\right)^{\nu - 1} + \mu - 2\mu z_s^{(i)}. \end{split}$$

Hence, the linearized coefficient matrix A_i (i = 0, 1, 2) of the right-hand sides of the hyperbolic system (4.12)-(4.14) at corresponding equilibrium point E_i has the form:

$$A_{i} = \begin{pmatrix} \lambda + \mu - \kappa \overline{D} - \mu z_{s}^{(i)} & 0 & 0 \\ 0 & \mu - \delta - \kappa \overline{D} - \mu z_{s}^{(i)} & 0 \\ \kappa \overline{D} - (\lambda + \mu) z_{s}^{(i)} & \kappa \overline{D} - \mu z_{s}^{(i)} & -\sigma_{0} \nu (z_{s}^{(i)})^{\nu - 1} + \mu - 2\mu z_{s}^{(i)} \end{pmatrix}.$$
(4.37)

Using the assumption (4.27) and noting $z_s^{(i)} > 0$ (i = 0, 1, 2), we easily find that

$$\lambda + \mu - \kappa \overline{D} - \mu z_s^{(i)} < 0, \tag{4.38}$$

$$\mu - \delta - \kappa \overline{D} - \mu z_s^{(i)} < 0. \tag{4.39}$$

Define

$$h(z) \equiv g_3(0,0,z,0) = -\sigma_0 z^{\nu} + (\mu - \mu z)z.$$
(4.40)

That is,

$$h(z) = -\mu z^{\nu} f(z), \tag{4.41}$$

where f(z) is defined by (4.32). Clearly, by (4.40) we have

$$h'(z) = -\sigma_0 \nu z^{\nu-1} + \mu - 2\mu z. \tag{4.42}$$

On the other hand, by (4.41) we get

$$h'(z) = -\mu\nu z^{1-\nu} f(z) - \mu z^{\nu} f'(z).$$
(4.43)

In the following we assume that condition (4.30) holds. Using (4.34), (4.35), (4.43) and Theorem 4.1 and noting $f(z_s^{(1)}) = f(z_s^{(2)}) = 0$, we have

$$h'(z_s^{(1)}) = -\mu(z_s^{(1)})^{\nu} f'(z_s^{(1)}) > 0, \qquad (4.44)$$

$$h'(z_s^{(2)}) = -\mu(z_s^{(2)})^{\nu} f'(z_s^{(2)}) < 0.$$
(4.45)

From (4.37), (4.43) and (4.44) we find that matrix A_1 has a *positive* eigenvalue. Therefore, arrive at the following theorem.

Theorem 4.3. Under assumptions (4.27) and (4.44), the non-zero equilibrium solution E_1 is linearized unstable.

By (4.37)-(4.39), (4.43) and (4.45) we find that under the assumptions (4.27) and (4.30), all eigenvalues of matrix A_2 are negative, and therefore the non-zero equilibrium solution E_2 is *linearly* stable. Furthermore, proceeding as in the proofs of Lemma 6.2, Theorem 7.1 and Theorem 8.1 in [61], we can get the locally *nonlinear* stability of the non-zero equilibrium solution E_2 . More precisely, we have

Theorem 4.4. Assume that (4.27) and (4.30) hold. If

$$||x_0(\rho), y_0(\rho), z_0(\rho) - z_s^{(2)}, v_0(\rho)||_{C^1[0,1]} \le \varepsilon$$

where ε is sufficiently small, then the global solution $(x(\rho, t), y(\rho, t), z(\rho, t), v(\rho, t), u(\rho, t), R(t))$ satisfies

$$|x(\xi(t),t)|, \ |y(\xi(t),t)|, \ |z(\xi(t),t) - z_s^{(2)}|, \ |v(\xi(t),t)| \le C\varepsilon e^{-\eta t} \quad for \ all \ t > 0$$

where C and η are some positive constants, and $\xi(t) = \xi(t; \rho_0)$ is the forward characteristic curve of the first-order hyperbolic equation (4.12) satisfying $\rho_0 = \xi(0; \rho_0).$

We are now in a position to study the convergence of the tumor radius R(t) to zero, which corresponds to the success of therapies. In fact, we have

Theorem 4.5. Assume that (4.27) and (4.30) hold. If

$$||x_0(\rho), y_0(\rho), z_0(\rho) - z_s^{(2)}, v_0(\rho)||_{C^1[0,1]} \le \varepsilon$$

where $\varepsilon < \mu(1-z_s^{(2)})/2$, then we have

$$R(t) \rightarrow 0$$
 exponentially.

Proof. (4.16) and (4.17) yield

$$\frac{\dot{R}(t)}{R(t)} = \int_0^1 s^2 \big[-\mu + (\lambda + \mu)x(s, t) + \mu y(s, t) + \mu z(s, t) \big] ds.$$

This, together with Theorem 4.4, yields

$$\begin{aligned} \frac{\dot{R}(t)}{R(t)} &< \int_0^1 s^2 \big(-\mu + \mu z_s^{(2)} + \varepsilon \big) ds \\ &= \frac{1}{3} \big[-\mu (1 - z_s^{(2)}) + \varepsilon \big] \\ &< -\frac{1}{6} \mu (1 - z_s^{(2)}) \quad \text{for sufficiently large } t. \end{aligned}$$

Thus,

$$R(t) < R(0)e^{-\frac{1}{6}\mu(1-z_s^{(2)})t} \to 0$$
 exponentially.

Remark 4.2. We conclude from Theorems 4.4 and 4.5 that the non-zero equilibrium point E_2 corresponds to a successful therapy, which has not been found in [51]. Hence, the condition (4.27) and (4.30) is a new explicit parameter condition for successful eradication of a tumor.

4.6 Possible Optimal Protocols

In this section we will numerically explore possible optimal therapy strategies. The typical parameter values for the numerical simulation are $\lambda = 0.0086$, $\delta = 0.0293$, $\gamma = 0.0119$, $\nu = 0.176$, $\sigma = 18.57/24$, d = 25/6, $\mu = 1/72$, $p_0 = 3.73$, $\theta = 10^6$ given in [146, 179, 117].



Fig. 4.1. The comparison between the effect of radiovirotherapy with $(x_0, y_0, v_0, z_0) = (0.01, 0.001, 0.1, 0.02), D = 0.01$ and the effect of virotherapy alone with $(x_0, y_0, v_0) = (0.01, 0.001, 0.1), D = 0, z(t) \equiv 0.$

Fig.4.1 shows that radiovirotherapy is more effective than virotherapy alone. Fig.4.1 clearly indicates that the combination of virotherapy with radiation (radiovirotherapy) may reduce tumors when virotherapy alone failed. This has been verified by experimental research [50]. The mathematical model and numerical simulation in [146] give us a deeper understanding of the design of radiovirotherapy strategy.



Fig. 4.2. The effects of the different timing of radio-iodine administration on tumor growth. Take $(x_0, y_0, v_0) = (0.01, 0.001, 0.1), D = 0.01 (t > t_r), z|_{t=t_r} = 0.02, z(t) \equiv 0 \text{ and } D = 0 \text{ for } 0 \leq t < t_r.$

Fig.4.2 shows the effects of the different timing of radio-iodine administration on tumor growth. If we take $(x_0, y_0, v_0) = (0.01, 0.001, 0.1), D = 0.01$ ($t > t_r$), $z|_{t=t_r} = 0.02, z(t) \equiv 0$ and D = 0 for $0 \le t < t_r$ (where t_r is the start timing of radio-iodine administration), then Fig.4.2 shows that there exists an optimal timing $t_r^{opt} \approx 300$ (hours) for radio-iodine administration. Earlier or later administrations of iodine result in a larger size of the tumor. Furthermore, our numerical simulations (not presented here) also show that the optimal timing t_r^{opt} depends on the dose D of the radioactive iodide. The optimal timing $t_r^{opt} = 0$ for small D and large D.



Fig. 4.3. The effects of the different doses of radioactive iodide on tumor growth. Take $(x_0, y_0, v_0, z_0) = (0.01, 0.001, 0.1, 0.02), t_r = 0.$

Fig.4.3 shows the effects of the different doses of radioactive iodide on tumor growth. If we take $(x_0, y_0, v_0, z_0) = (0.01, 0.001, 0.1, 0.02)$ and $t_r = 0$, then Fig.4.3 shows that there exists an optimal dose $D^{opt} \approx$ 0.005 of the radioactive iodide. Fig.4.3 shows that the dose of radio-iodine administration is another important issue for radiovirotherapy. It suggests

that given *intermediate* dose of radioactive iodide for treatment may be an optimal therapy strategy. Another reason why the dose of radioactive iodide given for treatment is important is that radiation damages the health tissue. Therefore, the optimal dose of iodine for treatments still needs to be further experimentally investigated.

The numerical simulations lead to the design of possible optimal therapeutic strategies. The timing of radio-iodine administration and the dose of iodine are two critical factors for the efficacy of a combined treatment of viral and radio therapy. The numerical results of this section may be helpful for experimental research.

4.7 Summary and Discussion

The combination of virotherapy with radiotherapy (radiovirotherapy) is more effective than treatments with virotherapy alone, which has been supported by experimental data. Radiovirotherapy is a very complex and sensitive dynamical system. To better understand its outcome, mathematical modeling may play a role. The PDE model [146] reviewed in this chapter is a generalization of the previously existing ODE model [51]. And it is also a generalization of the previously existing PDE model [61], which is a model of virotherapy. This chapter reviews the modeling of a combined action of viral and radio therapy, which allows us to study the optimal strategies for treatments.
Under some appropriate assumptions on model parameters, we found that there are two non-zero equilibrium solutions $E_1 = (x_s^{(1)}, y_s^{(1)}, z_s^{(1)}, v_s^{(1)}) =$ $(0, 0, z_s^{(1)}, 0)$ and $E_2 = (x_s^{(2)}, y_s^{(2)}, z_s^{(2)}, v_s^{(2)}) = (0, 0, z_s^{(2)}, 0)$ with $0 < z_s^{(1)} <$ $z_s^{(2)} < 1$. The non-zero equilibrium solution E_1 is linearly unstable. However, we proved the locally nonlinear stability of the non-zero equilibrium solution E_2 , which corresponds to a successful therapy of a tumor.

The numerical simulations in Section 4.6 verify that radiovirotherapy is more effective than treatments with virotherapy alone. These simulations also suggest that there is an optimal timing of radio-iodine administration and an optimal dose of the radioactive iodide, which need to be further experimentally tested.

ONYX-015, a genetically modified adenovirus, is one of oncolytic viruses that have been tested in clinical trials. Studies in clinical trials have shown that the expression of the coxsackie-adenovirus receptor (CAR) strongly influences the entry of virus into cancer cells (for example, see [10]). Mitogen-activated protein kinase kinase (MEK, also known as MAP-kinase kinase) inhibitors have been shown to promote CAR expression, and could result in increased ONYX-015 entry into target cells (see [183] and references therein). This could lead to a novel combined therapeutic approach to cancer, using ONYX-015 and MEK inhibitors. However, MEK inhibitors can cause temporary cell-cycle arrest, which inhibits the life-cycle of ONYX-015 (the cell cycle is generally considered to consist of four phases of proliferate growth: the growth phase (G_1) , a phase of DNA synthesis (S), a period before cell division (G_2) and

98 4 Modeling Cancer Radiovirotherapy

mitosis (M); and a single phase of quiescent behavior (G_0). ONYX-015 needs to lock the cell in S-phase to replicate and lyse the cell). So, MEK inhibitors may limit the replication of viruses. To design an effective protocol of combined therapies against cancer using ONYX-015 and MEK inhibitors, the positive effect of MEK inhibitors should be optimally balanced with the negative effect of MEK inhibitors. This complicates the dynamics of MEK inhibitors, viruses and tumor cells. Zurakowski and Wodarz [183] initially introduced an ODE model to study the effects of MEK inhibitors and viruses on tumor cells. They used their model to explore the reduction of the tumor size, and the tumor size reduction can be achieved by the combined therapies. Tao and Guo [148] extended Zurakowski and Wodarz's model to a PDE model and used the PDE model to analytically and numerically explore a possible optimal dose and the optimal timing of MEK inhibitors.

Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma

5.1 Introduction

Glioblastoma multiforme, a type of glioma, is the most aggressive of brain tumors. The life expectancy from the time when it is diagnosed is typically one year. The current treatment is surgical resection followed by radiotherapy and chemotherapy. There are only a few consistent clinical studies which compare life expectancies of patients who underwent different resections (residual or complete) and different protocols of radiotherapy and chemotherapy. Among the most consistent studies are those of Albert et al. [3], Lacroix et al. [107], and Stupp et al. [137].

A detailed study of 135 patient data by Albert et al. [3] showed that patients who underwent subtotal surgery postoperatively had 6.6 times higher risk of death in comparison to patients who underwent complete resection, and patients treated by radiotherapy had 0.26 times lower risk of death in comparison to patients who were not treated with radiation. Lacroix et al. [107] analyzed 416 patients data and showed that a significant survival 100 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma advantage was associated with resection of 98% or more of the tumor volume, and generally, gross total tumor resection led to longer life expectancy.

The efficacy of chemotherapy has been steadily improved with the development of new cancer drugs. Stupp et al. [137] analyzed the data of 573 patients and showed that the median survival time (MST) was 14.6 months for patients who underwent radiotherapy plus chemotherapy with temozolomide, but only 12.1 months for those with radiotherapy alone.

All these clinical data analysis are retrospective. They have value for reference, but they are likely quite biased in nature, and cannot give any perspective prediction. Tian et al. [160] developed a mathematical model which integrates the treatment of patients by surgery, radiotherapy and chemotherapy. The model parameters were chosen in Tian et al. [160] so that the simulation results fit with the patient data analysis reported in [3, 107, 137]. The Study of the model in [160] suggested a combination of treatment protocols that can give patients maximal survival time. This chapter reviews the mathematical model proposed by Tian et al [160].

5.2 Mathematical Model and Clinical Data

The mathematical model describes a spherical tumor regrowing after surgical resection. The tumor contains tumor stem cells x and necrotic cells y. The quantity x represents the number density of tumor cells (i.e., the number of tumor cells in 1 mm^3), the quantity y represents the number density of

necrotic cells. It is assumed that the number density of cells in a tumor is a constant [178], that is, x + y = number of cells in 1 mm^3 , which is 10⁶ [119]. New tumor cells are produced by proliferation, and they transit to necrotic cells by lysis.

Tumor cells that are near to the expanding surface of the solid tumor receive more nutrients and proliferate faster than tumor cells that are near the core of the tumor. Indeed, as mentioned in [127], the proportion of proliferating cells varies considerably from the outer region to the inner region of the tumor. For simplicity we assume that the proliferation rate, λ , is constant. According to [68], $\lambda = 2 \times 10^{-2} h^{-1}$. We shall also assume the rate of cells become necrotic, δ , is constant. We take δ to be slightly smaller that λ , namely, $\delta =$ $1.89 \times 10^{-2} h^{-1}$. According to [67], necrotic cells are removed on the average of 2–3 days. We shall take the removal rate μ to be $1/72 h^{-1}$.

According to [107], the median preoperative tumor volume was 34 cm^3 . If we assume that the tumor is spherical, then this corresponds to radius at resection time of $R_0 = 20 \ mm$. In the partial resection case, a smaller ball of radius R_* is removed, and residual tumor cells remain in the region between the two concentric balls. After surgery the ball of radius R_* fills with cerebro-spinal fluid, and the residual tumor begins to grow outward, as illustrated schematically in Fig. 5.1. From the rate λ of tumor cell proliferation and the reported life expectancy, we estimate that the patient dies at the time when the tumor radius reaches 40 mm. This estimate is confirmed in [107].

102 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma



Fig. 5.1. The inner ball is the surgically removed part of the tumor, the dark concentric ball is the residual tumor, and the outer shell is the regrowth part of the tumor.

Consider a radially symmetrical tumor and denoted by r the distance from a point to the origin. We denote the boundary of the tumor by r = R(t).

The proliferation and removal of cells cause a movement of cells within the tumor, with a convection term, for tumor cells x, in the form $\frac{1}{r} \frac{\partial}{\partial r} [r^2 u(r,t) x(r,t)]$, where u(r,t) is the radial velocity field, and $u(R_*,t) = 0$ since the tumor does not grow inward. By the conservation law of mass, we have the equation for tumor cell population,

$$\frac{\partial x(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 u(r,t) x(r,t) \right) = \lambda x(r,t) - \delta x(r,t).$$
(5.1)

Similarly, we have the for necrotic cell population,

$$\frac{\partial y(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 u(r,t) y(r,t) \right) = \delta x(r,t) - \mu y(r,t).$$
(5.2)

As mentioned before, the total number density of tumor cells is constant, that is, $x(r,t) + y(r,t) = const. = \theta$, and $\theta = 10^6/mm^3$ [119]. Adding Eqs. (5.1) and (5.2) together, we obtain the equation for the radial velocity field:

$$\frac{\theta}{r^2}\frac{\partial}{\partial r}(r^2u) = (\lambda + \mu)x(r, t) - \mu\theta.$$
(5.3)

The tumor radius evolves according to

$$\frac{dR}{dt} = u(R(t), t). \tag{5.4}$$

We assume that,

$$x(r,0) = \frac{9}{10}\theta, \quad \text{for } R_* \le r \le R_0.$$
 (5.5)

That is, initially 90% of cells are tumor (stem) cells, and 10% are necrotic cells.

We need to solve Eqs. (5.1), (5.3) in $R_* \leq r \leq R(t)$ with the initial condition (5.5) and with the tumor growth condition (5.4). The above model does not include radiotherapy and chemotherapy yet. According to [137], within 6 weeks after the histologic diagnosis of glioblastoma, patients were assigned to receive standard focal radiotherapy alone or standard radiotherapy plus concomitant daily temozolomide followed by adjuvant temozolomide, whether or not they had previously undergone debulking surgery. The standard radiotherapy consists of fractionated focal irradiation at a dose of 2 Gy per fraction given daily, 5 days per week (Monday through Friday), over a period of 6 weeks, for a total dose of 60 Gy. Accordingly, we take the radiation activity function to be 104 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma

$$\rho(t) = \begin{cases}
1 & \text{if } 6 \le t \le 12, \\
0 & \text{otherwise.}
\end{cases}$$

We assume that the radiation kills tumor cells at a rate A, so that the death rate by radiotherapy is $A\rho(t)$. For simplicity, we lump together the cells killed by radiation with necrotic cells.

Chemotherapy we consider is administration of temozolomide at a dose of 75 mg per square meter of body surface per day, given 7 days a week from the first day of radiation until the last day of radiation. Then, after a 4-week break, chemotherapy continues, and patients receive a double dose of temozolomide daily for 28 days. After the end of this period, another cycle of temozolomide dosing is administered at $\frac{8}{3}$ level of the original dose, that is 200 mg per square meter. We therefore introduce the temozolomide dosing function as

$$\tau(t) = \begin{cases} 1 & \text{if } 6 \le t \le 12, \\ 2 & \text{if } 16 \le t \le 20, \\ \frac{8}{3} & \text{if } 20 \le t \le 40, \\ 0 & \text{otherwise.} \end{cases}$$

If chemotherapy with 75 mg dose kills tumor cells at a rate B, then the killing rate by chemotherapy treatment is $B\tau(t)$.

If the standard radiotherapy is administered over a period of 6 weeks during the time period $6 \le t \le 12$ and the temozolomide is given for 40 weeks. Eqs. (5.1) and (5.2) are replaced by 5.2 Mathematical Model and Clinical Data 105

$$\frac{\partial x(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 u(r,t) x(r,t) \right)$$

$$= \lambda x(r,t) - \delta x(r,t) - A\rho(t) x(r,t) - B\tau(t) x(r,t), \qquad (5.6)$$

$$\frac{\partial y(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 u(r,t) y(r,t) \right)$$

$$= \delta x(r,t) + A\rho(t)x(r,t) + B\tau(t)x(r,t) - \mu y(r,t).$$
(5.7)

By adding this two equations together, we obtain the same Eq. (5.3), as before, for the velocity field u(r, t).

In next section we will numerically solve Eqs. (5.6), (5.3) in $R_* \leq r \leq R(t)$ together with (5.4) and (5.5).

Albert et al. [3] provide the median survival time (MST) for various age groups of patients. For definiteness we consider the group of patients between the age of 20 and 39. This group is further divided into three subgroups [3]:

(a) Patients had complete resection (no residual tumor) and undergone radiotherapy; their MST was 92 weeks.

(b) Patients had partial resection (with residual tumor) and undergone radiotherapy; their MST was 46 weeks.

(c) Patients had partial resection (with residual tumor) without radiotherapy; their MST was 15 weeks.

A recent study by Stummer et al. [136] of 243 patients compares the MST of patients who underwent complete resection versus subtotal (partial) resection while both groups received radiation therapy. Repeat surgery and /or initiation of chemotherapy were applied to some patients after tumor progression. The MST was 71 weeks for the first group and 49 weeks for 106 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma the second group. By contrast with the data in [3] quoted above (in (a) and (b)) the study in [136] lumps together all ages ≤ 60 ; these data may indicate that older patients may not do as well as younger patients undergoing complete resection. Studies on the effect of different modes of radiation (without distinguishing between complete and subtotal resection) are reported in [25, 54, 139, 171].

It is commonly believed that by the time glioblastoma is diagnosed, some cancer cells have already migrated from the main body of the tumor. Thus, even when resection is complete there are residual tumor cells in the vicinity of the tumor. The model [160] accounts for these cells by defining complete resection to be the removal of not all the ball of radius R_0 , but of a slightly smaller ball of radius $R_0 - \varepsilon$. We take $\varepsilon = 5 \ \mu m$, half the size of a typical cell, thereby making implicit assumption that glioma cells in the thin shell $R_0 - \varepsilon \leq r \leq R_0$ are in 'migration mode' from the solid tumor.

In the next section we use data from [3, 107, 137] to determine the parameter R_* corresponding to partial resection and the parameters A and ε .

5.3 Parameter Estimation

In all the numerical simulations discussed below we assume that the initial density of tumor cells in the shell $R_* \leq r \leq R_0$, in nine times higher than the density of the necrotic cells. We first use numerical simulations to determine what the partial resection means mathematically. That is, in order to call

a surgery is partial resection, how much should the tumor be removed? or what is R_* ? Second, we determine the radiotherapy killing rate A. Third, we determine what the complete resection means, $R(0) = R_0 - \varepsilon$, what is ε ? After we obtain these parameter values, we can use our model to study some treatment protocols.



Fig. 5.2. Partial resection $R_* = 18$ mm: residual tumor regrowth without radiotherapy and chemotherapy

Fig. 5.2 shows the growth of the tumor radius R(t) without any therapy if $R_* = 18 \ mm$ (partial resection). The time T at which R(T) becomes 40 mm, that is the survival time, is approximately 15 weeks, as reported for the subgroup (c) above. This agreement validates our choice of R_* . We notice that the initial growth of the tumor is extremely fast. Thus although 108 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma the simulation begins with tumor radius of 18 mm, the radius very quickly arises to over 20 mm. The same holds for the subsequent figures.



Fig. 5.3. Partial resection $R_* = 18$ mm: residual tumor regrowth with radiotherapy only, at regular strength (A = 1.0)

Fig. 5.3 shows the growth of the tumor radius R(t) if we take in the model the radiation killing rate A = 1.0 and partial resection $R_* = 18 \ mm$. We see that $R(T) = 40 \ mm$ at approximately T = 46, as reported for the subgroup (b) above. This validates the choice of the parameter value A = 1.0. Note that the tumor radius begins growing until the start of radiation. Radiation treatment decreases the radius, but as soon as radiation is stopped, the tumor begins to grow again.



Fig. 5.4. Complete resection $R(0) = R_0 - \varepsilon$, $\varepsilon = 5\mu m$, with radiotherapy only, at regular strength (A = 1.0)

Fig. 5.4 shows the growth of R(t) after the complete resection (that is, $R(0) = R_0 - \varepsilon$, where $\varepsilon = 5\mu m$) and radiotherapy. We see that R(T) =40 mm at T = 92 weeks which is in agreement with the MST reported for the subgroup (a). This agreement validates the choice of the parameter ε .

5.4 Mathematical Protocols

The mathematical model can be used to explore the effect of different radiation protocols and resections. The standard radiation is given for a period of 6 weeks. In Fig. 5.5 we see the result of giving the same total amount 60 Gy of radiation within 3 weeks instead of 6 weeks, and of giving the same total amount distributed over 12 weeks; both profiles are computed for the



Fig. 5.5. Radiation with $R_* = 18$ mm. Protocol 1 (dashed line): radiotherapy only, at double strength, half time. Protocol 2 (solid line): radiotherapy only, at half strength, double time

residual case. The survival time increases in the first case from 46 weeks to 50 weeks, and in the second case from 46 weeks to 49.5 weeks. Thus both procedure of 6 weeks. It should be pointed out that the use of 60 Gy is actually radiobiologically impertinent as it ignores overriding issues of normal cerebral toxicity and radioresistance, but is interesting to consider in the abstract.

The model can also be extended to explore the effect of chemotherapy. As in the case of radiotherapy, chemotherapy kills tumor cells at some rate B. Fig. 5.6 profiles R(t) in the residual tumor case. By choosing B = 0.03 we achieve survival time of 60 weeks as given in [137]. We see that chemotherapy has



Fig. 5.6. Residual tumor, $R_* = 18$ mm, regrowth with radiotherapy at regular strength (A = 1) and chemotherapy B = 0.03

very little benefits compared to radio therapy (i.e., B = 0.03 is much smaller than A = 1.0).

Our model can predict the benefits that will occur if the resection will be more complete, or if the radiation dose is increased. For example, Fig. 5.7 shows that resection with $R_* = 19$ mm followed by radiotherapy yields survival time of 52 weeks as compared to 46 weeks when $R_* = 18$ mm, and 92 weeks when resection is complete. Fig. 5.8 shows, in the case of residual resection ($R_* = 18$ mm), that if in the standard radiation treatment the amount of dose is doubled, then the MST will increase from 46 weeks to 80 weeks. But this of course does not take into account toxic side effects due to increased radiation.



Fig. 5.7. Residual tumor, $R_* = 19$ mm, regrowth with radiotherapy at regular strength (A = 1) but no chemotherapy

5.5 Explicit solutions

It is well known that most nonlinear free boundary problems are impossible to solve in terms of explicit analytical solutions. In contrast, the hyperbolic free boundary problem 5.1 - 5.7 is solvable, and the explicit solution is found by using the backward characteristic curve method [180]. An interesting finding is that the original free boundary problem can be reformulated as a fixed boundary problem defined on an infinite domain with discontinuous initial condition. To our knowledge, this analytical treatment of nonlinear free boundary problems is new, and was published in [180] recently. These solutions will not only confirm the numerical prediction in the previous sections, but



Fig. 5.8. Radiation protocol 3: $R_* = 18$ mm with radiotherapy only as standard treatment, but with double stength dosage

also shed light on further analysis for problems of this type towards better understanding the complicated phenomena of tumor growth.

Below we first transform this model for ease of analysis.

Notice $\alpha(t) := \lambda - \delta - A\rho(t) - B\tau(t)$. We will treat $\alpha(t)$ as step function or constant. We apply the following minor change of notations:

$$\tilde{x}(r,t) = x(r,t)/\theta$$
 $\beta = \lambda + \mu$,

and the system becomes:

114 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma

$$\begin{cases} \frac{\partial \tilde{x}(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial t} (r^2 u(r,t) \tilde{x}(r,t)) = \alpha \tilde{x}(r,t), & R_* \le r \le R(t), t \ge 0, \\ & \frac{1}{r^2} \frac{\partial}{\partial t} (r^2 u(r,t)) = \beta \tilde{x}(r,t) - \mu, & R_* \le r \le R(t), t \ge 0, \\ & u(R_*,t) = 0, \quad t \ge 0, \\ & \frac{dR(t)}{dt} = u(R(t),t), \quad t \ge 0, \end{cases}$$

with the initial conditions: $R(0) = R_0$, $\tilde{x}(r, 0) = c$, for $R_* \leq r \leq R(0)$. Here the second equation is also replaced by the sum of the first two equations.

To get rid of all r^2 or r terms, we introduce the "volume velocity":

$$v(r,t) := r^2 u(r,t)$$

The system then becomes:

$$\begin{cases} \frac{\partial \tilde{x}(r,t)}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial t} (v(r,t)\tilde{x}(r,t)) = \alpha \tilde{x}(r,t), & R_* \le r \le R(t), t \ge 0, \\ & \frac{1}{r^2} \frac{\partial v(r,t)}{\partial t} = \beta \tilde{x}(r,t) - \mu, & R_* \le r \le R(t), t \ge 0, \\ & v(R_*,t) = 0, \quad t \ge 0, \\ & R(t)^2 \frac{dR(t)}{dt} = v(R(t),t), \quad t \ge 0, \end{cases}$$

with the initial conditions: $R(0) = R_0$, $\tilde{x}(r, 0) = c$, for $R_* \leq r \leq R(0)$.

Then we apply the following change of variables:

$$s := r^3,$$

 $V(t) := R(t)^3,$
 $V_* := R_*^3,$
 $V_0 := R_0^3,$
 $\omega(s,t) := 3v(\sqrt[3]{s}, t).$

By abusing notations, we redefine "x" as function \tilde{x} in terms of variable s:

$$x(s,t) := \tilde{x}(\sqrt[3]{s},t).$$

Then we have:

$$\begin{split} \frac{\partial \omega(s,t)}{\partial s} &= 3 \frac{\partial}{\partial s} v(\sqrt[3]{s},t) = \frac{\partial v(r,t)}{\partial r} s^{-\frac{2}{3}} = v_r(r,t) r^{-2},\\ \frac{\partial x(s,t)}{\partial s} &= \frac{1}{3} \frac{\partial \tilde{x}(r,t)}{\partial r} s^{-\frac{2}{3}} = \frac{1}{3} \frac{\partial \tilde{x}(r,t)}{\partial r} r^{-2},\\ \frac{dV(t)}{dt} &= \frac{d}{dt} (R(t)^3) = 3R(t)^2 \frac{dR(t)}{dt} = 3v(R(t),t) = \omega(V(t),t) \end{split}$$

After the change of variables, the system becomes:

$$\begin{cases} \frac{\partial x(s,t)}{\partial t} + \frac{\partial}{\partial s}(\omega(s,t)x(s,t)) = \alpha x(s,t), \quad V_* \le s \le V(t), t \ge 0, \\\\ \frac{\partial \omega(s,t)}{\partial s} = \beta x(s,t) - \mu, \quad V_* \le s \le V(t), t \ge 0, \\\\ \omega(V_*,t) = 0, \quad t \ge 0, \\\\ \frac{dV(t)}{dt} = \omega(V(t),t), \quad t \ge 0, \end{cases}$$

with the initial conditions: $V(0) = V_0$, x(s,0) = c, for $V_* \leq s \leq V(0)$.

Substitute the second equation into the first one to obtain:

$$\frac{\partial x(s,t)}{\partial t} + \omega(s,t)\frac{\partial x(s,t)}{\partial s} = \beta x(s,t) \left(K - x(s,t)\right),$$

where $K := \frac{\alpha + \mu}{\beta}$, and K < 1.

Thus, we have transformed the original system into the following system:

$$\frac{\partial x(s,t)}{\partial t} + \omega(s,t)\frac{\partial x(s,t)}{\partial s} = \beta x(s,t) \left(K - x(s,t)\right), \quad V_* \le s \le V(t), t \ge 0,$$
$$\frac{\partial \omega(s,t)}{\partial s} = \beta x(s,t) - \mu, \quad V_* \le s \le V(t), t \ge 0,$$
$$\omega(V_*,t) = 0, \quad t \ge 0,$$
$$\frac{dV(t)}{dt} = \omega(V(t),t), \quad t \ge 0,$$
(5.8)

116 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma with the initial conditions: $V(0) = V_0$, $x(s, 0) = \varphi(s)$, for $V_* \leq s \leq V(0)$. Here a more general initial condition $\varphi(s)$ will be considered.

We define the "characteristic" curve of the first equation in the system (5.8) as follows,

$$\gamma = \gamma(s, t), \quad V_* \le s \le V_0, t \ge 0, \tag{5.9}$$

which satisfies:

$$\begin{cases} \frac{d\gamma(s,t)}{dt} = \omega(\gamma(s,t),t), \quad V_* \le s \le V_0, t \ge 0, \\ \gamma(s,0) = s, \quad V_* \le s \le V_0. \end{cases}$$
(5.10)

Since $\omega(s,t)$ is continuous in (s,t) and continuously differentiable in γ , these curves are uniquely defined, satisfying $V_* < \gamma(s,t) < V(t)$ for $V_* < s < V_0, t \ge$ 0, and $\gamma(V_*,t) = V_*, \ \gamma(V_0,t) = V(t)$ for $t \ge 0$. Setting $\hat{x}(s,t) := x(\gamma(s,t),t)$, $\hat{\omega}(s,t) := \omega(\gamma(s,t),t)$, the system (5.8) reduces to:

$$\begin{cases} \frac{\partial \hat{x}(s,t)}{\partial t} = \beta \hat{x}(s,t) \left(K - \hat{x}(s,t)\right), & V_* \le s \le V_0, t \ge 0, \\ \frac{\partial \hat{\omega}(s,t)}{\partial s} = \left(\beta \hat{x}(s,t) - \mu\right) \frac{\partial \gamma(s,t)}{\partial s}, & V_* \le s \le V_0, t \ge 0, \\ \frac{d \gamma(s,t)}{dt} = \hat{\omega}(s,t), & V_* \le s \le V_0, t \ge 0, \end{cases}$$
(5.11)

with the initial conditions $\hat{x}(s,0) = x(s,0) = \varphi(s)$ for $V_* \leq s \leq V_0$.

The first equation of the system (5.11) is a logistic equation, which has a standard solution

$$\hat{x}(s,t) = \begin{cases} \frac{K\varphi(s)e^{\beta Kt}}{K+\varphi(s)(e^{\beta Kt}-1)}, & \text{if } K \neq 0;\\ \frac{\varphi(s)}{\varphi(s)\beta t+1}, & \text{if } K = 0. \end{cases}$$
(5.12)

Combine the second and the third equations in the system (5.11), we get an ordinary differential equation of $\frac{\partial \gamma}{\partial s}(s,t)$,

5.5 Explicit solutions 117

$$\begin{cases} \frac{\partial}{\partial t} \left(\frac{\partial \gamma(s,t)}{\partial s} \right) = (\beta \hat{x}(s,t) - \mu) \frac{\partial \gamma(s,t)}{\partial s}, \\ \frac{\partial \gamma(s,0)}{\partial s} = 1, \end{cases}$$

which has the solution:

$$\frac{\partial \gamma(s,t)}{\partial s} = \exp\left(\int_0^t (\beta \hat{x}(s,\rho) - \mu) d\rho\right).$$

Hence

$$\gamma(s,t) = V_* + \int_{V_*}^s \exp\left(\int_0^t (\beta \hat{x}(\sigma,\rho) - \mu) d\rho\right) d\sigma, \tag{5.13}$$

where $\hat{x}(s,t)$ is given by the formula (5.12). Since $\frac{\partial \gamma(s,t)}{\partial s} = \exp\left(\int_0^t (\beta \hat{x}(s,\rho) - \mu) d\rho\right) > 0$, we can solve for the inverse of $\gamma(s,t)$ for fixed t. Denote the inverse of $\gamma(s,t)$ by n(s,t). It follows that

by
$$\eta(s,t)$$
. It follows that

$$\left\{ \begin{aligned} x(s,t) &= \hat{x}(\eta(s,t),t), \\ \omega(s,t) &= \int_{V_*}^s (\beta x(\sigma,t) - \mu) d\sigma. \end{aligned} \right.$$

The free boundary is given by:

$$V(t) = \gamma(V_0, t) = V_* + \int_{V_*}^{V_0} \exp\left(\int_0^t (\beta \hat{x}(\sigma, \rho) - \mu) d\rho\right) d\sigma,$$

$$R(t) = \sqrt[3]{V(t)} = \sqrt[3]{R_*^3 + \int_{R_*^3}^{R_0^3} \exp\left(\int_0^t (\beta \hat{x}(\sigma, \rho) - \mu) d\rho\right) d\sigma},$$
(5.14)

where $\hat{x}(s,t)$ is calculated by formula (5.12).

In order to solve the system we assumed α to be a constant. If α is a step function, the solution can be given piece-wise by applying the formula (5.12) piece by piece. Let $\alpha(t) = \alpha_i, t \in [t_{i-1}, t_i), i = 1, 2, ..., n, \hat{x}_0(s, 0) :=$ $\varphi(s), K_i := \frac{\alpha_i + \mu}{\beta}$, then the solution in (5.12) changes to $\hat{x}(s, t) = \hat{x}_i(s, t),$ $t \in [t_{i-1}, t_i), i = 1, 2, ..., n$, where 118 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma

$$\hat{x}_{i}(s,t) = \begin{cases} \frac{K_{i}\hat{x}_{i-1}(s,t_{i-1})e^{\beta K_{i}(t-t_{i-1})}}{K_{i}+\hat{x}_{i-1}(s,t_{i-1})(e^{\beta K_{i}(t-t_{i-1})}-1)}, & \text{if } K_{i} \neq 0;\\ \frac{\hat{x}_{i-1}(s,t_{i-1})}{\hat{x}_{i-1}(s,t_{i-1})\beta(t-t_{i-1})+1}, & \text{if } K_{i} = 0, \end{cases}$$
(5.15)

whereas the formula (5.14) will also be given piece by piece.

Here we look at several interesting cases or applications of this explicit solution. First, we look at how tumor regrows after the surgical resection without any treatment. In this case, $K = \frac{\lambda + \mu - \delta}{\lambda + \mu}$, where λ is the tumor cell proliferation rate, δ is death rate of tumor cells (the rate at which the tumor cells becomes necrotic), μ is the removal rate of necrotic cells.

When K = 0, that is, $\delta = \lambda + \mu$, the growth of the tumor radius R(t), or scaled volume $V(t) = R^3(t)$ will follow the following curve,

$$V(t) = (\beta \int_{V_*}^{V_0} \varphi(s) ds) e^{-\mu t} t + V_0 e^{-\mu t} + V_* (1 - e^{-\mu t}).$$

From this expression, we find there is a time T, approximately $T = \frac{1}{\mu}$, such that the tumor radius will increase before T and decrease after it. When time goes to infinity, the tumor shrinks to the size V_* .

When $K \neq 0$, we have

$$V(t) = V_0 + \frac{(e^{\beta Kt} - 1)e^{-\mu t}}{K} \int_{V_*}^{V_0} \varphi(s) ds.$$
 (5.16)

Since $\int_{V_*}^{V_0} \varphi(s) ds$ is a constant, the growth of the radius depends on $\frac{(e^{\beta Kt} - 1)e^{-\mu t}}{K}$ Substituting original parameters, we have $(e^{\beta Kt} - 1)e^{-\mu t} = e^{(\beta K - \mu)t} - e^{-\mu t} = e^{(\lambda - \delta)t} - e^{-\mu t}$. Hence, if $\lambda > \delta$, the tumor will grow infinitely. If $\lambda < \delta$, the tumor will shrink to the size V_0 . If $\lambda = \delta$, the tumor radius will reach a stationary solution

5.6 Summary and Discussion 119

$$V_s = V_0 + \frac{\lambda + \mu}{\mu} \int_{V_*}^{V_0} \varphi(s) ds.$$

Second, we consider radiotherapy and chemotherapy. After the surgical resection, the patient has to rest for a period of time, usually six weeks, and then is treated by radiotherapy and chemotherapy. Set the rest period to be $0 \le t \le t_1$. Then, the tumor grows from V_0 to $V(t_1)$, where $V(t_1)$ is given by (5.16) at $t = t_1$, or explicitly

$$V(t_1) = V_0 + \frac{(\lambda + \mu)(e^{(\lambda - \delta)t_1} - e^{-\mu t_1})}{\lambda + \mu - \delta} \int_{V_*}^{V_0} \varphi(s) ds.$$

Let the treatment period of radio therapy with chemotherapy be $t_1 \le t \le t_2$, then the tumor growth follows

$$V(t) = V_0 + \frac{(\lambda + \mu)(e^{(\lambda - \delta - A - B)t} - e^{-\mu t})}{\lambda + \mu - \delta - A - B} \int_{V_*}^{V_0} \varphi(s) ds, \quad t_1 \le t \le t_2.$$
(5.17)

Let the treatment period of chemotherapy be $t_2 \leq t \leq t_3$, then the tumor growth follows

$$V(t) = V_0 + \frac{(\lambda + \mu)(e^{(\lambda - \delta - B)t} - e^{-\mu t})}{\lambda + \mu - \delta - B} \int_{V_*}^{V_0} \varphi(s) ds, \quad t_2 \le t \le t_3.$$
(5.18)

These solutions clearly give the tumor growth pattern in any finite period of time. They provide some information for treatments. We will discuss this issue in next section.

5.6 Summary and Discussion

We have reviewed a mathematical model of glioblastoma treatment by radiotherapy and chemotherapy, which also incorporates the size of the tumor which is 120 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma removed by surgery. The model can be used to explore the benefits of different protocols of treatment. In particular we have shown that somewhat greater benefits incur if the same total amount of radiation is given over a period of 12 weeks instead of over 6 weeks.

We have also shown that the benefits of chemotherapy are very little for patients already undergoing radiotherapy. Subtotal resection occurs either by design (if the tumor borders a critically essential part of the brain) or because of failure of the surgeon to determine the precise boundary of the tumor. We have estimated the average diameter for the residual tumor at $R_* = 18$ mm, when the tumor size is $R_0 = 20$ mm. Our model can predict the benefits that will occur if the resection will be more complete, or if the radiation dose is increased.

If instead of using the data of the group of patients between the age of 20 and 39 in Albert et al. [3] we use the patients data of different age groups in [3] or the patients data of Stummer et al. [136] for the age group B60, we obtain slightly different parameters A, B, ε , but this does not affect the qualitative conclusions as described in Figs. 5.2 – 5.8.

The present chapter is based on data from [3, 107, 137]. A recent article by Gorlia et al. [72] provides similar data which agree with those of [3, 107, 137] in the case of patients who undergone partial resection with radiotherapy and with or without chemotherapy. However it gives a shorter MST for patients who undergone complete resection with radiotherapy. This inconsistency may be the result of how one interprets complete resection. Our model parameters can be adjusted to the data in [72], but the qualitative results described in Figs. 5.2 – 5.8 will not be affected.

We conclude this chapter with some comments and discussion about explicit solutions we obtained in the previous section.

There is an interesting feature of the explicit solution. Observing the solution formula, we see the dependent domain of $\hat{x}(s,t)$ is one point $\{s\}$ and the dependent domain of $\gamma(s,t)$, as calculated by the formula (5.13), is the interval $[V_*, s]$, hence the dependent domain of the tumor radius R(t), or V(t), is the interval $[V_*, V_0]$. Therefore, if we extend the domain of $\varphi(s)$ from $[V_*, V_0]$ to $[V_*, +\infty)$ by assigning arbitrary values to the extended part of $\varphi(s)$, the solution will not be affected.

Due to this property of the solution, the original free boundary problem can be reformulated as a fixed boundary problem defined on an infinite domain $[V_*, +\infty) \times [0, +\infty)$ with a discontinuous initial condition. That is, the solution of the system (5.8) on its domain coincides with that of the following system:

$$\begin{cases} \frac{\partial x(s,t)}{\partial t} + \omega(s,t) \frac{\partial x(s,t)}{\partial s} = \beta x(s,t) \left(K - x(s,t) \right), & V_* \leq s, t \geq 0, \\ & \frac{\partial \omega(s,t)}{\partial s} = \beta x(s,t) - \mu, & V_* \leq s, t \geq 0, \\ & \omega(V_*,t) = 0, & t \geq 0, \\ & \frac{\partial V(t)}{\partial t} = \omega(V(t),t), & t \geq 0, \end{cases}$$
(5.19)

with the initial conditions: $V(0) = V_0$, $x(s, 0) = \psi(s)$, for $V_* \leq s$, where

$$\psi(s) = \begin{cases} \varphi(s) \,, & s \le V_0 \,; \\ 0 \,, & s > V_0 \,. \end{cases}$$
(5.20)

122 5 Modeling of Resection, Radiation, and Chemotherapy in Glioblastoma For the hyperbolic free boundary problem obtained by the mass conversation law where the free boundary moves only because of the expansion of the inside mass, this property of the solution seems true. Conceptually, we can change this type of free boundary problem to a fixed boundary problem. Moreover, the methods developed in this paper can be used in analysis of more general free boundary problems.

We now discuss some biological significance of our explicit solutions. Based on the solution (5.16) where there is no treatment after the surgical resection, if the tumor cell proliferation rate λ is greater than the tumor cell death rate δ , the tumor will grow until the patient dies. If the the tumor cell proliferation rate λ is smaller than the tumor cell death rate δ , the tumor will shrink, and the patient survives. If the the tumor cell proliferation rate λ is equal to the tumor cell death rate δ , the tumor will grow to a certain size and then stop growing, so that it reaches a stationary state. This threshold phenomenon is biologically reasonable. Unfortunately, the tumor cell proliferation rate λ is always greater than the tumor cell death rate δ in reality; otherwise, there will be no tumor. By the solution (5.17), the tumor is treated by the combined radiotherapy and chemotherapy after the tumor regrows to a size of $V(t_1)$. Theoretically, we can make the combined parameter $\lambda - \delta - A - B$ as small as we want by increasing A and B. This means, we need to increase the strength of the radiotherapy and chemotherapy. Within the tolerable toxicity of these the rapies, $\lambda - \delta - A - B$ may be negative. It is obvious that the longer the tumor is treated by the combined radiotherapy and chemotherapy, the more

the tumor cells are killed. However, the radiotherapy cannot be applied too long because of its side effects and toxicity. Then, the chemotherapy has to be applied individually as the solution (5.18) shows. It may be the case where the density of tumor cells drops to such a low level that is beyond the detection after these treatments. A condition on which the tumor could be eradicated is $\lambda < \delta + B$, and then it is automatically true that $\lambda < \delta + A + B$. Since these are parameters of exponential functions, there is no guarantee that all tumor cells are killed with a period of finite time. However, these solutions can be used to compute the survival times when different protocols of radiotherapy and chemotherapy are applied.

Tumor Modeling with Different Cell Velocities

6.1 Introduction

Cancers appear with multiscale features: genes, cells, and biological tissues, corresponding to the molecular, cellular, and tissue scales. Hence, selecting the proper modeling scale from the multiple scales is an important issue. Bellomo *et al.* [8, 9] discussed the multiscale aspects of cancer modeling. According to the classification of scales, there are basically three types of mathematical models of cancers: microscopic models (at the molecular and the cellular scales), macroscopic models (at the tissue scale), multiscale models (the overall system is viewed as a system of subsystems with specific scales). Microscopic models usually refer to the early stage of cancer onset and developments, and they are derived in the framework of the kinetic theory for active interactions (see [7, 9, 47], for instance). After a suitable maturation time, tumor cells may start to condense and aggregate into a solid form. At this stage, various space phenomena, such as cell motion and tumor size, play a relevant role in the overall dynamics, and macroscopic models may be 126 6 Tumor Modeling with Different Cell Velocities

needed for understanding of tumor growth. Macroscopic models are usually based on mass conservation laws and on reaction-diffusion processes within tumor (see [1, 11, 26, 63, 73, 74, 93, 94, 143, 145, 147, 168], for instance). Multiscale models illustrate that the molecular and cellular events continue to play a crucial role in macroscopic tumor progression (see [23], for instance). Microscopic models often consist of nonlinear integro-differential equations (IDEs), macroscopic models lead to systems of nonlinear partial differential equations (PDEs), and multiscale models are usually hybrid systems. For more detailed descriptions of microscopic models, macroscopic models and multiscale models, the reader is refer to two recent review articles [9, 112].

Multiscale models are good approximations of biological realities, although they can become analytically intractable. On the other hand, microscopic models are often complex and abstract, and they can also become analytically intractable. So, the present chapter and the next two chapters will focus on the mathematical analysis of macroscopic models.

In this chapter, we reviews a mathematical model describing the cell cycle dynamics and chemotactic driven cell movement in a multicellular tumor spheroid. Tumor cells consist of two types of cells: proliferating cells and quiescent cells, which have different chemotactic responses to an extracellular nutrient supply.

((This chapter is organized as follows. Section 6.2 presents the model. Section 6.3 transforms the problem in a moving-domain into a new problem in a fixed domain. Section 6.4 establishes some useful *a priori* bounds. Section 6.5 gives the main ideas of the proof. Section 6.6 studies a parabolic problem with a *nonlinear* boundary condition. Section 6.7 proves the local existence and uniqueness of solutions of the problem. Section 6.8 extends the above local solution to all t > 0. Finally, this chapter is closed with a summary section.))

6.2 Mathematical Model

Multicellular tumor spheroids (MCTSs) are three-dimensional cell cultures which have structural similarity to in vivo tumors, and MCTSs are routinely used as in vitro models of tumor growth. A number of mathematical models of partial differential equations (PDEs) have been developed to describe the growth of MCTSs (see [19, 20, 46, 73, 125, 133, 163, 166, 167, 168], for instance). Rigorous mathematical analysis of these models, such as global existence, uniqueness and stability of a solution are interesting (see [43, 44, 58, 62, 63, 143, 145], for instance).

In this chapter, we review a mathematical model describing the cell cycle and cell movement in a MCTS. This model was proposed by Tindall and Please [163]. One novelty of this model is that the model includes an explicit description of proliferating and quiescent cells within a MCTS. A common feature of most continuum mathematical models of avascular tumor growth is the assumption that all cells within a tumor have a common spatial velocity profile. However, in the model [163], Tindall and Please have considered the possibility of differing cell velocities, which is another novelty of their model.

128 6 Tumor Modeling with Different Cell Velocities

The different cell velocities are created when proliferating and quiescent cells have different chemotactic responses to an extracellular nutrient supply. Considering a spherically symmetric MCTS, Tindall and Please's model reads as follows:

$$\frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial c}{\partial r} \right) = \lambda(c)c, \qquad \text{for } 0 < r < R(t), \ t > 0, \tag{6.1}$$

$$p + q = N,$$
 for $0 \le r \le R(t), t \ge 0,$ (6.2)

$$u_q(r,t) = u_p(r,t) + \chi \frac{\partial c}{\partial r}, \quad \text{for } 0 \le r \le R(t), \ t > 0, \tag{6.3}$$

$$\frac{\partial p}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2(u_p p) \right)$$

$$= D \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial p}{\partial r} \right) + \left(K_b(c) - K_q(c) - K_a(c) \right) p$$

$$+ K_p(c)q, \quad \text{for } 0 < r < R(t), \ t > 0, \quad (6.4)$$

$$\frac{\partial q}{\partial t} + \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2(u_q q) \right)$$

$$= D \frac{1}{r^2} \frac{\partial}{\partial r} \left(r^2 \frac{\partial q}{\partial r} \right) + K_q(c) p$$

$$- \left(K_d(c) + K_p(c) \right) q, \quad \text{for } 0 < r < R(t), \ t > 0, \quad (6.5)$$

$$\frac{\partial c}{\partial r}(r,t) = 0 \text{ at } r = 0, \quad \text{for } t > 0,$$
(6.6)

$$c(r,t) = c_{\infty} \text{ at } r = R(t), \quad \text{for } t > 0,$$
 (6.7)

$$p(r,0) = p_0(r), \text{ for } 0 \le r \le R(0), t > 0,$$
 (6.8)

$$\frac{\partial p}{\partial r}(r,t) = \frac{\partial q}{\partial r}(r,t) = u_p(r,t) = u_q(r,t) = 0 \text{ at } r = 0, \text{ for } t > 0, \quad (6.9)$$

$$p\frac{dR(t)}{dt} - \left(pu_p - D\frac{\partial p}{\partial r}\right) = 0 \quad \text{at } r = R(t),$$
(6.10)

$$q\frac{dR(t)}{dt} - \left(qu_q - D\frac{\partial q}{\partial r}\right) = 0 \quad \text{at } r = R(t),$$
(6.11)

$$R(0) = R_0 > 0 \quad \text{is prescribed.} \tag{6.12}$$

Here R(t), c, p, q, u_p and u_q are unknown functions, which will be explained in the following. R(t) represents the spheroid radius. In (6.1), c(r,t) is the concentration of nutrient, $\lambda(c)$ is any positive smooth function, and $\lambda(c)c$ is a consumption rate of nutrient which is zero when c = 0. In (6.2), N is the total number of live cells per unit volume, and we assume that tumor cells consist of two types of cells: proliferating and quiescent cells. p(r, t) and q(r, t) are the proliferating and quiescent cell densities, respectively. The cells are taken to fill any region within the tumor. In addition, for simplicity, we shall neglect the space taken by any dead cell material [163, 168]. In (6.3), $u_p = u_p(r, t)$ is the velocity of proliferating cells, $u_q = u_q(r, t)$ is the velocity of quiescent cells, and χ is a parameter introduced to describe the relative strength of the chemotactic response of the two cell phases [163]: proliferating cells move up the chemotactic gradient relative to quiescent cells when $\chi < 0$; proliferating cells move at the same velocity as quiescent cells when $\chi = 0$; and proliferating cells move down the chemotactic gradient relative to quiescent cells if $\chi > 0$. In (6.4) and (6.5), cell motion is described by both random motion of the cells (diffusion) and directed motion stimulated by nutrient gradients (chemotaxis). D is a positive constant, which is the random diffusion coefficient of the cells. $K_b(c)$ is the rate of cell birth, $K_p(c)$ is the rate at which cells return to the proliferative compartment from quiescence, $K_q(c)$ is the rate at which proliferating cells become quiescent, $K_a(c)$ is the death rate of proliferating cells, and $K_d(c)$ is the death rate of quiescent cells. $K_a(c)p$ in (6.4) is the death of proliferating cells due to apoptosis, while $K_d(c)q$ in (6.5) is the death 130 6 Tumor Modeling with Different Cell Velocities

of quiescent cells due to necrosis. (6.6) is a result of the radial symmetry assumption of the problem, and (6.7) assumes that the spheroid is supported in a nutrient-rich medium. (6.8) is an initial condition for the cell distribution. (6.9) is also a result of the radial symmetry assumption of the problem. On the outer boundary of the spheroid we impose no-flux conditions as given in (6.10) and (??) for proliferating cells and quiescent cells, respectively; see Remark 6.2 below for further explanation. (6.12) is an initial condition for tumor radius.

By adding Eqs. (6.4) and (6.5) and invoking assumptions (6.2) and (6.3) an equation for the velocity of the proliferating cells is obtained

$$\frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2u_p\right) = \frac{1}{N}\left(K_b(c)p - K_a(c)p - K_d(c)(N-p)\right) \\ + \frac{\chi}{N}\frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2(p-N)\frac{\partial c}{\partial r}\right).$$
(6.13)

By adding Eqs. (6.10) and (6.11) and using assumptions (6.2) and (6.3) an equation for the velocity of the outer boundary of the spheroid is obtained

$$\frac{dR(t)}{dt} = u_p + \frac{\chi}{N} q \frac{\partial c}{\partial r} \quad \text{at } r = R(t).$$
(6.14)

This, together with (6.10), yields a boundary condition for proliferating cells

$$D\frac{\partial p}{\partial r} + \left(\frac{\chi}{N}q\frac{\partial c}{\partial r}\right)p = 0 \quad \text{at } r = R(t).$$
(6.15)

We note that Eq. (6.5) is a consequence of Eqs. (6.2)-(6.4) and (6.13), so that in the sequel we may drop this equation and replace q by N - p in (6.4), (6.14) and (6.15). We also notice that the no-flux boundary conditions (6.10)-(6.11) are equivalent to the boundary conditions (6.14)-(6.15) under the assumptions (6.2) and (6.3), so that in what follows we shall replace (6.10)-(6.11) with (6.14)-(6.15).

We further note that the empirical rules used for the functional dependence of K_a, K_b, K_d, K_p and K_q on c are *not* critical to our analytical result. For the global existence of a solution to the model, we only need the following simple,

$$K_a(c) \ge 0, K_b(c) \ge 0, K_d(c) \ge 0, K_p(c) \ge 0, K_q(c) \ge 0$$

and these functions are C^1 -smooth functions. (6.16)

This assumption is physically realistic. The requirement of the C^1 -smoothness will be explained in Section 6.7.

Remark 6.1. In Tindall and Please's model [163], they neglected the random cell motion term in Eqs. (6.4) and (6.5). Their numerical results clearly indicate the formation of *shocks* in the proliferating cell distributions. Mathematically, by dropping the diffusion terms in Eqs. (6.4) and (6.5), the solution of corresponding first-order hyperbolic equations may evolve shocks due to the dominant cell motion directed by chemotaxis. Here we retain random cell motion in Eqs. (6.4) and (6.5), and prove that afore-mentioned shocks can be smoothed by this random cell motion. Indeed, we will prove the global existence and uniqueness of a $C^{1+\lambda,(1+\lambda)/2}$ -smooth (0 < λ < 1) solution of the model (6.1)-(6.16). Furthermore, in section 6.8 we will give some indication 132 6 Tumor Modeling with Different Cell Velocities

where the analysis might break down if the diffusion of the cell types is sent to zero.

Remark 6.2. If $\chi = 0$, the global existence and uniqueness of a solution to the model (6.1)-(6.16) can be proved using the methods of [145, 169]. So, throughout the remainder of this chapter, we assume that $\chi \neq 0$. On the outer boundary of the spheroid, the flux of proliferating cells has two components, One is the diffusion flux $-D\frac{\partial p}{\partial r}(R(t),t)$, and the other is the flux, $pu_p(R(t),t)$, due to proliferation and death of cells. Since the spheroid changes at the rate $\dot{R}(t)$, the no-flux boundary condition for p should be $p\dot{R} - \left(pu_p - D\frac{\partial p}{\partial r}\right) = 0$ as shown in (6.10). The no-flux boundary condition (6.11) can be similarly explained.

Remark 6.3. Once diffusion is introduced, the analysis in this chapter will be independent of the sign of χ . However, for definiteness and clarity of the statement, we will assume that $\chi > 0$ throughout the remainder of this chapter.

Tindall and Please [163] numerically studied the model (6.1)-(6.16) with D = 0 and empirical linear functions $K_a(c), K_b(c), K_d(c), K_p(c)$ and $K_q(c)$. In particular, they investigated the different distributions of quiescent and proliferating cells that can occur within a MCTS. Tao [150] proved the global existence and uniqueness of a solution to the model (6.1)-(6.16). This chapter will mainly review Tao's results [150]. The main difficulties of the proof are due to the *chemotactic* term in (6.13), to the *nonlinear* boundary condition
(6.15), and to possible singularity at tumor center if we regard Eqs. (6.4) and (6.5) as two 1-dimension parabolic equations with a radial spatial variable r (Note that $\frac{1}{r^2} \frac{\partial}{\partial r} (r^2 \frac{\partial p}{\partial r}) = \frac{\partial^2 p}{\partial r^2} + \frac{2}{r} \frac{\partial p}{\partial r}$). To overcome these difficulties, we establish some necessary estimates, employ the Leray-Schauder fixed point theorem, and use the three-dimensional Cartesian coordinate.

6.3 Transformation and Main Results

After re-scalings (see Appendix in [163]), the system (6.1)-(6.16) takes the following form in $\{0 < \bar{r} < \bar{R}(\bar{t}), \ \bar{t} > 0\}$:

$$\Delta_{\bar{r}}\bar{c} = \bar{\lambda}(\bar{c})\bar{c},\tag{6.17}$$

$$\frac{\partial \bar{c}}{\partial \bar{r}}(0,\bar{t}) = 0, \tag{6.18}$$

$$\bar{c}(\bar{R}(\bar{t}),\bar{t}) = 1, \tag{6.19}$$

$$\frac{\partial \bar{p}}{\partial \bar{t}} + \frac{1}{\bar{r}^2} \frac{\partial}{\partial \bar{r}} \left(\bar{r}^2 (\bar{u}_p \bar{p}) \right)$$

$$= \bar{D} \bigtriangleup_{\bar{r}} \bar{p} + \left(\bar{K}_b(\bar{c}) - \bar{K}_q(\bar{c}) - \bar{K}_a(\bar{c}) \right) \bar{p} + \bar{K}_p(\bar{c})(1 - \bar{p}), \quad (6.20)$$

$$\bar{p}(r,0) = \bar{p}_0(r),$$
(6.21)

$$\frac{\partial \bar{p}}{\partial \bar{r}}(0,\bar{t}) = 0, \quad D\frac{\partial \bar{p}}{\partial \bar{r}} + \left(\bar{\chi}(1-\bar{p})\frac{\partial \bar{c}}{\partial \bar{r}}\right)\bar{p} = 0 \quad \text{at } \bar{r} = \bar{R}(\bar{t}), \tag{6.22}$$

$$\frac{1}{\bar{r}^2} \frac{\partial}{\partial \bar{r}} \left(\bar{r}^2 \bar{u}_p \right)$$

$$= \bar{K}_b(\bar{c})\bar{p} - \bar{K}_a(\bar{c})\bar{p} - \bar{K}_d(\bar{c})(1-\bar{p}) - \frac{\bar{\chi}}{\bar{r}^2} \frac{\partial}{\partial \bar{r}} \left(\bar{r}^2(1-\bar{p})\frac{\partial \bar{c}}{\partial \bar{r}} \right), \quad (6.23)$$

$$\bar{u}_p(0,\bar{t}) = 0,$$
 (6.24)

$$\frac{d\bar{R}(\bar{t})}{d\bar{t}} = \bar{u}_p + \bar{\chi} (1-\bar{p}) \frac{\partial\bar{c}}{\partial\bar{r}} \quad \text{at } \bar{r} = \bar{R}(\bar{t}), \tag{6.25}$$

$$\bar{R}(0) = \bar{R}_0,$$
 (6.26)

where

$$0 \le \bar{K}_a(\bar{c}), \bar{K}_b(\bar{c}), \bar{K}_d(\bar{c}), \bar{K}_p(\bar{c}), \bar{K}_q(\bar{c}) \in C^1$$
(6.27)

and

$$\Delta_{\bar{r}} = \frac{1}{\bar{r}^2} \frac{\partial}{\partial \bar{r}} \Big(\bar{r}^2 \frac{\partial}{\partial \bar{r}} \Big).$$

To transform the moving domain $\{\bar{r} < \bar{R}(\bar{t})\}$ into a fixed domain, as shonw in [145], we introduce a change of variables $(\bar{r}, \bar{t}, \bar{c}, \bar{p}, \bar{u}_p, \bar{R}) \mapsto (\rho, \bar{t}, \tilde{c}, \tilde{p}, \tilde{u}, \bar{R})$ as follows:

$$\rho = \bar{r}/\bar{R}(\bar{t}), \quad \bar{t} = \bar{t},$$

$$\tilde{c}(\rho, \bar{t}) = \bar{c}(\rho\bar{R}(\bar{t}), \bar{t}), \quad \tilde{p}(\rho, \bar{t}) = \bar{p}(\rho\bar{R}(\bar{t}), \bar{t}), \quad (6.28)$$

$$\tilde{u}(\rho, \bar{t}) = \bar{u}_p(\rho\bar{R}(\bar{t}), \bar{t})/\bar{R}(\bar{t}), \quad \bar{R}(\bar{t}) = \bar{R}(\bar{t}).$$

In terms of the new variables and after dropping the tildes of \tilde{c} , \tilde{p} and \tilde{u} and the bars of \bar{t} , \bar{R} , \bar{D} , \bar{K}_a , \bar{K}_b , \bar{K}_d , \bar{K}_p , \bar{K}_q , $\bar{\chi}$ and \bar{R}_0 for notational convenience, the system (6.17)-(6.27) takes the following form in $\{0 < \rho < 1, t > 0\}$:

$$\triangle_{\rho}c = R^2(t)\lambda(c)c, \tag{6.29}$$

$$\frac{\partial c}{\partial \rho}(0,t) = 0, \quad c(1,t) = 1,$$

$$\frac{\partial p}{\partial t} + \left[u(\rho,t) - \rho u(1,t)\right] \frac{\partial p}{\partial \rho} \\
+ \left[\frac{\chi}{R^2(t)} \frac{\partial c}{\partial \rho} p - \chi \rho \frac{1}{R^2(t)} \frac{\partial c(1,t)}{\partial \rho} (1 - p(1,t))\right] \frac{\partial p}{\partial \rho} - \frac{D}{R^2(t)} \Delta_{\rho} p \\
= \left[-K_q(c) + \left(K_b(c) + K_d(c) - K_a(c) + \chi \lambda(c)c\right)(1 - p)\right] p \\
+ K_p(c)(1 - p),$$
(6.31)

6.3 Transformation and Main Results 135

$$p(\rho, 0) = p_0(\rho),$$
 (6.32)

$$\frac{\partial p}{\partial \rho}(0,t) = 0, \quad \left[D \frac{\partial p}{\partial \rho} + \left(\chi (1-p) \frac{\partial c}{\partial \rho} \right) p \right] \Big|_{\rho=1} = 0, \tag{6.33}$$

$$u(\rho,t) = \frac{1}{\rho^2} \int_0^{\rho} \left[K_b(c)p - K_a(c)p - K_d(c)(1-p) \right] s^2 ds -\chi(1-p) \frac{1}{R^2(t)} \frac{\partial c}{\partial \rho},$$
(6.34)

$$\frac{dR(t)}{dt} = R(t) \int_0^1 \left[K_b(c)p - K_a(c)p - K_d(c)(1-p) \right] s^2 ds, \quad (6.35)$$

$$R(0) = R_0, \quad R_0 \text{ is given}, \tag{6.36}$$

$$0 \le K_a(c), K_b(c), K_d(c), K_p(c), K_q(c) \in C^1,$$
(6.37)

where we have used the fact that $\triangle_{\rho}c = R^2(t)\lambda(c)c$ in deriving Eq. (6.31). We shall also assume that

$$0 \le p_0(\rho) \le 1,$$

$$p_0(\rho) \in W_k^2(B_1(0)),$$

$$\frac{\partial p_0}{\partial \rho}(0) = \left(D \frac{\partial p_0}{\partial \rho} + \chi p_0(1-p_0) \frac{\partial c}{\partial \rho} \Big|_{t=0} \right) \Big|_{\rho=1} = 0,$$
(6.38)

here $B_1(0) = \{y \in \mathbb{R}^3 : |y| \le 1\}$ and $W_k^2(B_1(0)) := \{\varphi(\rho) | \varphi, \varphi_{y_i}, \varphi_{y_i y_j} \in L^k(B_1(0)\}$, in which k > 5, i, j = 1, 2, 3, and the derivatives are in the weak sense. We note that (6.38) is physically realistic as it ensures that both of the cell populations, p and q, are initially non-negative. Throughout this chapter, we also assume that

$$\lambda(c)$$
 is any positive C^1 -smooth function, (6.39)

which is physically realistic as it ensures that the consumption rate of nutrient, $\lambda(c)c$, is non-negative smooth function of c which is zero when c = 0. The requirement of the C^1 -smoothness of $\lambda(c)$ will be explained in next section.

Remark 6.4. Tindall and Please's model [163] empirically assumed that $\lambda(c) =$ 1. However, physically any positive smooth function should be adequate for global existence of a solution to the model.

We shall use the following notation:

$$Q_T = B_1(0) \times [0,T], \quad W_k^{2,1}(Q_T) = \left\{ p(\rho,t) | p, \ p_{y_i}, \ p_{y_iy_j}, \ p_t \in L^k(Q_T) \right\}$$

where $1 \le k \le \infty$, i, j = 1, 2, 3, and the derivatives are in the weak sense.

The main result of this chapter is as follows:

Theorem 6.1. Under the assumptions (6.37)-(6.39), there exists a unique solution $(R(t), c(\rho, t), u(\rho, t), p(\rho, t))$ of the problem (6.29)-(6.36) for all t >0; furthermore, $R(t) \in C^1[0, \infty)$, $u(\rho, t) \in C^1([0, 1] \times [0, \infty))$, $c(\rho, t) \in$ $C^{2,1}([0, 1] \times [0, \infty))$, $p(\rho, t) \in W_k^{2,1}(Q_T)$ for some k > 5 and any T > 0, and

$$0 < c(\rho, t) \le 1, \tag{6.40}$$

$$0 \le p(\rho, t) \le 1,\tag{6.41}$$

$$|u(\rho,t)| \le \beta,\tag{6.42}$$

$$R_0 e^{-\beta t} \le R(t) \le R_0 e^{\beta t} \tag{6.43}$$

for some $\beta > 0$.

6.4 A Priori Bounds

In this section we establish several *a priori* bounds which will be used later.

Lemma 6.1. Under the assumptions (6.37)-(6.39), for any solution of (6.29)-(6.36) there hold:

$$0 \le c(\rho, t) \le 1, \quad for \ 0 \le \rho \le 1, \ t > 0,$$
 (6.44)

$$0 \le \frac{1}{\rho R^2(t)} \frac{\partial c}{\partial \rho} \le M_0, \quad for \ 0 \le \rho \le 1, \ t > 0, \tag{6.45}$$

where $M_0 := \frac{1}{3} \max_{0 \le c \le 1} \lambda(c) > 0$.

Proof. Set $r := \rho R(t)$. Then Eqs. (6.29)-(6.30) can be rewritten as follows:

$$\Delta_r c = \lambda(c)c, \quad \text{for } 0 < r < R(t), \ t > 0, \tag{6.46}$$

$$\frac{\partial c}{\partial r}(0,t) = 0, \quad c(R(t),t) = 1.$$
(6.47)

By $\lambda(c) > 0$ and the maximum principle for elliptic equations, we have

$$c(r,t) \ge 0 \tag{6.48}$$

and

$$c(r,t) \le 1. \tag{6.49}$$

Hence, the proof of (6.44) is completed.

We now turn to prove (6.45). We derive from (6.46), the first equation in (6.47), (6.39), (6.48) and (6.49) that

$$0 \le r^2 \frac{\partial c}{\partial r}(r,t) = \int_0^r \lambda(c(s,t))c(s,t)s^2 ds \le \frac{1}{3}\lambda_1 r^3$$

and therefore

$$0 \le \frac{1}{r} \frac{\partial c}{\partial r}(r, t) \le \frac{1}{3} \lambda_1, \tag{6.50}$$

where $\lambda_1 = \max_{0 \le c \le 1} \lambda(c) > 0$. This completes of the proof of (6.45). \Box

Remark 6.5. We note that the C^1 -smoothness of function $\lambda(c)$ as assumed in (6.39) is used for deriving the C^2 -regularity of the solution to problem (6.46)-(6.47) by Schauder theory.

Lemma 6.2. For any solution of (6.29)-(6.39) with $p \in C(Q_T)$ and $R(t) \in C[0,T]$, there holds:

$$0 \le p(\rho, t) \le 1.$$
 (6.51)

Proof. We first assert that

if the minimum of p in Q_T is negative, then it *cannot*

be attained at the boundary
$$\rho = 1.$$
 (6.52)

Suppose, to the contrary, that there exists a point (y_0, t_0) $(|y_0| = 1, 0 < t_0 \le T)$ such that

$$p(y_0, t_0) = \min_{0 \le t \le T} p(y, t) < 0.$$
(6.53)

Then we have

$$\frac{\partial p}{\partial \rho}\Big|_{(y_0,t_0)} \le 0. \tag{6.54}$$

We note that

$$\frac{\partial c}{\partial \rho}(1,t) > 0 \tag{6.55}$$

by (6.29)-(6.30) and the strong maximum principle of elliptic equations. The inequality (6.55), together with (6.53)-(6.54) and $\chi > 0$, further yields

6.4 A Priori Bounds 139

$$\Big(D\frac{\partial p}{\partial \rho} + \chi p(1-p)\frac{\partial c}{\partial \rho}\Big)\Big|_{(y_0,t_0)} < 0$$

which contradicts the boundary condition of p at the boundary $\rho = 1$ in (6.33). So, (6.52) holds.

On the other hand, by $K_p(c) \ge 0$, Eq. (6.31) can be written as follows:

$$\frac{\partial p}{\partial t} - \frac{D}{R^2(t)} \,\Delta_\rho \, p + a(p, u, c) \frac{\partial p}{\partial \rho} + b_1(p, u, c) p \ge 0, \tag{6.56}$$

where

$$a(p, u, c) = u(\rho, t) - \rho u(1, t) + \frac{\chi}{R^2(t)} \frac{\partial c}{\partial \rho} p - \chi \rho \frac{1}{R^2(t)} \frac{\partial c(1, t)}{\partial \rho} (1 - p(1, t)),$$

$$b_1(p, u, c) = K_p(c) + K_q(c)$$

$$- (K_b(c) + K_d(c) - K_a(c) + \chi \lambda(c)c)(1 - p).$$

Without loss of generality, we may assume that $b_1 \ge 0$ due to the standard exponential transform (i.e. $p = e^{k_0 t} \bar{p}$ for large $k_0 > 0$). Therefore, we derive from $b_1 \ge 0$ and (6.56) that

> if the minimum of p in Q_T is negative, then it *cannot* be attained in the interior of Q_T . (6.57)

This, together with (6.52), (6.32) and (6.38), that

$$p(\rho, t) \ge 0. \tag{6.58}$$

Next we shall prove that $p(\rho, t) \leq 1$. To this end, we set q =: 1 - p. We easily derive from (6.31)-(6.33), (6.38), (6.58) and $K_q(c) \geq 0$ that

$$\frac{\partial q}{\partial t} - \frac{D}{R^2(t)} \bigtriangleup_{\rho} q + a(p, u, c) \frac{\partial q}{\partial \rho} + b_2(p, u, c) q \ge 0 \quad \text{in } Q_T, \quad (6.59)$$

$$q(\rho, 0) = 1 - p_0(\rho) \ge 0, \quad \frac{\partial q}{\partial \rho}(0, t) = 0,$$
 (6.60)

$$\left[D\frac{\partial q}{\partial \rho} + \left(-\chi p\frac{\partial c}{\partial \rho}\right)q\right]_{\rho=1} = 0, \tag{6.61}$$

here

$$b_2(p, u, c) = K_p(c) + (K_b(c) + K_d(c) - K_a(c) + \chi \lambda(c)c)p$$

and $\frac{\partial}{\partial \rho} \equiv \frac{\partial}{\partial \nu}$, in which ν is the *outward* vector of the domain $B_1(0)$. By (6.45), $p \in C(Q_T)$ and $R(t) \in C[0,T]$, we also have

$$-\chi p \frac{\partial c}{\partial \rho} \ge -A(T) \Leftrightarrow -\chi p \frac{\partial c}{\partial \rho}$$
 is bounded from below,

where A(T) > 0 is some constant possibly depending on T. Therefore, the comparison principle holds for problem (6.59)-(6.61) with the third boundary condition (cf. [109, Theorem 2.10] and the remarks following that). Hence

$$q(\rho, t) \ge 0.$$

This completes the proof of Lemma 6.2. \Box

From (6.34)-(6.35) and Lemmas 6.1 and 6.2, we easily deduce that

Lemma 6.3. For any solution of (6.29)-(6.39) with $p \in C(Q_T)$ and $R(t) \in C[0,T]$, there hold:

$$-\beta \le \frac{u(\rho, t)}{\rho} \le \beta, \tag{6.62}$$

$$R_0 e^{-\beta t} \le R(t) \le R_0 e^{\beta t},\tag{6.63}$$

where $\beta > 0$ is some positive constant.

If we regard Eq. (6.31) as a 1-dimensional parabolic equation with the spatial variable ρ , then the coefficient of $\partial p/\partial \rho$ has singularity at tumor center $\rho = 0$ due to

$$\Delta_{\rho} p \equiv \frac{\partial^2 p}{\partial \rho^2} + \frac{2}{\rho} \frac{\partial p}{\partial \rho}.$$

However, this singularity can be eliminated by using the estimates (6.45) and (6.62) and employing the three-dimensional Cartesian coordinate.

It is easily checked that

$$\rho p_{\rho} \equiv y \cdot \nabla p, \quad \triangle_{\rho} p \equiv \triangle p, \tag{6.64}$$

where $y = (y_1, y_2, y_3)$, $\rho = \sqrt{y_1^2 + y_2^2 + y_3^2}$, $\nabla = (\frac{\partial}{\partial y_1}, \frac{\partial}{\partial y_2}, \frac{\partial}{\partial y_3})$, $\Delta = \frac{\partial^2}{\partial y_1^2} + \frac{\partial^2}{\partial y_2^2} + \frac{\partial^2}{\partial y_3^2}$. Then the system (6.29)-(6.36) can be rewritten in the following form in Q_T :

$$\Delta_{\rho}c = R^2(t)\lambda(c)c, \tag{6.65}$$

$$\frac{\partial c}{\partial \rho}(0,t) = 0, \quad c(1,t) = 1, \tag{6.66}$$

$$\frac{\partial p}{\partial t} - \frac{D}{R^2(t)} \bigtriangleup p$$

$$+ \left[\frac{u(\rho, t)}{\rho} - u(1, t) + \frac{\chi}{\rho R^2(t)} \frac{\partial c}{\partial \rho} p - \frac{\chi}{R^2(t)} \frac{\partial c(1, t)}{\partial \rho} (1 - p(1, t))\right] y \cdot \bigtriangledown p$$

$$= \left[-K_q(c) + K(c)(1 - p)\right] p + K_p(c)(1 - p),$$
(6.67)

$$p(\rho, 0) = p_0(\rho), \tag{6.68}$$

$$\frac{\partial p}{\partial \rho}(0,t) = 0, \quad \left[D \frac{\partial p}{\partial \rho} + \left(\chi(1-p) \frac{\partial c}{\partial \rho} \right) p \right] \Big|_{\rho=1} = 0, \tag{6.69}$$

$$u(\rho,t) = \frac{1}{\rho^2} \int_0^{\rho} \left[K_b(c)p - K_a(c)p - K_d(c)(1-p) \right] s^2 ds -\chi(1-p) \frac{1}{R^2(t)} \frac{\partial c}{\partial \rho},$$
(6.70)

$$\frac{dR(t)}{dt} = R(t) \int_0^1 \left[K_b(c)p - K_a(c)p - K_d(c)(1-p) \right] s^2 \, ds, \quad (6.71)$$

$$R(0) = R_0, \quad R_0 \text{ is given}, \tag{6.72}$$

$$0 \le K_a(c), K_b(c), K_d(c), K_p(c), K_q(c) \in C^1,$$
(6.73)

where

$$K(c) = K_b(c) + K_d(c) - K_a(c) + \chi \lambda(c)c.$$
 (6.74)

Lemma 6.4. Let T be any finite positive number. Then, for any solution of (6.29)-(6.39) with $p \in C^{\gamma,\gamma/2}(Q_T)$ and $R(t) \in C^1[0,T]$, there holds $p \in W_k^{2,1}(Q_T)$ with $k(1-\gamma) < 1$. Furthermore,

$$\|p\|_{W_k^{2,1}(Q_T)} \le M(T), \tag{6.75}$$

where M(T) > 0 is some constant which may depend on T.

Proof. By (6.37), Lemmas 6.6, 6.8 and 6.9, and the assumption $R(t) \in C[0, T]$, Eq. (6.67) can be rewritten in the following form:

$$\frac{\partial p}{\partial t} - \frac{D}{R^2(t)} \bigtriangleup p + a_0 \ y \cdot \bigtriangledown p + b_0 \ p = h \tag{6.76}$$

with

$$\frac{D}{R^2(t)} \in C(Q_T), \quad \frac{D}{R_0^2} e^{-2\beta T} \le \frac{D}{R^2(t)} \le \frac{D}{R_0^2} e^{2\beta T}, \tag{6.77}$$

$$||a_0||_{L^{\infty}}, ||b_0||_{L^{\infty}}, ||h||_{L^{\infty}} \le M_1,$$
(6.78)

where M_1 is some positive constant. We easily derive from (6.65)-(6.66) and $R(t) \in C^1[0,T]$ that

$$\frac{\partial c}{\partial \rho} \in C^{1,1}(Q_T).$$

This, together with $p \in C^{\gamma,\gamma/2}(Q_T)$, yields

$$\chi(1-p)\frac{\partial c}{\partial \rho} \in C^{\gamma,\gamma/2}(Q_T).$$
(6.79)

Hence, by (6.76)-(6.79), (6.68)-(6.69), (6.38) and the parabolic L^p -theory (cf. [109, Theorem 7.20]) we see that $\hat{p} \in W_k^{2,1}(Q_T)$ with $k(1 - \gamma) < 1$ and

$$\|p\|_{W_k^{2,1}(Q_T)} \le A(T) \Big(\|p_0(\rho)\|_{W^{2,k}(B_1(0))} + \|h\|_{L^k(Q_T)} \Big), \tag{6.80}$$

where A(T) is some constant which may depend on T. This, combined with $\|h\|_{L^{\infty}} \leq M_1$, further yields

$$\|p\|_{W_k^{2,1}(Q_T)} \le A(T) \Big(\|p_0(\rho)\|_{W^{2,k}(B_1(0))} + M_1 \big| B_1(0) \big| T^{\frac{1}{k}} \Big) := M(T).$$

This completes the proof of Lemma 6.4. \Box

6.5 Main Ideas

For clarity, in this section we will give the sketch of the proof of Theorem ??. The detailed proof will be left in the next three sections. We shall use the contraction mapping principle to prove that (6.65)-(6.74) has a unique local solution. For given T > 0, we introduce a metric space (X_T, d) as follows:

$$X_T = \left\{ \begin{array}{ll} (R,p) = (R(t), p(\rho, t)) & (0 \le \rho \le 1, \ 0 \le t \le T) : \\ R(t) \in C^1[0,T], \ R(0) = R_0, \ \frac{1}{2}R_0 \le R(t) \le 2R_0; \\ p(\rho,t) \in C^{\gamma,\gamma/2}(Q_T) \text{ with } \gamma \in (\frac{4}{5},1), \ 0 \le p(\rho,t) \le 1, \\ \frac{\partial p}{\partial \rho}(0,t) = 0, \ p(\rho,0) = p_0(\rho), \\ \left[D\frac{\partial p}{\partial \rho} + \left(\chi(1-p)\frac{\partial c}{\partial \rho}\right)p \right] \Big|_{\rho=1} = 0 \quad \text{where } c \text{ is} \end{array}$$

the solution to problem (6.65)-(6.66) for given R(t) }.

The metric d in X_T is defined by

$$d((R_1, p_1), (R_2, p_2)) = ||R_1 - R_2||_{C^1[0,T]} + ||p_1 - p_2||_{C^{\gamma,\gamma/2}(Q_T)}.$$

For any given $(R(t), p(\rho, t)) \in X_T$ we define $c(\rho, t)$ being the solution of (6.65)-(6.66) and define $u(\rho, t)$ by (6.70). Let $\hat{R}(t)$ and $\hat{p}(\rho, t)$ solve the following two decoupled problems in $\{0 \le \rho \le 1, t \ge 0\}$:

$$\frac{d\hat{R}(t)}{dt} = \hat{R}(t) \int_0^1 \left[K_b(c)p - K_a(c)p - K_d(c)(1-p) \right] s^2 \, ds, \tag{6.81}$$

$$\hat{R}(0) = R_0,$$
 (6.82)

$$\frac{\partial \hat{p}}{\partial t} - \frac{D}{R^2(t)} \bigtriangleup \hat{p} \\
+ \left[\frac{u(\rho, t)}{\rho} - u(1, t) + \frac{\chi}{\rho R^2(t)} \frac{\partial c}{\partial \rho} p - \frac{\chi}{R^2(t)} \frac{\partial c(1, t)}{\partial \rho} (1 - p(1, t)) \right] y \cdot \bigtriangledown \hat{p} \\
= \left[-K_q(c) + K(c)(1 - \hat{p}) \right] \hat{p} + K_p(c)(1 - \hat{p}),$$
(6.83)

$$\hat{p}(\rho, 0) = p_0(\rho),$$
(6.84)

$$\frac{\partial \hat{p}}{\partial \rho}(0,t) = 0, \quad \left[D \frac{\partial \hat{p}}{\partial \rho} + \left(\chi (1-\hat{p}) \frac{\partial c}{\partial \rho} \right) \hat{p} \right] \Big|_{\rho=1} = 0.$$
(6.85)

Then, we define a mapping

$$F: (R(t), p(\rho, t)) \longrightarrow (\hat{R}(t), \hat{p}(\rho, t)).$$

In next two sections we will prove that F is contractive if T is sufficiently small, and this will complete the proof of the local existence and uniqueness of a solution of the system (6.65)-(6.74). The global existence will be proved in Section 6.8.

We first consider the problem (6.81)-(6.82). Clearly,

$$\hat{R}(t) = R_0 e^{\int_0^t \left(\int_0^1 \left[K_b(c)p - K_a(c)p - K_d(c)(1-p)\right]s^2 \, ds\right)d\tau} \in C^1[0,T], \qquad (6.86)$$

$$\frac{R_0}{2} \le \hat{R}(t) \le 2R_0, \quad \text{for } 0 \le t \le T,$$
(6.87)

where T > 0 is sufficiently small.

Since the problem (6.83)-(6.85) is a *new* nonlinear parabolic problem, the solvability of it will be left in next section.

6.6 Problem with Nonlinear Boundary Condition

In this section we shall solve the problem (6.83)-(6.85). For notational convenience, in what follows we shall denote various constants which are independent of Tby A_0 . The main result of this section is as follows.

Theorem 6.2. Under the assumptions (6.37)-(6.39), for any $(R,p) \in X_T$, the problem (6.83)-(6.85) has a unique solution $\hat{p} \in C^{\gamma,\gamma/2}(Q_T)$ for 0 < T < 1. Furthermore,

$$0 \le \hat{p}(\rho, t) \le 1,\tag{6.88}$$

$$\|\hat{p}(\rho, t)\|_{W_k^{2,1}(Q_T)} \le A_0, \tag{6.89}$$

where $A_0 > 0$ is some constant being independent of T.

Proof. Existence: We will use the Leray-Schauder fixed point theorem (cf. [71, Theorem 11.6]) to prove the existence of a solution $\hat{p} \in C^{\gamma,\gamma/2}(Q_T)$ to the problem (6.83)-(6.85). To this end, we set $P =: \{p \mid p(\rho, t) \in C^{\gamma,\gamma/2}(Q_T)\}$, which is a Banach space. For any $w \in P$ and $\sigma \in [0,1]$, we let \hat{p} solve the following *linear* parabolic problem:

$$\frac{\partial \hat{p}}{\partial t} - \frac{D}{R^2(t)} \bigtriangleup \hat{p} \\
+ \left[\frac{u(\rho, t)}{\rho} - u(1, t) + \frac{\chi}{\rho R^2(t)} \frac{\partial c}{\partial \rho} p - \frac{\chi}{R^2(t)} \frac{\partial c(1, t)}{\partial \rho} \left(1 - p(1, t) \right) \right] y \cdot \bigtriangledown \hat{p} \\
= \sigma \left[-K_q(c) + K(c)(1 - w) \right] \hat{p} + \sigma K_p(c)(1 - \hat{p}),$$
(6.90)

$$\hat{p}(\rho, 0) = \sigma p_0(\rho),$$
(6.91)

$$\frac{\partial \hat{p}}{\partial \rho}(0,t) = 0, \tag{6.92}$$

$$\left[D\frac{\partial\hat{p}}{\partial\rho} + \sigma\left(\chi(1-w)\frac{\partial c}{\partial\rho}\right)\hat{p}\right]\Big|_{\rho=1} = 0.$$
(6.93)

Using $(R, p) \in X_T$, Lemmas 6.1 and 6.3, the assumptions (6.37)-(6.39), and the parabolic L^p -theory (cf. [109, Theorem 7.20]), we find that problem (6.90)-(6.93) has a unique solution $\hat{p}(\rho, t) \in W_k^{2,1}(Q_T)$ with $k(1 - \gamma) < 1$. Furthermore, as before,

$$\|\hat{p}(\rho,t)\|_{W_k^{2,1}(Q_T)} \le A_0 \tag{6.94}$$

provided 0 < T < 1. We then can take some k > 5 satisfying $k(1-\gamma) < 1$ due to $\gamma \in (\frac{4}{5}, 1)$. Using the Sobolev embedding $W_k^{2,1}(Q_T) \hookrightarrow C^{1+\lambda,(1+\lambda)/2}(Q_T)$ $(k > 5, \lambda = 1 - \frac{5}{k}$; cf. [108]) and therefore by (6.94), we have

$$\|\hat{p}(\rho,t)\|_{C^{1+\lambda,(1+\lambda)/2}(Q_T)} \le A_0.$$
(6.95)

We now can define a mapping

6.6 Problem with Nonlinear Boundary Condition 147

$$\begin{array}{ll} S: & P\times [0,1]\mapsto P\\ & \hat{p}=S(w,\sigma). \end{array}$$

By $C^{1+\lambda,(1+\lambda)/2}(Q_T) \hookrightarrow C^{\gamma,\gamma/2}(Q_T)$ compactly,

the mapping S is well defined and it is a compact mapping. (6.96)

Clearly, by the maximum principle,

$$S(w,0) = 0 \quad \text{for all } w \in P. \tag{6.97}$$

If $\hat{p} = S(\hat{p}, \sigma)$ for some $\sigma \in [0, 1]$, then by the define of the mapping S, \hat{p} satisfies

$$\frac{\partial \hat{p}}{\partial t} - \frac{D}{R^2(t)} \bigtriangleup \hat{p} \\
+ \left[\frac{u(\rho, t)}{\rho} - u(1, t) + \frac{\chi}{\rho R^2(t)} \frac{\partial c}{\partial \rho} p - \frac{\chi}{R^2(t)} \frac{\partial c(1, t)}{\partial \rho} (1 - p(1, t))\right] y \cdot \bigtriangledown \hat{p} \\
= \sigma \left[-K_q(c) + K(c)(1 - \hat{p})\right] \hat{p} + \sigma K_p(c)(1 - \hat{p}),$$
(6.98)

$$\hat{r}(a, 0) = \sigma r_{a}(a)$$
 (6.00)

$$p(\rho, 0) = \sigma p_0(\rho), \tag{6.99}$$

$$\frac{\partial \hat{p}}{\partial \rho}(0,t) = 0, \tag{6.100}$$

$$\left[D\frac{\partial\hat{p}}{\partial\rho} + \sigma\left(\chi(1-\hat{p})\frac{\partial c}{\partial\rho}\right)\hat{p}\right]\Big|_{\rho=1} = 0.$$
(6.101)

Proceeding as in the proof of Lemma 6.2, we can prove that

$$0 \le \hat{p} \le 1. \tag{6.102}$$

Then, by Lemma 6.4, we have

$$\|\hat{p}(\rho, t)\|_{W_k^{2,1}(Q_T)} \le A_0 \tag{6.103}$$

for 0 < T < 1, and therefore by the Sobolev embedding $W_k^{2,1}(Q_T) \hookrightarrow C^{1+\lambda,(1+\lambda)/2}(Q_T)$ $(k > 5, \lambda = 1 - \frac{5}{k})$ again

$$\|\hat{p}(\rho,t)\|_{C^{1+\lambda,(1+\lambda)/2}(Q_T)} \le A_0.$$

This further yields

$$\begin{split} \|\hat{p}(\rho,t)\|_{C^{\gamma,\gamma/2}(Q_{T})} &= \|\hat{p}(\rho,t)\|_{C^{0}(Q_{T})} + \|\hat{p}(\rho,t)\|_{C^{\gamma,0}(Q_{T})} + \|\hat{p}(\rho,t)\|_{C^{0,\gamma/2}(Q_{T})} \\ &\leq \|\hat{p}(\rho,t) - \hat{p}(\rho,0)\|_{C^{0}(Q_{T})} + \|\hat{p}(\rho,0)\|_{C^{0}(Q_{T})} \\ &+ \|\hat{p}(\rho,t)\|_{C^{1,0}(Q_{T})} + \|\hat{p}(\rho,t)\|_{C^{0,\gamma/2}(Q_{T})} \\ &\leq T^{\frac{1+\lambda}{2}} \|\hat{p}(\rho,t)\|_{C^{0,(1+\lambda)/2}(Q_{T})} + \|\hat{p}(\rho,0)\|_{C^{1}(Q_{T})} \\ &+ \|\hat{p}(\rho,t) - \hat{p}(\rho,0)\|_{C^{1,0}(Q_{T})} + \|\hat{p}(\rho,0)\|_{C^{1}(Q_{T})} \\ &+ T^{\frac{1+\lambda-\gamma}{2}} \|\hat{p}(\rho,t)\|_{C^{0,(1+\lambda)/2}(Q_{T})} \\ &\leq \left(T^{\frac{1+\lambda}{2}} + T^{\frac{1+\lambda-\gamma}{2}}\right) \|\hat{p}(\rho,t)\|_{C^{0,(1+\lambda)/2}(Q_{T})} \\ &+ T^{\frac{1+\lambda}{2}} \|\hat{p}(\rho,t)\|_{C^{1,(1+\lambda)/2}(Q_{T})} + 2\|\hat{p}_{0}(\rho)\|_{C^{1}(B_{1}(0))} \\ &\leq \left(2T^{\frac{1+\lambda}{2}} + T^{\frac{1+\lambda-\gamma}{2}}\right) A_{0} + 2\|\hat{p}_{0}(\rho)\|_{C^{1}(B_{1}(0))} \\ &\leq 3A_{0} + \|\hat{p}_{0}(\rho)\|_{C^{1}(B_{1}(0)))} \quad \text{provided } 0 < T < 1 \\ &\coloneqq M, \end{aligned}$$
 (6.104)

where M > 0 is a constant. Summarizing (6.98)-(6.102) and (6.104), we have that there exists a constant M > 0 such that

$$\|\hat{p}(\rho, t)\|_{C^{\gamma, \gamma/2}(Q_T)} \le M \tag{6.105}$$

for all $(\hat{p}, \sigma) \in P \times [0, T]$ satisfying $\hat{p} = S(\hat{p}, \sigma)$. We then conclude from (6.96), (6.97), (6.105), and the Leray-Schauder fixed point theorem that $S(\hat{p}, 1)$ has a fixed point in P for 0 < T < 1. That is, (6.83)-(6.85) has a solution $\hat{p}(\rho, t) \in C^{\gamma, \gamma/2}(Q_T)$ for 0 < T < 1.

Uniqueness: By the maximum principle for parabolic equations with the third boundary condition as afore-mentioned, we easily prove the uniqueness of a solution to problem (6.83)-(6.85) with given $(R, p) \in X_T$.

Estimates: The estimates (6.88) and (6.89) follow from Lemmas 6.2 and 6.4. \Box

Now, we conclude from (6.82), (6.84)-(6.87), (6.88) and (6.105), that $(\hat{R}(t), \hat{p}(\rho, t)) \in X_T$ for small T > 0. Thus, the mapping F is well defined and it maps X_T into itself for small T > 0.

In next section we shall prove that F is contractive provided T is sufficiently small.

6.7 Local Existence and Uniqueness

To complete the proof of the local existence and uniqueness of a solution to problem (6.65)-(6.74), we still need to prove that F is contractive provided Tis sufficiently small.

Take (R_1, p_1) and (R_2, p_2) in X_T , denote $(\hat{R}_i, \hat{p}_i) = F(R_i, p_i)$, i = 1, 2, and set $R^* = \hat{R}_1 - \hat{R}_2$, $p^* = \hat{p}_1 - \hat{p}_2$. Then, by direct calculations we see that $R^*(t)$ and $p^*(\rho, t)$ satisfy the following two decoupled problems:

$$\frac{dR^*(t)}{dt} = R^*(t)g_1(t) + g_2(t), \quad \text{for } t > 0, \tag{6.106}$$

$$R^*(0) = 0, (6.107)$$

where

$$g_{1}(t) = \int_{0}^{1} \left[K_{b}(c_{1})p_{1} - K_{a}(c_{1})p_{1} - K_{d}(c_{1})(1-p_{1}) \right] s^{2} ds, \quad (6.108)$$

$$g_{2}(t) = \hat{R}_{2}(t) \int_{0}^{1} \left[K_{b}(c_{1})p_{1} - K_{a}(c_{1})p_{1} - K_{d}(c_{1})(1-p_{1}) \right] s^{2} ds$$

$$-\hat{R}_{2}(t) \int_{0}^{1} \left[K_{b}(c_{2})p_{2} - K_{a}(c_{2})p_{2} - K_{d}(c_{2})(1-p_{2}) \right] s^{2} ds, \quad (6.109)$$

$$\frac{\partial p^*}{\partial t} - \frac{D}{R_1^2(t)} \bigtriangleup p^* \\
+ \Big[\frac{u_1(\rho, t)}{\rho} - u_1(1, t) + \frac{\chi}{\rho R_1^2(t)} \frac{\partial c_1}{\partial \rho} p_1 - \frac{\chi}{R_1^2(t)} \frac{\partial c_1(1, t)}{\partial \rho} \big(1 - p_1(1, t) \big) \Big] y \cdot \bigtriangledown p^* \\
- \Big[- K_q(c_1) - K_p(c_1) + K(c_1)(1 - \hat{p}_1) \Big] p^* = f(\rho, t) \quad \text{in } Q_T,$$
(6.110)

$$p^*(\rho, 0) = 0, \tag{6.111}$$

$$\frac{\partial p^*}{\partial \rho}(0,t) = 0, \tag{6.112}$$

$$D\frac{\partial p^*}{\partial \rho} + \left(\chi(1-\hat{p}_1-\hat{p}_2)\frac{\partial c_1}{\partial \rho}\right)p^* = g(\rho,t) \quad \text{at } \rho = 1,$$
(6.113)

where

$$\begin{split} f(\rho,t) &= D\Big(\frac{1}{R_1^2(t)} - \frac{1}{R_2^2(t)}\Big) \bigtriangleup \hat{p}_2 \\ &- \Big[\Big(\frac{u_1(\rho,t)}{\rho} - u_1(1,t) + \frac{\chi}{\rho R_1^2(t)} \frac{\partial c_1}{\partial \rho} p_1 - \frac{\chi}{R_1^2(t)} \frac{\partial c_1(1,t)}{\partial \rho} \big(1 - p_1(1,t)\big)\Big) \\ &- \Big(\frac{u_2(\rho,t)}{\rho} - u_2(1,t) + \frac{\chi}{\rho R_2^2(t)} \frac{\partial c_2}{\partial \rho} p_2 - \frac{\chi}{R_2^2(t)} \frac{\partial c_2(1,t)}{\partial \rho} \big(1 - p_2(1,t)\big)\Big)\Big] y \cdot \bigtriangledown \hat{p}_2 \\ &+ \Big[\Big(-K_q(c_1) + K(c_1)(1 - \hat{p}_1)\Big) - \Big(-K_q(c_2) + K(c_2)(1 - \hat{p}_2)\Big)\Big] \hat{p}_2 \\ &+ \Big(K_p(c_1) - K_p(c_2)\Big)(1 - \hat{p}_2), \\ g(\rho,t) &= -\chi \hat{p}_2(1 - \hat{p}_2)\Big(\frac{\partial c_1}{\partial \rho} - \frac{\partial c_2}{\partial \rho}\Big). \end{split}$$

We first consider the problem (6.106)-(6.107). By (6.65)-(6.66), (6.37), (6.44), (6.45), (6.87) and $(R_i, p_i) \in X_T$ (i = 1, 2), we get

$$||g_1(t), g_2(t)||_{L^{\infty}[0, T]} \le A_0 \big(||R_1 - R_2||_{C[0, T]} + ||p_1 - p_2||_{C(Q_T)} \big).$$
(6.114)

We easily derive from (6.106) and (6.107) that

$$R^*(t) = \int_0^t g_2(\tau) e^{\int_\tau^t g_1(\tilde{\tau})d\tilde{\tau}} d\tau.$$
 (6.115)

This, along with (6.114), yields

$$\begin{aligned} \|R^*(t)\|_{C^1[0, T]} \\ &\leq Te^{A_0T} \|g_2(t)\|_{L^{\infty}[0, T]} + T^{\frac{\gamma}{2}} \|g_2(t)\|_{C^{\gamma/2}[0, T]} \\ &\leq A_0 \left(T + T^{\frac{\gamma}{2}}\right) \left(\|R_1 - R_2\|_{C^1[0, T]} + \|p_1 - p_2\|_{C^{\gamma, \gamma/2}(Q_T)}\right), \quad (6.116) \end{aligned}$$

here we have used $R_1(0) - R_2(0) = p_1(0) - p_2(0) = c_1(\rho, 0) - c_2(\rho, 0) = 0$ and we have assumed that 0 < T < 1.

Next, we turn to consider the parabolic problem (6.110)-(6.113). By $(R_i, p_i) \in X_T$ (i = 1, 2), (6.88), (6.103), and the parabolic L^p -estimate as before, we have

$$\begin{aligned} \|p^*\|_{W_k^{2,1}(Q_T)} \\ &\leq A_0 \left(\|f\|_{L^k(Q_T)} + \|g\|_{W_k^{1,1}(Q_T)} \right) \\ &\leq A_0 \left(\|R_1 - R_2\|_{C^1[0, T]} + \|p_1 - p_2\|_{C(Q_T)} \right) \left(1 + \|\hat{p}_2\|_{W_k^{2,1}(Q_T)} \right). (6.117) \end{aligned}$$

However, by (6.103), we have

$$\|\hat{p}_2\|_{W_k^{2,1}(Q_T)} \le A_0 \big(1 + \|p_0(\rho)\|_{W^{2,k}(B_1(0))} \big).$$
(6.118)

Combining (6.117) and (6.118) we have

$$\|p^*\|_{W_k^{2,1}(Q_T)} \le A_0 \big(\|R_1 - R_2\|_{C^1[0, T]} + \|p_1 - p_2\|_{C(Q_T)} \big), \tag{6.119}$$

where the constant A_0 depends on $||p_0(\rho)||_{W^{2,k}(B_1(0))}$. Using the embedding $W_k^{2,1}(Q_T) \hookrightarrow C^{1+\lambda,\frac{1+\lambda}{2}}(Q_T)$ $(k > 5, \ \lambda = 1 - \frac{5}{k})$, we find that

$$\|p^*\|_{C^{1+\lambda,\frac{1+\lambda}{2}}(Q_T)} \le A_0 \big(\|R_1 - R_2\|_{C^1[0, T]} + \|p_1 - p_2\|_{C^{\gamma,\gamma/2}(Q_T)}\big).$$
(6.120)

Hence, proceeding as in (6.104) and using (6.111), we have

$$\begin{aligned} \|p^{*}(\rho,t)\|_{C^{\gamma,\gamma/2}(Q_{T})} & (6.121) \\ &\leq \left(2T^{\frac{1+\lambda}{2}} + T^{\frac{1+\lambda-\gamma}{2}}\right) \|p^{*}\|_{C^{1+\lambda,\frac{1+\lambda}{2}}(Q_{T})} + 2\|p_{0}^{*}(\rho)\|_{C^{1}(B_{1}(0))} \\ &= \left(2T^{\frac{1+\lambda}{2}} + T^{\frac{1+\lambda-\gamma}{2}}\right) \|p^{*}\|_{C^{1+\lambda,\frac{1+\lambda}{2}}(Q_{T})} \\ &\leq A_{0}\left(2T^{\frac{1+\lambda}{2}} + T^{\frac{1+\lambda-\gamma}{2}}\right) \left(\|R_{1} - R_{2}\|_{C^{1}[0, T]} + \|p_{1} - p_{2}\|_{C^{\gamma,\gamma/2}(Q_{T})}\right). \end{aligned}$$

Finally, we derive from (6.116) and (6.121) that

$$d((\hat{R}_1, \hat{p}_1), (\hat{R}_2, \hat{p}_2)) \le A_0 \left(2T^{\frac{1+\lambda}{2}} + T^{\frac{1+\lambda-\gamma}{2}}\right) d((R_1, p_1), (R_2, p_2)).$$
(6.122)

This yields that the mapping F is contractive provided T is sufficiently small such that $A_0\left(2T^{\frac{1+\lambda}{2}}+T^{\frac{1+\lambda-\gamma}{2}}\right) < 1$. Hence, F has a unique fixed point (R,p)in X_T . That is, (R,p) together with (c,u) defined by (6.65)-(6.66) and (6.70), is the unique solution of the problem (6.65)-(6.74) for $0 \le t \le T$. Summarizing above results, we obtain

Theorem 6.3. Under the assumptions (6.37)-(6.39), there exists a T > 0which only depends on $\|p_0(\rho)\|_{W^{2,k}(B_1(0))}$, such that the problem (6.65)-(6.72) has a unique solution $(R(t), c(\rho, t), u(\rho, t), p(\rho, t))$ with $R(t) \in C^1[0, T], c(\rho, t) \in C^{2,1}(Q_T), u(\rho, t) \in C^1(Q_T)$ and $p(\rho, t) \in W_k^{2,1}(Q_T)$ (k > 5).

Remark 6.6. The C^1 -smoothness of functions $K_a(c), K_b(c), K_d(c), K_p(c)$ and $K_q(c)$ assumed in (6.37) is used for deriving the estimates (6.114) and (6.117).

6.8 Global Existence

Theorem 6.4. Under the assumptions (6.37)-(6.39), there exists a unique solution (R(t), $c(\rho, t)$, $u(\rho, t)$, $p(\rho, t)$) of the problem (6.65)-(6.72) for all t >0; furthermore, $R(t) \in C^{1}[0, \infty)$, $c(\rho, t) \in C^{2,1}([0, 1] \times [0, \infty))$, $u(\rho, t) \in$ $C^{1}([0, 1] \times [0, \infty))$, $p(\rho, t) \in W_{k}^{2,1}(Q_{T})$ for k > 5 and any T > 0, and

$$0 \le c(\rho, t) \le 1,$$
 (6.123)

$$0 \le p(\rho, t) \le 1,$$
 (6.124)

$$|u(\rho,t)| \le \beta, \tag{6.125}$$

$$R_0 e^{-\beta t} \le R(t) \le R_0 e^{\beta t} \tag{6.126}$$

for some $\beta > 0$.

Proof. Suppose to the contrary that $[0, \tilde{T})$ is the maximum time interval for the existence of the solution. By *a priori* estimates established in Section 6.4, we find that (6.44),(6.45), (6.51), (6.62), (6.63) and (6.77) hold for all $t < \tilde{T}$. Therefore, we have

$$0 \le c(\rho, t) \le 1, \ 0 \le p(\rho, t) \le 1, \text{ for all } t < \tilde{T} \text{ and } 0 \le \rho \le 1, \ (6.127)$$

$$R_0 e^{-A_0 \tilde{T}} \le R(t) \le R_0 e^{A_0 \tilde{T}}, \quad \text{for all } t < \tilde{T} ,$$
 (6.128)

$$\|\frac{u(\rho,t)}{\rho}\|_{L^{\infty}}, \ \|\frac{1}{\rho R^2(t)}\frac{\partial c}{\partial \rho}\|_{L^{\infty}} \le A_0, \quad \text{for all } t < \tilde{T} , \qquad (6.129)$$

$$\|p\|_{W^{2,1}_{k}(Q_{\tilde{T}})} \le M(\tilde{T}),$$
(6.130)

where $M(\tilde{T})$ is some constant which may depend on \tilde{T} .

We take $p(\rho, \tilde{T} - \varepsilon)$ (where $0 < \varepsilon < \tilde{T}$ is arbitrary) as a *new initial data*, then we can extend the solution to $Q_{(\tilde{T}-\varepsilon)+\delta}$ for small $\delta > 0$ proceeding as in the proof of Theorem 6.3. Furthermore, the proof of Theorem 6.3 shows that δ depends only on an upper bound on $\parallel p(\rho, \tilde{T} - \varepsilon) \parallel_{W^{2,k}(B_1(0))}$. By a priori estimate (6.130) we find that δ depends on $A(\tilde{T})$ (but δ is independent of ε), i.e., $\delta = \delta(\tilde{T})$. If we take $\varepsilon < \delta(\tilde{T})$, then we get

$$(\tilde{T} - \varepsilon) + \delta > \tilde{T},$$

which contradicts the assumption that $[0, \tilde{T})$ is the maximum time interval for the existence of the solution. Therefore, the maximum time interval for the existence of the solution is $[0, \infty)$. \Box

Remark 6.7. The estimate (6.130) may break down if the diffusion of the cell types is sent to zero (i.e., D = 0 in (6.67)). To prove that the mapping Fdefined in Section 6.5 is contractive, we need to establish some necessary bound $\frac{\partial p}{\partial \rho}$ (see the estimate (6.117)). However, the hyperbolic equation (6.67) with D = 0 has the feature that the characteristic curves come from the region $\{0 < \rho < 1\}$ at the outer boundary $\rho = 1$. So difficulties will arise on this boundary. This is why the diffusion is so central to our results.

6.9 Summary

In this chapter, we reviewed Tindall and Please's model [163] and Tao's analytical results [150]. The model describes the cell cycle dynamics and chemotactic driven cell movement in a multicellular tumor spheroid which consists of proliferating and quiescent cells. The two types of cells are assumed to move with *different* velocity whereas most partial differential equation models of tumor growth assume that all cells within a tumor have a common spatial velocity profile. This chapter extended Tindall and Please's model to a new one with diffusion of the two cell types. The extended model [150] assumes that cell movement is affected by not only chemotaxis but also diffusion. Noting the *relative* velocity of cells on the outer boundary of the spheroid, we clarified how to impose appropriate no-flux boundary condition for reaction-diffusion-advection equations in a moving-domain. By including the diffusion terms the formation of possible shock should be avoided. Indeed, we have proven the global existence and uniqueness of a solution to the newly extended model. The methods of the proof include a fixed point argument and L^p -theory for parabolic equations with the third boundary condition.

In Tindall and Please's model, as well as the PDE models studied in previous chapters, is confined to a spherical geometry. However, the morphological

instability of a tumor, such as fingering and fragmentation, provides a mechanism for invasion. So, the stability of spherical tumors to asymmetric perturbations and the symmetry-breaking bifurcation from stability to instability are biologically and mathematically interesting topics (see [6, 21, 22, 28, 42, 45, 59, 62, 64], for instance), which are not included in this book.

Chemotaxis-Haptotaxis Modeling for Cancer Invasion

7.1 Introduction

Cancer invasion is a very complex process which involves many various biological mechanisms. In fact, a variety of mathematical models have been proposed for various aspects of cancer invasion, and various attempts of including more relevant biology into models have been made by different researchers. Gatenby and Gawlinski [69] used a reaction-diffusion population competition model to examine how the tumor invades the surrounding normal tissue or extracellular matrix (ECM). They suggested that tumor cells produce lactic acid toxic to normal tissue, and the high acidity leads to the death of the normal tissue, which creates space for tumor cells to proliferate and invade into the surrounding tissue. In contrast to the acid-invasion mechanism, Perumpanani and Byrne [124] found that ECM heterogeneity affects invasion. They proposed a model under the assumptions that ECM is degraded by proteases; in addition to random diffusion, the movement of tumor cells is biased towards a gradient of the non-diffusible ECM, which is referred to

158 7 Chemotaxis-Haptotaxis Modeling for Cancer Invasion

as *haptotaxis*; the protease production is proportional to the product of the tumor cell density and the collagen gel concentration. Chaplain and Anderson [29] also developed a haptotaxis model to describe the interactions between the tumor and surrounding tissue. They made assumptions that the tumor cells produce MDEs to degrade the ECM; the degradation gives rise to the haptotactic movement of the tumor cells. Later on, Chaplain and Lolas [30] suggested that, in addition to random diffusion and haptotactic movement, the migration of cancer cells is biased towards a gradient of the diffusible MDEs, which is referred to as *chemotaxis*. Besides, the proliferation of tumor cells and the re-modeling of ECM are taken into account. Recently, Gerisch and Chaplain [70] developed a novel non-local model which incorporates cell-cell adhesion and cell-matrix adhesion, playing important roles in the tumor invasion process. We note that Szymańska et al. [142] proposed another non-local model which focuses on the role of nonlocal kinetic terms modeling competition for space and degradation. We also mention that Lachowicz [105, 106] constructed some microscopic models for tumor invasion and bridged the microscopic model [106] with the macroscopic model [29].

The qualitative analysis of various models of cancer invasion is mathematically interesting. Walker and Webb [165] examined the issues of global existence and uniqueness for Chaplain and Andersons's model [29]. Tao and Wang [149] and Tao [151] studied the global existence of solutions to the Chaplain and Lolas model [30] for large logistic growth rate of cells in dimension 3 and for any positive logistic growth rate in dimension 2, respectively. Tao and Wang [153] also proved the global existence and boundedness of solutions to a simplified version of the Chaplain and Lolas model. Marciniak-Czochra and Ptashnyk recently [114] proved the uniform boundedness of solutions to the haptotaxis model [29]. Szymańska et al. [142] studied the global existence and uniqueness of solutions to their non-local model. Very recently, Liţcanu and Morales-Rodrigo [110] established the asymptotic behavior of solutions to a simplified haptotaxis model of cancer invasion. We should note that the global existence for the chemotaxis-haptotaxis model [30] is still open for small positive logistic growth rate of cells in dimension 3.

This chapter mainly reviews a chemotaxis-haptotaxis model [30] and its mathematical analysis [153]. Actually, Chaplain and Lolas' model can be regarded as an extension of the classical chemotaxis model which may be first proposed in 1970 by Keller and Segel [99].

The 2 × 2 Keller-Segel model takes into account the density of cells and the concentration of the chemical substance which is assumed to influence the movement of the cell population, and the total mass of cells is formally conserved. The Keller-Segel model has now been greatly extended and studied in the last two decades (see [24, 35, 36, 39, 40, 55, 60, 79, 81, 84, 85, 97, 103, 104, 120, 147, 161, 162, 172] and the references cited therein). The interesting feature of Keller-Segel types of models is the possibility of blow-up of solutions in finite time, which strongly depends on the space dimension and initial mass (see [15, 36, 37, 79, 97, 115, 116, 123, 126, 173], for instance). Jäger and Luckhaus [97], to our knowledge, initiated the technique of L^p estimates in

160 7 Chemotaxis-Haptotaxis Modeling for Cancer Invasion

the analysis of the Keller-Segel system and proved the blow-up of solutions for small initial mass. Blanchet, Dolbeault and Perthame [15] found the critical mass for the blow-up of solutions in the two-dimensional case. Some recent studies show that the nonlinear chemotactic sensitivity function (see [24, 80, 81, 86, 177]), the nonlinear porous medium diffusion (see [104, 138]) and the logistic growth term (see [162, 172, 174]) may prevent the blow-up of solutions. However, Chaplain and Lolas' model under consideration in this paper is a 3×3 system which considers the competition between chemotaxis, haptotaxis and logistic cell growth, and the total mass of cells is *not* conserved.

Chaplain and Lolas' model [30] consists of a parabolic chemotaxis-haptotaxis PDE describing the evolution of tumor cell density, a parabolic PDE governing the evolution of proteolytic enzyme concentration, and an ODE modeling the proteolysis of ECM. Since the proteolytic enzyme diffuses much faster than cancer cells do, in this chapter we will consider a simplified version of Chaplain and Lolas' model in which an elliptic PDE, instead of the above-mentioned parabolic PDE, governs the evolution of proteolytic enzyme concentration. The above quasi-stationary simplification was initially proposed by Jäger and Luckhaus [97], and it was mostly considered in the study of the blow-up of smooth solutions to the classical chemotaxis model (see [97, 115, 123], for instance).

We should note that there exists an important technical difference between a 2×2 parabolic-elliptic chemotaxis system and a 3×3 parabolic-ODE-elliptic chemotaxis-haptotaxis system. For a 2×2 parabolic-elliptic chemotaxis system, one may easily estimate the chemotaxis-related integral term $\int_{\Omega} c^{s-1} \bigtriangledown c \cdot \bigtriangledown u dx$ by directly multiplying the elliptic equation for the chemoattractant concentration by c^s and integrating the product in Ω . However, for a 3×3 parabolic-ODE-elliptic chemotaxis-haptotaxis system, we need to develop new techniques to estimate the chemotaxis-related integral term $\int_{\Omega} c^{s-1} |\bigtriangledown c \cdot \bigtriangledown u | dx$. Here, we should emphasize that we must estimate the term $\int_{\Omega} c^{s-1} |\bigtriangledown c \cdot \bigtriangledown u | dx$ rather than the term $\int_{\Omega} c^{s-1} \bigtriangledown c \cdot \bigtriangledown u dx$.

This chapter is organized as follows. Section 7.2 describes the mathematical model. Section 7.3 proves the local existence and uniqueness of smooth solutions to the model. Finally, Sections 7.4 and 7.5 study the global existence and uniform-in-time boundedness of solutions in three dimensional space for large logistic growth rate of cells and in two dimensional space for any positive logistic growth rate of cells, respectively.

7.2 Mathematical Model

The mathematical model of cancer invasion is involved in the following three key physical variables: the cancer cell density c, the extracellular matrix density v, and the matrix degrading enzyme (MDE) concentration u.

The migration of cancer cells is assumed to undergo random motion, chemotaxis, and haptotaxis. In the absence of any extracellular matrix (ECM), cancer cell proliferation is assumed to follow a logistic growth law. The presence of ECM leads to competition for space between the cancer cells and

162 7 Chemotaxis-Haptotaxis Modeling for Cancer Invasion

the ECM. Hence, the equation describing the evolution of cancer cell density is given as follows [30]:

$$\frac{\partial c}{\partial t} = \underbrace{D_c \bigtriangleup c}_{\text{random motion}} - \underbrace{\bigtriangledown (\chi c \bigtriangledown u)}_{\text{chemotaxis}} - \underbrace{\bigtriangledown (\xi c \bigtriangledown v)}_{\text{haptotaxis}} + \underbrace{\mu c(1 - c - v)}_{\text{proliferation}}, \quad (7.1)$$

where D_c is the random motility coefficient, χ and ξ are the chemotactic and haptotactic coefficients, respectively, μ is the logistic proliferation rate of the cells.

Since ECM is "static", we neglect any diffusion and focus solely on its degradation by MDEs upon contact. For simplicity, we assume that no remodeling of the ECM takes place. The equation modeling the proteolysis of the ECM is therefore given by [30]:

$$\frac{\partial v}{\partial t} = -\underbrace{\delta u v}_{\text{proteolysis}},\tag{7.2}$$

where $\delta > 0$ is a rate parameter of degradation.

The MDE concentration is assumed to be influenced by diffusion, production, and decay. Specifically, MDE is produced by cancer cells, diffuses throughout ECM, and undergoes decay through a simple degradation. Hence, we have the following equation for the MDE concentration [30]:

$$\frac{\partial u}{\partial t} = \underbrace{D_u \bigtriangleup u}_{\text{diffusion}} + \underbrace{\alpha c}_{\text{production}} - \underbrace{\beta u}_{\text{decay}}, \tag{7.3}$$

where D_u , α , and β are assumed to be positive constants.

We define a non-dimensional variable:

$$\tilde{t} = D_c t$$

and new parameters via the following scaling:

$$\tilde{\chi} = \frac{\chi}{D_c}, \ \tilde{\xi} = \frac{\xi}{D_c}, \ \tilde{\mu} = \frac{\mu}{D_c}, \ \tilde{\delta} = \frac{\delta}{D_c}, \ \epsilon = \frac{D_c}{D_u}, \ \tilde{\alpha} = \frac{\alpha}{D_u}, \ \tilde{\beta} = \frac{\beta}{D_u}.$$

Henceforth, we omit the tildes for notational simplicity. The dimensionless governing equations can then be written as follows:

$$\frac{\partial c}{\partial t} = \Delta c - \nabla \cdot (\chi c \nabla u) - \nabla \cdot (\xi c \nabla v) + \mu c (1 - c - v), \qquad (7.4)$$

$$\frac{\partial v}{\partial t} = -\delta u v, \tag{7.5}$$

$$\epsilon \frac{\partial u}{\partial t} = \Delta u + \alpha c - \beta u. \tag{7.6}$$

For the parameter values given in [30], ϵ is typically very small compared with other parameters: α , β , δ , χ , ξ , and μ .

In this chapter we will consider the limit $\epsilon \to 0$ and fix the positive parameters α , β , δ , χ , ξ , and μ (this approach of quasi-steady-state approximation was mostly used to study the classical chemotaxis model; see [97, 123] and references cited therein). Finally, we obtain ($\epsilon = 0$ in (7.6))

$$\frac{\partial c}{\partial t} = \Delta c - \nabla \cdot (\chi c \nabla u) - \nabla \cdot (\xi c \nabla v) + \mu c (1 - c - v), \qquad (7.7)$$

$$\frac{\partial v}{\partial t} = -\delta u v, \tag{7.8}$$

$$0 = \Delta u + \alpha c - \beta u. \tag{7.9}$$

The equations (7.7)-(7.9) are considered on some bounded domain $\Omega \subset \mathcal{R}^d$ (d = 2 or 3) with boundary $\partial \Omega$. To close the system of equations, we need to impose boundary and initial conditions.

Boundary conditions: Guided by the in vitro experimental protocol in

164 7 Chemotaxis-Haptotaxis Modeling for Cancer Invasion

which invasion takes place within an isolated system [124], we assume that there is no-flux of cancer cells or MDEs across the boundary of the domain,

$$-\frac{\partial c}{\partial \nu} + \chi c \frac{\partial u}{\partial \nu} + \xi c \frac{\partial v}{\partial \nu} = 0 \quad \text{on } \partial \Omega \times (0, T),$$
(7.10)

$$\frac{\partial u}{\partial \nu} = 0 \quad \text{on } \partial \Omega \times (0, T),$$
 (7.11)

where ν is the outward normal vector to $\partial\Omega$. We note that the choice of the boundary conditions (7.10) and (7.11) is rather mathematically than biologically motivated. This chapter does not consider possible boundary effects. The boundary conditions (7.10) and (7.11), however, may be justified by the fact that the tumor is far from the boundary of the domain containing the tissue under consideration [31].

Initial conditions: We prescribe the initial data

$$c(x,0) = c_0(x), \quad v(x,0) = v_0(x), \quad x \in \Omega.$$
 (7.12)

For any $0 < T \leq \infty$, we set

$$Q_T = \Omega \times \{ 0 < t < T \}, \quad \Gamma_T = \partial \Omega \times \{ 0 < t < T \}.$$

For consistency, we shall use the following notations:

$$W_p^2(\Omega) = \{ u | u, D_x u, D_x^2 u \in L^p(\Omega) \},\$$
$$W_p^{2,1}(Q_T) = \{ u | u, D_x u, D_x^2 u, D_t u \in L^p(Q_T) \}$$

with norm

$$\| u \|_{W_{p}^{2}(\Omega)} = \| u \|_{L^{p}(\Omega)} + \| D_{x}u \|_{L^{p}(\Omega)} + \| D_{x}^{2}u \|_{L^{p}(\Omega)},$$
$$\| u \|_{W_{p}^{2,1}(Q_{T})} = \| u \|_{L^{p}(Q_{T})} + \| D_{x}u \|_{L^{p}(Q_{T})} + \| D_{x}^{2}u \|_{L^{p}(Q_{T})} + \| D_{t}u \|_{L^{p}(Q_{T})}.$$

in which $p \ge 1$ is integer, T > 0, and the derivatives are in the weak sense.

We denote by $C^{k+\lambda,\theta}(Q_T)$ $(k \ge 0$ integer, $0 < \lambda < 1$, $0 < \theta < 1$) the space of function u(x,t) with finite norm

$$\| u \|_{C^{k+\lambda,\theta}(Q_T)} = \sum_{|j|=0}^k \left[\sup_{Q_T} |D_x^j u| + \langle D_x^j u \rangle_{x, Q_T}^{(\lambda)} + \langle D_x^j u \rangle_{t, Q_T}^{(\theta)} \right]$$

where

$$< u >_{x, Q_T}^{(\lambda)} = \sup_{(x,t), (y,t) \in Q_T} \frac{|u(x,t) - u(y,t)|}{|x - y|^{\lambda}},$$

$$< u >_{t, Q_T}^{(\theta)} = \sup_{(x,t), (x,\tau) \in Q_T} \frac{|u(x,t) - u(x,\tau)|}{|t - \tau|^{\theta}}.$$

We denote by $C^{2+\lambda,1+\theta}(Q_T)$ the space of functions u(x,t) with norm

$$|| u ||_{C^{2+\lambda,\theta}(Q_T)} + || u_t ||_{C^{\lambda,\theta}(Q_T)}.$$

We introduce the following variable change:

$$a = c e^{-\xi v}.\tag{7.13}$$

In terms of the variables a, v, and u, the system (7.7)-(7.12) takes the following form:

$$\frac{\partial a}{\partial t} = e^{-\xi v} \bigtriangledown \cdot (e^{\xi v} \bigtriangledown a) - e^{-\xi v} \bigtriangledown \cdot (\chi e^{\xi v} a \bigtriangledown u) + \xi \delta a uv + \mu a (1 - e^{\xi v} a - v) \quad \text{in } Q_T,$$
(7.14)

$$\frac{\partial v}{\partial t} = -\delta u v \quad \text{in } Q_T, \tag{7.15}$$

$$0 = \Delta u + \alpha e^{\xi v} a - \beta u \quad \text{in } Q_T, \tag{7.16}$$

$$\frac{\partial a}{\partial \nu} = \frac{\partial u}{\partial \nu} = 0 \quad \text{on } \Gamma_T,$$
(7.17)

$$a(x,0) = a_0(x), v(x,0) = v_0(x)$$
 in Ω . (7.18)

166 7 Chemotaxis-Haptotaxis Modeling for Cancer Invasion

Remark 7.1. The L^p estimate techniques for the hapotactic term and the chemotactic term in Eq. (7.7) are quite *different* in three space dimensions, since the ECM density v satisfies the first-order ODE (7.8) whereas the proteolytic enzyme concentration u solves the elliptic equation (7.9). We note that there appears the second-order spatial derivative of v in Eq. (7.7), but this second-order spatial derivative term Δv can be transformed into the time derivative term $\partial v/\partial t$ (i.e., $-\delta uv$ by Eq. (7.8)) under the transformation (7.13). However, it doesn't make sense to transform the second-order spatial derivative term Δu in Eq. (7.7) into the time derivative term $\partial u/\partial t$, since u solves the second-order elliptic equation (7.9). We also note that the *divergence* form of Eq. (7.14) on a is favorable for L^p estimates in three space dimensions (see Section 7.4 below). The transformation (7.13) has already been performed in [39, 55, 60] in order to prove the existence of classical solutions. Here, we should further note that, in order to guarantee that (7.14)-(7.16) and (7.7)-(7.9) are equivalent, we need $(a, v, u) \in C^{2,1}(Q_T) \times C^{2,1}(Q_T) \times C^{2,0}(Q_T)$. Hence, we will study the existence of $C^{2,1}(Q_T) \times C^{2,1}(Q_T) \times C^{2,0}(Q_T)$ -smooth solutions to the system (7.14)-(7.18).

Throughout this chapter we assume that

$$\begin{cases} a_0(x) \ge 0, \ 0 \le v_0(x) \le 1, \\\\ \partial \Omega \in C^{2+\sigma}, a_0(x) \in C^{2+\sigma}(\overline{\Omega}), \ v_0(x) \in C^3(\overline{\Omega}), \\\\ \frac{\partial a_0(x)}{\partial \nu} = \frac{\partial u_0(x)}{\partial \nu} = \frac{\partial v_0(x)}{\partial \nu} = 0 \quad \text{on } \partial \Omega, \end{cases}$$
(7.19)

where $\sigma = 3/4$.

In next section we will prove that the system (7.14)-(7.2) has a unique local (in time) classical solution.

7.3 Local Existence and Uniqueness

For notational convenience, in what follows we denote various constants which depend on T by A, while we denote various constants which are independent of T by A_0 .

Theorem 7.1. Under the assumption (7.2), there exists a unique solution $(a(x,t), v(x,t), u(x,t)) \in C^{2+\sigma,1+\sigma/2}(Q_T) \times C^{2+\sigma,1+\sigma/2}(Q_T) \times C^{2+\sigma,\sigma/2}(Q_T)$ to the system (7.14)-(7.18) for small T > 0 which depends on

$$M := \| c_0(x) \|_{C^{2+\sigma}(\Omega)} + \| v_0(x) \|_{C^{2+\sigma}(\Omega)} + 1$$

Proof. We shall prove the local existence by a fixed point argument. We introduce the Banach space \mathcal{X} of the function c with norms

$$\| c \|_{C^{\sigma, \frac{\sigma}{2}}(\bar{Q}_T)} \quad (0 < T < 1)$$

and a subset

.

$$\mathcal{X}_M = \{ c \in \mathcal{X} : \ c \ge 0, \ \| \ c \|_{C^{\sigma, \frac{\sigma}{2}}(\bar{Q}_T)} \le M \}.$$

Given any $\tilde{c} \in \mathcal{X}_M$, we define a corresponding function $c = F\tilde{c}$ by

$$c = e^{\xi v} a,$$

168 7 Chemotaxis-Haptotaxis Modeling for Cancer Invasion

where (a, v), together with u, satisfies the following system:

$$-\Delta u + \beta u = \alpha \tilde{c} \quad \text{in } Q_T, \tag{7.20}$$

$$\frac{\partial u}{\partial \nu} = 0 \quad \text{on } \Gamma_T, \tag{7.21}$$

$$\frac{\partial v}{\partial t} = -\delta u v \quad \text{in } Q_T, \tag{7.22}$$

$$v(x,0) = v_0(x) \quad \text{in } \Omega, \tag{7.23}$$

$$\frac{\partial a}{\partial t} - \bigtriangleup a + (\chi \bigtriangledown u - \xi \bigtriangledown v) \cdot \bigtriangledown a
+ [\chi \bigtriangleup u + \chi \xi \bigtriangledown v \cdot \bigtriangledown u - \xi \delta uv - \mu (1 - \tilde{c} - v)]a = 0 \quad \text{in } Q_T, (7.24)
\frac{\partial a}{\partial \nu} = 0 \quad \text{on } \Gamma_T, \quad a(x, 0) = a_0(x) \quad \text{in } \Omega.$$
(7.25)

By (7.20)-(7.21) and the standard elliptic Schauder theory [71], we have

$$\| u \|_{C^{2+\sigma,0}(\bar{Q}_T)} \le A_0 \| \tilde{c} \|_{C^{\sigma,0}(\bar{Q}_T)} \le A_0 M.$$
(7.26)

By $\tilde{c} \ge 0$ and the maximum principle [71], we also have

$$u \ge 0 \quad \text{in } Q_T. \tag{7.27}$$

Note that for any $x \in \Omega$ and $0 < t_1 < t_2 < T < 1$,

$$- \bigtriangleup \left(\frac{u(x, t_1) - u(x, t_2)}{|t_1 - t_2|^{\sigma/2}} \right) + \beta \frac{u(x, t_1) - u(x, t_2)}{|t_1 - t_2|^{\sigma/2}} = \alpha \frac{\tilde{c}(x, t_1) - \tilde{c}(x, t_2)}{|t_1 - t_2|^{\sigma/2}} \quad \text{in } Q_T,$$
(7.28)

$$\frac{\partial}{\partial\nu} \left(\frac{u(x, t_1) - u(x, t_2)}{|t_1 - t_2|^{\sigma/2}} \right) = 0 \quad \text{on } \Gamma_T.$$
(7.29)

By (7.28)-(7.29) and the maximum principle, we further have

$$\begin{split} & \max_{\Omega} \frac{\mid u(x, t_1) - u(x, t_2) \mid}{\mid t_1 - t_2 \mid^{\sigma/2}} \le A_0 \parallel \frac{\tilde{c}(x, t_1) - \tilde{c}(x, t_2)}{\mid t_1 - t_2 \mid^{\sigma/2}} \parallel_{L^{\infty}(\Omega)} \\ & \le A_0 \parallel \tilde{c}(x, t) \parallel_{C^{0, \sigma/2}(Q_T)} \le A_0 \parallel \tilde{c}(x, t) \parallel_{C^{\sigma, \sigma/2}(Q_T)} \le A_0 M, \end{split}$$
\mathbf{so}

$$\| u \|_{C^{0,\sigma/2}(\bar{Q}_T)} \le A_0 M. \tag{7.30}$$

We easily conclude from (7.2), (7.22)-(7.23), (7.3)-(7.27), (7.30), and 0 < $T < 1 \mbox{ that}$

$$\| u \|_{C^{2+\sigma,\sigma/2}(\bar{Q}_T)} \le A_0 M, \quad \| v \|_{C^{2+\sigma,\sigma/2}(\bar{Q}_T)} \le A_0 M.$$
(7.31)

This, together with Eq. (7.22), yields

$$\|v\|_{C^{2+\sigma,1+\sigma/2}(\bar{Q}_T)} \le A_0 M.$$
 (7.32)

Furthermore, by (7.2), (7.22)-(7.23), and (7.27), we easily get

$$0 \le v \le 1. \tag{7.33}$$

We now turn to Eq. (7.24). Note that Eq. (7.24) can be rewritten as

$$\frac{\partial a}{\partial t} - \Delta a + (\chi \bigtriangledown u - \xi \bigtriangledown v) \cdot \bigtriangledown a + ha = 0$$
(7.34)

with

$$\| \chi \bigtriangledown u - \xi \bigtriangledown v \|_{C^{\sigma,\sigma/2}(\bar{Q}_T)} \leq B_1,$$

$$\| h \|_{C^{\sigma,\sigma/2}(\bar{Q}_T)}$$

$$= \| \chi \bigtriangleup u + \chi \xi \bigtriangledown v \cdot \bigtriangledown u - \xi \delta uv - \mu (1 - \tilde{c} - v) \|_{C^{\sigma,\sigma/2}(\bar{Q}_T)} \leq B_2 (7.36)$$

by the estimates (7.3), (7.30), (7.32) and $\tilde{c} \in \mathcal{X}_M$; here and in the following we shall denote various constants which depend only on M by B_k (k = 1, 2, ...). Hence, Eq. (7.3) is a *linear* parabolic equation with $C^{\sigma,\sigma/2}(\bar{Q}_T)$ coefficients

and, by the Schauder theory, the parabolic problem (7.3) and (7.25) has a unique solution satisfying

$$\| a \|_{C^{2+\sigma,1+\sigma/2}(\bar{Q}_T)} \le \| a |_{t=0} \|_{C^{2+\sigma}(\Omega)} + B_3$$
$$\le M + B_3 := B_4.$$
(7.37)

By (7.2), (7.24)-(7.25), and the maximum principle [108], we easily get

$$a \ge 0. \tag{7.38}$$

We conclude from (7.32)-(7.33) and (7.37)-(7.38) that

$$c \ge 0, \quad \| c \|_{C^{2+\sigma,1+\sigma/2}(\bar{Q}_T)} \le B_5.$$
 (7.39)

For any function c(x, t), it is easy to check that

$$\| c(x,t) - c(x,0) \|_{C^{\sigma,\sigma/2}(\bar{Q}_T)} \le A_0 \max(T^{\sigma/2}, T^{1-(\sigma/2)}) \| c \|_{C^{2+\sigma,1+\sigma/2}(\bar{Q}_T)}.$$
(7.40)

Combining (7.39) and (7.40), we conclude that if T is sufficiently small (T depends only on M), then

$$\| c(x,t) \|_{C^{\sigma,\sigma/2}(\bar{Q}_T)} \le \| c(x,0) \|_{C^{\sigma,\sigma/2}(Q_T)} + A_0 \max(T^{\sigma/2}, T^{1-(\sigma/2)}) B_5$$
$$\le \| c_0(x) \|_{C^{\sigma}(\Omega)} + 1 \le M.$$
(7.41)

This, together with $c \ge 0$, yields $c \in \mathcal{X}_M$. Hence, F maps \mathcal{X}_M into itself.

Next, proceeding as in the proof of the contraction in Theorem 3.1 in [149], we can prove that F is contractive in \mathcal{X}_M if we take T sufficiently small. By the contraction mapping principle F has a unique fixed point c. Furthermore, we can raise the regularity of c to $C^{2+\sigma,1+\sigma/2}(Q_T)$ by using the parabolic Schauder estimates. \Box

7.4 Global Existence and Boundedness in Dimension 3

We shall need the following Gagliardo-Nirenberg interpolation inequality [57, 75]: Assume that $\Omega \subset R^d$ is a bounded domain, $\partial \Omega \in C^k$, $u(x) \in W_q^k(\Omega) \bigcap L^r(\Omega)$, and $0 \le l \le k$. Then,

$$\| D^{l} u \|_{L^{p}(\Omega)} \leq A_{0} \| u \|_{W^{k}_{q}(\Omega)}^{\theta} \| u \|_{L^{r}(\Omega)}^{1-\theta},$$
(7.42)

where

$$\frac{l}{k} \le \theta \le 1, \ 1 \le q, \ r \le \infty, \ \frac{d}{p} - l = \theta \left(\frac{d}{q} - k\right) + \left(1 - \theta\right) \frac{d}{r}$$

when $k - l - \frac{d}{q}$ is not a nonnegative integer;

$$\theta = \frac{l}{k} < 1, \ 1 < q < \infty, \ 1 < r < \infty$$

when $k - l - \frac{d}{q}$ is a nonnegative integer.

We shall also need the following Sobolev imbedding Theorem (see [32, Theorem 3.5]): Assume that $u \in W_p^{2l,l}(Q_T)$, $\partial \Omega \in C^{2l}$, where l is a positive integer. Then, for $0 \leq r+2s = \mu < 2l$, $p > \frac{d+2}{2l-\mu}$ (where $\frac{d+2}{p}$ is not an integer), we have

$$\| D_t^s D_x^r u \|_{C^{\alpha,\alpha/2}(Q_T)} \le A \| u \|_{W_n^{2l,l}} (Q_T),$$
(7.43)

where $\alpha = 2l - \mu - (d+2)/p$.

To continue the local solution established in the above section to all t > 0, we need to establish some a priori estimates.

Lemma 7.1. There holds

$$a \ge 0, \quad u \ge 0, \quad 0 \le v \le 1.$$
 (7.44)

Proof. The inequality $a \ge 0$ follows from the parabolic maximum principle, and the inequality $u \ge 0$ follows from $a \ge 0$ and the elliptic maximum principle. The inequalities $0 \le v \le 1$ follow from $u \ge 0$ and the maximum principle. This completes the proof of Lemma 7.1. \Box

Lemma 7.2. There holds

$$||a||_{L^{1}(\Omega)} \leq ||c||_{L^{1}(\Omega)} \leq \max(||c_{0}||_{L^{1}(\Omega)}, |\Omega|),$$
 (7.45)

$$\| u \|_{L^{1}(\Omega)} \leq \alpha \beta^{-1} \max(\| c_{0} \|_{L^{1}(\Omega)}, |\Omega|).$$
(7.46)

Proof. Integrating (7.7) in Ω and noting (7.10) and (7.1), we have

$$\frac{d}{dt} \| c \|_{L^{1}(\Omega)} \leq \mu \| c \|_{L^{1}(\Omega)} - \mu \int_{\Omega} c^{2} dx
\leq \mu \| c \|_{L^{1}(\Omega)} - \frac{\mu}{|\Omega|} \| c \|_{L^{1}(\Omega)}^{2},$$
(7.47)

where we have used the Hölder's inequality:

$$\left(\int_{\Omega} c dx\right)^2 \leq \int_{\Omega} 1^2 dx \cdot \int_{\Omega} c^2 dx = |\Omega| \int_{\Omega} c^2 dx.$$

We derive from (7.47) that

$$\| c \|_{L^{1}(\Omega)} \leq \frac{1}{\frac{1}{|\Omega|} + \left(\frac{1}{\|c_{0}\|_{L^{1}(\Omega)}} - \frac{1}{|\Omega|}\right)e^{-\mu t}} \leq \max(\| c_{0} \|_{L^{1}(\Omega)}, |\Omega|).$$
(7.48)

This, along with (7.13) and (7.1), yields (7.45).

7.4 Global Existence and Boundedness in Dimension 3 173

Integrating (7.9) in Ω and noting (7.11), we have

$$\frac{d}{dt} \parallel u \parallel_{L_1(\Omega)} \leq \parallel c \parallel_{L_1(\Omega)} - \parallel u \parallel_{L_1(\Omega)}.$$

This, along with (7.48), yields (7.46). \Box

Up to now we have had the $L^1(\Omega)$ -estimate on a. In the following we shall raise the a priori estimate of a from $L^1(\Omega)$ to $L^2(\Omega)$ and then to $L^4(\Omega)$.

Theorem 7.2. There exists some positive constant

$$\mu^* = \mu^*(\chi, \xi, \alpha, \beta, \delta, |\Omega|, \|a_0(x)\|_{L^2(\Omega)})$$

such that

$$\|a\|_{L^4(\Omega)} \le A_0$$
 (7.49)

for all $t \in (0,T]$ and $\mu \ge \mu^*$.

Proof. For $s \geq 2$, we will perform the $L^s(\Omega)$ -estimate of a. To this end, we need to consider the integral $\int_{\Omega} a^s dx$, which is equivalent to $\int_{\Omega} e^{\xi v} a^s dx$ by the estimate $0 \leq v \leq 1$ in (7.1). There is a strong difference between the cases $\xi > 0$ and $\xi = 0$ when we proceed to estimate the integral $\int_{\Omega} e^{\xi v} a^s dx$ (this point will be addressed more clearly later on). We also note that the quantities a and c are almost equivalent by the transformation (7.13) and $0 \leq v \leq 1$. It should be pointed out that there are bad influences of the haptotaxis contribution ($\xi > 0$) and chemotaxis contribution ($\chi > 0$) in the following computations for getting a priori estimates, whereas there is a good influence of the logistic damping ($\mu > 0$).

We derive from (7.14), (7.15), (7.17) and (7.1) that

 $\frac{d}{dt}$

$$\begin{split} \int_{\Omega} e^{\xi v} a^{s} dx &= \int_{\Omega} \xi e^{\xi v} \frac{\partial v}{\partial t} a^{s} dx + \int_{\Omega} e^{\xi v} s a^{s-1} \frac{\partial a}{\partial t} dx \\ &= -\xi \delta \int_{\Omega} e^{\xi v} a^{s} uv dx + \int_{\Omega} s a^{s-1} \bigtriangledown (e^{\xi v} \bigtriangledown a) dx \\ &- \chi \int_{\Omega} s a^{s-1} \bigtriangledown (e^{\xi v} a \bigtriangledown u) dx + \xi \delta \int_{\Omega} e^{\xi v} s a^{s-1} auv dx \\ &+ \mu \int_{\Omega} e^{\xi v} s a^{s-1} a(1 - e^{\xi v} a - v) dx \\ &\leq -\int_{\Omega} s(s-1) a^{s-2} |\bigtriangledown a|^{2} e^{\xi v} dx \\ &+ \chi \int_{\Omega} s(s-1) a^{s-1} e^{\xi v} \bigtriangledown a \cdot \bigtriangledown u dx \\ &+ \xi \delta s \int_{\Omega} e^{\xi v} a^{s} u dx + \mu s \int_{\Omega} e^{\xi v} a^{s} dx - \mu s \int_{\Omega} e^{2\xi v} a^{s+1} dx \\ &\leq -\frac{4(s-1)}{s} \int_{\Omega} |\bigtriangledown a^{s/2}|^{2} dx \\ &+ \mu e^{\xi} s \int_{\Omega} a^{s} dx - \mu s \int_{\Omega} a^{s+1} dx \\ &+ \xi \delta e^{\xi} s \int_{\Omega} u a^{s} dx \\ &+ \chi e^{\xi} s(s-1) \int_{\Omega} a^{s-1} |\bigtriangledown a \cdot \bigtriangledown u| dx. \end{split}$$
(7.50)

We note that the integral $\int_{\Omega} ua^s dx$ in (7.50) comes from the haptotaxis term in Eq. (7.7), whereas the integral $\int_{\Omega} a^{s-1} | \nabla a \cdot \nabla u | dx$ in (7.50) comes from the chemotaxis term in Eq. (7.7). From the derivation of the inequality (7.50), we find that we need only to estimate the integral $\int_{\Omega} a^{s-1} \nabla a \cdot \nabla u dx$ in order to deal with the chemotaxis term in Eq. (7.7) for the case $\xi = 0$. However, we must estimate the integral $\int_{\Omega} a^{s-1} | \nabla a \cdot \nabla u | dx$ for the case $\xi > 0$. Multiplying Eq. (7.16) by a^s , integrating the product in Ω , and using the no-flux boundary condition $\frac{\partial u}{\partial \nu}|_{\Gamma_T} = 0$ in (7.17), one may easily estimate the integral $\int_{\Omega} a^{s-1} \nabla a \cdot \nabla u dx$ for the case $\xi = 0$. However, it needs many more techniques to estimate

7.4 Global Existence and Boundedness in Dimension 3 175

the integral $\int_{\Omega} a^{s-1} | \nabla a \cdot \nabla u | dx$ for the case $\xi > 0$ (see the remainder of the proof of Lemma 7.2).

The proof of the estimate (7.49) is divided into the following two steps. Step 1: Estimate $||a||_{L^2(\Omega)}$. Taking s = 2 in (7.50), one finds

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^2 dx = -2 \int_{\Omega} |\nabla a|^2 dx + 2\mu e^{\xi} \int_{\Omega} a^2 dx - 2\mu \int_{\Omega} a^3 dx + 2\xi \delta e^{\xi} \int_{\Omega} u a^2 dx + 2\chi e^{\xi} \int_{\Omega} a |\nabla a \cdot \nabla u| dx.$$
(7.51)

We first note that, by the Young inequality

$$yz \le \epsilon y^p + A_0 \epsilon^{-\frac{q}{p}} z^q \quad (y, \ z \ge 0, \ \epsilon > 0, \ p, \ q > 0, \ \frac{1}{p} + \frac{1}{q} = 1),$$
 (7.52)

one may estimate

$$\int_{\Omega} a^2 dx \le \varepsilon \int_{\Omega} a^3 dx + A_0(\varepsilon), \qquad (7.53)$$

with $\varepsilon > 0$ arbitrary.

We next estimate the integral $\int_{\Omega} ua^2 dx$. Applying the Young inequality (7.52), one obtains

$$\int_{\Omega} ua^2 dx \le \int_{\Omega} a^3 dx + A_0 \int_{\Omega} u^3 dx.$$
(7.54)

Using Eq. (7.16), the estimate $0 \le v \le 1$ and the elliptic L^p -estimate [71], one finds

$$\| u \|_{W^{2,3}(\Omega)} \le A_0 \| a \|_{L^3(\Omega)} . \tag{7.55}$$

Combining (7.54) and (7.55), one has

$$\int_{\Omega} ua^2 dx \le A_0 \int_{\Omega} a^3 dx.$$
(7.56)

We now consider the integral $\int_{\Omega} a |\nabla a \cdot \nabla u| dx$. Using the Young inequality (7.52), one finds

$$\int_{\Omega} a | \nabla a \cdot \nabla u | dx$$

$$\leq \epsilon \int_{\Omega} | \nabla a |^{2} dx + A_{0}(\epsilon) \int_{\Omega} a^{2} | \nabla u |^{2} dx$$

$$\leq \epsilon \int_{\Omega} | \nabla a |^{2} dx + A_{0}(\epsilon) \int_{\Omega} a^{3} dx + A_{0}(\epsilon) \int_{\Omega} | \nabla u |^{6} dx.$$
(7.57)

Applying the interpolation inequality (7.42) with l = 1, p = 6, k = 2, q = 3, r = 1, d = 3, $\theta = 1/2$ and using the estimates (7.46) and (7.55), one obtains

$$\| \nabla u \|_{L^{6}(\Omega)} \leq A_{0} \| u \|_{W^{2}_{3}(\Omega)}^{1/2} \| u \|_{L^{1}(\Omega)}^{1/2} \leq A_{0} \| u \|_{W^{2}_{3}(\Omega)}^{1/2} \leq A_{0} \| a \|_{L^{3}(\Omega)}^{1/2},$$

and therefore

$$\| \nabla u \|_{L^{6}(\Omega)}^{6} \leq A_{0} \| a \|_{L^{3}(\Omega)}^{3} .$$
(7.58)

One derives from (7.57) and (7.58) that

$$\int_{\Omega} a | \nabla a \cdot \nabla u | dx \le \epsilon \int_{\Omega} | \nabla a |^2 dx + A_0(\epsilon) \int_{\Omega} a^3 dx.$$
 (7.59)

Inserting (7.53), (7.56), and (7.59) into (7.51), and taking ϵ sufficiently small, one finds

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^2 dx \le -\int_{\Omega} |\bigtriangledown a|^2 dx - (\mu - A_0 \xi \delta e^{\xi} - A_0 \chi e^{\xi}) \int_{\Omega} a^3 dx + A_0 \mu e^{\xi}.$$
(7.60)

Note that

$$\mu - A_0 \xi \delta e^{\xi} - A_0 \chi e^{\xi} \ge 1 \quad \text{if } \mu \ge A_0 (\xi \delta + \chi) e^{\xi} + 1.$$
 (7.61)

Clearly, it follows from the fully explicit bound of μ from below in (7.61) that this bound is worse than for the haptotaxis system without chemotaxis.

Also, this bound is worse than for the Fisher-Kolmogorov-Petrovski-Piskunov equation without chemotaxis/haptotaxis. We further find from the explicit bound of μ from below in (7.61) that this bound becomes worse for faster degradation of the ECM (i.e., for larger δ).

By (7.60), (7.61), the estimate $0 \le v \le 1$, and the Hölder inequality, one obtains

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^{2} dx \leq -\int_{\Omega} a^{3} dx + A_{0} \mu e^{\xi} \\
\leq -\frac{1}{|\Omega|^{\frac{1}{2}}} \left(\int_{\Omega} a^{2} dx\right)^{\frac{3}{2}} + A_{0} \mu e^{\xi} \\
\leq -\frac{1}{e^{\frac{3}{2}\xi} |\Omega|^{\frac{1}{2}}} \left(\int_{\Omega} e^{\xi v} a^{2} dx\right)^{\frac{3}{2}} + A_{0} \mu e^{\xi}.$$
(7.62)

Denote $\tilde{A_0}:=e^{-\frac{3}{2}\xi}|\varOmega|^{-\frac{1}{2}},\,h(t):=\int_{\varOmega}e^{\xi v}a^2dx.$ Then, h(t) satisfies

$$h'(t) \le -\tilde{A}_0 h(t)^{\frac{3}{2}} + A_0 \mu e^{\xi},$$
(7.63)

implying (see [162], for instance)

$$h(t) \le \max\left\{h(0), \ \left(\frac{A_0\mu e^{\xi}}{\tilde{A}_0}\right)^{\frac{2}{3}}\right\}.$$
 (7.64)

Hence,

$$\int_{\Omega} a^2 dx \le A_0,\tag{7.65}$$

where A_0 may depend on $|| a_0 ||_{L^2(\Omega)}$. From the fully explicit bound of $|| a ||_{L^2(\Omega)}$ from above in (7.64), we find that the bound is worse than for the chemotaxis system without haptotaxis ($\xi = 0$) and that this bound is also worse than for the haptotaxis system without chemotaxis ($\chi = 0$) (note that we can choose a smaller μ for the haptotaxis system without chemotaxis ($\chi = 0$) by (7.61)).

Step 2: Estimate $|| a ||_{L^4(\Omega)}$. Taking s = 4 in (7.50), one finds

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^4 dx \leq -3 \int_{\Omega} |\bigtriangledown a^2|^2 dx + 4\mu e^{\xi} \int_{\Omega} a^4 dx - 4\mu \int_{\Omega} a^5 dx + 4\xi \delta e^{\xi} \int_{\Omega} u a^4 dx + 12\chi e^{\xi} \int_{\Omega} a^3 |\bigtriangledown a \cdot \bigtriangledown u| dx.$$
(7.66)

First, employing the Young inequality and proceeding as in Step 1, one has, for $\varepsilon > 0$ arbitrary,

$$\int_{\Omega} a^4 dx \le \varepsilon \int_{\Omega} a^5 dx + A_0(\varepsilon), \tag{7.67}$$

$$\int_{\Omega} u a^4 dx \le A_0 \int_{\Omega} a^5 dx.$$
(7.68)

We now focus on estimating the integral $\int_{\Omega} a^3 | \bigtriangledown a \cdot \bigtriangledown u | dx$ in (7.66). Applying the Young inequality (7.52), one finds

$$\begin{split} &\int_{\Omega} a^{3} |\bigtriangledown a \cdot \bigtriangledown u| dx \\ &\leq \int_{\Omega} (a^{3} |\bigtriangledown a|)^{\frac{10}{9}} dx + A_{0} \int_{\Omega} |\bigtriangledown u|^{10} dx \\ &= \int_{\Omega} (a^{2} \cdot a| \bigtriangledown a|)^{\frac{10}{9}} dx + A_{0} \int_{\Omega} |\bigtriangledown u|^{10} dx \\ &\leq \epsilon \int_{\Omega} (a|\bigtriangledown a|)^{\frac{10}{9} \cdot \frac{9}{5}} dx + A_{0}(\epsilon) \int_{\Omega} a^{\frac{20}{9} \cdot \frac{9}{4}} dx + A_{0} \int_{\Omega} |\bigtriangledown u|^{10} dx \\ &= \frac{\epsilon}{4} \int_{\Omega} |\bigtriangledown a^{2}|^{2} dx + A_{0}(\epsilon) \int_{\Omega} a^{5} dx + A_{0} \int_{\Omega} |\bigtriangledown u|^{10} dx. \end{split}$$
(7.69)

Using Eq. (7.16), the estimates (7.65), $0 \le v \le 1$, and the elliptic L^p -estimate [71], one finds

$$|| u ||_{W_2^2(\Omega)} \le A_0 || a ||_{L^2(\Omega)} \le A_0.$$

This, together with the Sobolev imbedding theorem [71], yields

$$\| \bigtriangledown u \|_{L^{6}(\Omega)} \leq A_{0} \| u \|_{W^{2}_{2}(\Omega)} \leq A_{0}.$$
 (7.70)

Again, by the elliptic L^p -estimate [71], one has

$$\| u \|_{W_{5}^{2}(\Omega)} \leq A_{0} \| a \|_{L^{5}(\Omega)} .$$
(7.71)

Applying the interpolation inequality (7.42) with l = 0, p = 10, k = 1, q = 5, r = 6, d = 3, $\theta = 2/9$ and using the estimates (7.70) and (7.71), one obtains

$$\| \nabla u \|_{L^{10}(\Omega)} \le A_0 \| u \|_{W_5^2(\Omega)}^{2/9} \| \nabla u \|_{L^6(\Omega)}^{7/9}$$

$$\le A_0 \| u \|_{W_5^2(\Omega)}^{2/9}$$

$$\le A_0 \| a \|_{L^5(\Omega)}^{2/9} .$$

This, together with the Young inequality, yields

$$\| \nabla u \|_{L^{10}(\Omega)}^{10} \leq A_0 \| a \|_{L^5(\Omega)}^{20/9} \leq A_0 \| a \|_{L^5(\Omega)}^5 + A_0.$$
 (7.72)

Combining (7.69) and (7.72), one has

$$\int_{\Omega} a^3 |\nabla a \cdot \nabla u| dx \le \frac{\epsilon}{4} \int_{\Omega} |\nabla a^2|^2 dx + A_0(\epsilon) \int_{\Omega} a^5 dx + A_0.$$
(7.73)

Inserting (7.67), (7.68), and (7.73) into (7.66) and taking ϵ sufficiently small, one obtains if $\mu \ge A_0 + 1/4$, then

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^{4} dx \leq -\int_{\Omega} |\nabla a^{2}|^{2} dx - 4(\mu - A_{0}) \int_{\Omega} a^{5} dx + A_{0} \\
\leq -4(\mu - A_{0}) \int_{\Omega} a^{5} dx + A_{0} \\
\leq -\int_{\Omega} a^{5} dx + A_{0} \\
\leq -\frac{1}{|\Omega|^{\frac{1}{4}}} \Big(\int_{\Omega} a^{4} dx\Big)^{\frac{5}{4}} + A_{0} \\
\leq -\frac{1}{e^{\frac{5}{4}\xi} |\Omega|^{\frac{1}{4}}} \Big(\int_{\Omega} e^{\xi v} a^{4} dx\Big)^{\frac{5}{4}} + A_{0}.$$
(7.74)

As before, the inequality (7.74) implies

$$\int_{\Omega} e^{\xi v} a^4 dx \le A_0.$$

This completes the proof of Theorem 7.2. \Box

Theorem 7.3. Assume that $\mu \ge \mu^*$, where μ^* is defined in Theorem 7.2. Then, there holds

$$\| u \|_{C^{\gamma}(\Omega)} \leq A_0, \quad \| v \|_{C^{\gamma}(\Omega)} \leq A \tag{7.75}$$

for any $0 \le \gamma < \frac{5}{4}$.

Proof. It follows from Eq. (7.16), the estimates (7.1) and (7.49), and the elliptic L^p -theory that

$$\| u \|_{W_4^2(\Omega)} \le A_0.$$
 (7.76)

This, together with the Sobolev imbedding theorem, yields

$$\| u \|_{C^{\gamma}(\Omega)} \le A_0 \quad \text{for any } 0 \le \gamma < \frac{5}{4}.$$

$$(7.77)$$

One derives from Eq. (7.15) that

$$v(x,t) = v_0(x)e^{-\delta \int_0^t u(x,s)ds},$$
(7.78)

$$\nabla v = e^{-\delta \int_0^t u(x,s)ds} \nabla v_0 - \delta v_0(x) e^{-\delta \int_0^t u(x,s)ds} \int_0^t \nabla u ds.$$
(7.79)

Combining (7.77)-(7.79), (7.1), and (7.2), one obtains

$$\|v\|_{C^{\gamma}(\Omega)} \le A \quad \text{for any } 0 \le \gamma < \frac{5}{4}.$$

$$(7.80)$$

This completes the proof of Theorem 7.3. $\hfill\square$

7.4 Global Existence and Boundedness in Dimension 3 181

Theorem 7.4. Assume that $\mu \ge \mu^*$, where μ^* is defined in Theorem 7.2. Then, there holds

$$\| a(x,t), v(x,t) \|_{C^{2+\sigma,1+\sigma/2}(Q_T)} \le A, \quad \| u(x,t) \|_{C^{2+\sigma,\sigma/2}(Q_T)} \le A, \quad (7.81)$$

where $\sigma = 3/4$.

Proof. Eq. (7.14) can be rewritten as in the following nondivergence form:

$$a_t - \triangle a - (\xi \bigtriangledown v - \chi \bigtriangledown u) \cdot \bigtriangledown a + [\chi \xi \bigtriangledown u \cdot \bigtriangledown v + \chi \triangle u - \xi \delta uv - \mu(1 - c - v)]a = 0,$$
(7.82)

where

$$\| \xi \bigtriangledown v - \chi \bigtriangledown u \|_{L^{\infty}(Q_T)} \le A, \tag{7.83}$$

$$\|\chi\xi \bigtriangledown u \cdot \bigtriangledown v + \chi \bigtriangleup u - \xi \delta u v - \mu(1 - c - v) \|_{L^4(Q_T)} \le A \qquad (7.84)$$

by the estimates (7.1), (7.75), and (7.76). By (7.82)-(7.84) and the parabolic L^p -estimates [108], one obtains

$$|| a ||_{W_4^{2,1}(Q_T)} \le A.$$

Applying the Sobolev imbedding theorem (7.43) with $l=1, \ p=4, \ d=3, \ r=s=\mu=0,$ one finds

$$\|a\|_{C^{\sigma,\sigma/2}(Q_T)} \le A,$$
 (7.85)

where $\sigma = 3/4$.

One may derive from (7.78), (7.2) and (7.77) that, for any $0 < t_1 < t_2 < T$,

$$\frac{|v(x, t_1) - v(x, t_2)|}{|t_1 - t_2|^{\sigma/2}} = |v_0(x)| \frac{|e^{-\delta \int_0^{t_1} u(x, s)ds} - e^{-\delta \int_0^{t_2} u(x, s)ds}|}{|t_1 - t_2|^{\sigma/2}} = |v_0(x)| |\delta u(x, \eta) e^{-\delta \int_0^{\eta} u(x, s)ds}| \cdot |t_1 - t_2|^{1 - \sigma/2} \quad \text{(for some } t_1 < \eta < t_2) \le A.$$

This, together with (7.80), yields

$$||v||_{C^{\sigma,\sigma/2}(Q_T)} \le A.$$
 (7.86)

Proceeding as in the proof of Theorem 7.1, one obtains from Eq. (7.16), the estimates (7.85) and (7.86), and the elliptic Schauder estimates that

$$\| u \|_{C^{2+\sigma,\sigma/2}(Q_T)} \le A.$$
(7.87)

Again, proceeding as in the proof of Theorem 7.1, one can get

$$\|v\|_{C^{2+\sigma,1+\sigma/2}(Q_T)} \le A.$$
 (7.88)

Finally, one derives from Eq. (7.82), the estimates (7.87) and (7.88), and the parabolic Schauder estimates that

$$|| a ||_{C^{2+\sigma,1+\sigma/2}(Q_T)} \le A.$$

This completes the proof of Theorem 7.4. \Box

The a priori estimate (7.81) allows us to continue the local solution in Theorem 7.1 step-by-step to all t > 0, as done in Chapter 6. Thus, we have the following result of global existence of solutions.

7.4 Global Existence and Boundedness in Dimension 3 183

Theorem 7.5. Under the assumption (7.2), there exists some constant

$$\mu^* = \mu^*(\chi, \xi, \alpha, \beta, \delta, |\Omega|, \|a_0(x)\|_{L^2(\Omega)}) > 0$$

such that, for any $\mu > \mu^*$, the system (7.14)-(7.18) admits a unique global solution $(a, v, u) \in C^{2+\sigma,1+\sigma/2}(Q_T) \times C^{2+\sigma,1+\sigma/2}(Q_T) \times C^{2+\sigma,\sigma/2}(Q_T)$ for any T > 0. Furthermore,

$$a \ge 0, \ u \ge 0, \ 0 \le v \le 1,$$
 (7.89)

$$\| u \|_{L^{\infty}(Q_T)} \le A_0, \quad \| \nabla u \|_{L^{\infty}(Q_T)} \le A_0.$$
 (7.90)

Remark 7.2. If $\mu = 0$, $v \equiv 0$ in (7.7)-(7.9), then blow-up of smooth solutions occurs for large initial mass as aforementioned in Section 7.1 in three dimensions and two dimensions. In next section, we shall prove that global existence holds true for $\mu > 0$ in two dimensions. However, it remains open whether blow-up of classical solutions occurs for small $\mu > 0$ in three dimensions

Furthermore, in the following we will derive the uniform-in-time boundedness of a(x,t) of the global solution.

Theorem 7.6. Let (a, v, u) be the unique global solution of the system (7.14)-(7.18) in Theorem 7.5. Then, there holds

$$||a||_{L^{\infty}(Q_T)} \le A_0 ||a_0||_{L^{\infty}(\Omega)}, \tag{7.91}$$

where A_0 depends on χ , ξ , α , β , δ , μ , and $|\Omega|$.

Proof. We go back to the differential inequality (7.50).

We first estimate the integral $\int_{\Omega} ua^s dx$. By the estimate $u \ge 0$ in (7.89) and the estimate $||u||_{L^{\infty}(Q_T)} \le A_0$ in (7.90), one obtains

$$\int_{\Omega} u a^s dx \le A_0 \int_{\Omega} a^s dx. \tag{7.92}$$

We next consider the integral $\int_{\Omega} a^{s-1} | \bigtriangledown a \cdot \bigtriangledown u | dx$. By the estimate $\| \bigtriangledown$

 $u||_{L^{\infty}(Q_T)} \leq A_0$ in (7.90) and the Cauchy inequality, one has

$$\int_{\Omega} a^{s-1} |\nabla a \cdot \nabla u| dx \leq A_0 \int_{\Omega} a^{s-1} |\nabla a| dx$$

$$= A_0 \int_{\Omega} a^{\frac{s}{2}} \cdot a^{\frac{s}{2}-1} |\nabla a| dx$$

$$\leq A_0 \epsilon \int_{\Omega} a^{s-2} |\nabla a|^2 dx + \frac{A_0}{\epsilon} \int_{\Omega} a^s dx$$

$$= \frac{4A_0 \epsilon}{s^2} \int_{\Omega} |\nabla a^{\frac{s}{2}}|^2 dx + \frac{A_0}{\epsilon} \int_{\Omega} a^s dx. \quad (7.93)$$

Inserting (7.92) and (7.93) into (7.50) and taking ϵ sufficiently small, one finds

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^s dx + \frac{2(s-1)}{s} \int_{\Omega} |\nabla a^{s/2}|^2 dx \le A_0 s(s-1) \int_{\Omega} a^s dx \quad \text{for all } s \ge 2.$$
(7.94)

Noting $0 \le v \le 1$ and using the iterative technique of Alikakos [4], one obtains from (7.94) that

$$\sup_{t \ge 0} \|e^{\xi v(x, t)} a(x, t)\|_{L^{\infty}(\Omega)} \le A_0 \|e^{\xi v_0(x)} a_0(x)\|_{L^{\infty}(\Omega)}.$$
 (7.95)

We also refer to recent references [40, 103] for a detailed derivation of the estimate (7.95). Hence, the proof of Theorem 7.6 is completed. \Box

7.5 Global Existence and Boundedness in dimension 2 185

7.5 Global Existence and Boundedness in dimension 2

Throughout this section, we assume that d = 2. We first note that Lemmas 7.1 and 7.2 are also true for d = 2. Up to now we have had the $L^1(\Omega)$ -estimate on c. In the following we shall raise the a priori estimate of c in the following way: $L^1(\Omega) \to L^3(Q_T) \to L^2(\Omega) \to L^4(Q_T) \to L^3(\Omega)$.

From Eq. (7.7), we have the following two lemmas [151].

Lemma 7.3. There holds

$$\frac{d}{dt} \int_{\Omega} \left[(c+1)\log(c+1) - c \right] dx + \int_{\Omega} \frac{1}{c+1} |\nabla c|^2 dx
\leq \mu \int_{\Omega} \left[(c+1)\log(c+1) - c \right] dx$$

$$+ A_0(\chi + \xi) \Big(\| \Delta u(t) \|_{L^2(\Omega)}^2 + \| \Delta v(t) \|_{L^2(\Omega)}^2 + \| c(t) \|_{L^2(\Omega)}^2 \Big) + A_0 \mu.$$
(7.96)

Proof. We easily derive from (7.8), the boundary condition (7.11), and the assumption $\frac{\partial v_0(x)}{\partial \nu}\Big|_{\partial \Omega} = 0$ in (7.2) that

$$\frac{\partial v}{\partial \nu}\Big|_{\partial \Omega_T} = 0. \tag{7.97}$$

Multiplying Eq. (7.7) by $\log(c+1)$, integrating the product in Ω , and using the no-flux boundary condition (7.10), one obtains

$$\frac{d}{dt} \int_{\Omega} \left[(c+1)\log(c+1) - c \right] dx + \int_{\Omega} \frac{1}{c+1} |\nabla c|^2 dx$$
$$= \chi \int_{\Omega} \frac{c}{c+1} \nabla c \cdot \nabla u \, dx + \xi \int_{\Omega} \frac{c}{c+1} \nabla c \cdot \nabla v \, dx$$
$$+ \mu \int_{\Omega} c(1-c-v)\log(c+1) \, dx. \tag{7.98}$$

Here, one observes that

$$\int_{\Omega} \frac{c}{c+1} \bigtriangledown c \cdot \bigtriangledown u \, dx = \int_{\Omega} \left[\log(c+1) - c \right] \bigtriangleup u \, dx, \tag{7.99}$$

$$\int_{\Omega} \frac{c}{c+1} \bigtriangledown c \cdot \bigtriangledown v \, dx = \int_{\Omega} \left[\log(c+1) - c \right] \bigtriangleup v \, dx, \qquad (7.100)$$

in which we have used the boundary condition (7.11) and (7.97). On the other hand, one observes that

$$\int_{\Omega} \left[\log(c+1) - c \right] \bigtriangleup u \, dx \le A_0(\|\bigtriangleup u(t)\|_{L^2}^2 + \|c(t)\|_{L^2}^2), \quad (7.101) \\
\int_{\Omega} \left[\log(c+1) - c \right] \bigtriangleup v \, dx \le A_0(\|\bigtriangleup v(t)\|_{L^2}^2 + \|c(t)\|_{L^2}^2), \quad (7.102) \\
\int_{\Omega} c(1 - c - v) \log(c+1) \, dx \le \int_{\Omega} c \log(c+1) \, dx \\
\le \int_{\Omega} \left[(c+1) \log(c+1) - c \right] \, dx + \|c\|_{L^1(\Omega)} \\
\le \int_{\Omega} \left[(c+1) \log(c+1) - c \right] \, dx + A_0, \quad (7.103)$$

in which we have used the basic inequality: $\log(1 + c) \leq c$ for any $c \geq 0$ in derivation of estimates (7.101) and (7.102), and we have used estimates (7.1) and (7.45) in derivation of estimate (7.103). Taking into account all above estimates, one obtains from (7.96) from (7.98). \Box

Lemma 7.4. Assume that $\mu > 0$ and that the following estimate

$$\| (c+1)\log(c+1) \|_{L^{1}(\Omega)} \le A$$
(7.104)

is true; then there holds

$$\frac{d}{dt} \int_{\Omega} c^{2} dx + \int_{\Omega} |\nabla c|^{2} dx$$

$$\leq \epsilon \Big(\|\nabla \Delta u(t)\|_{L^{2}(\Omega)}^{2} + \|\nabla \Delta v(t)\|_{L^{2}(\Omega)}^{2} \Big)$$

$$+ A \epsilon^{-1/2} g(\epsilon^{-1}) + A, \qquad (7.105)$$

where $\epsilon > 0$ is any number and $g(\cdot)$ is some increasing function.

Proof. This lemma is the core of the argument concerning global existence in the two-dimensional case.

Multiply Eq. (7.7) by c, integrate the product in Ω , and use the no-flux boundary condition (7.10). One obtains

$$\frac{1}{2}\frac{d}{dt}\int_{\Omega}c^{2}dx + \int_{\Omega}|\bigtriangledown c|^{2}dx = \chi\int_{\Omega}c\bigtriangledown c\cdot\bigtriangledown udx + \xi\int_{\Omega}c\bigtriangledown c\cdot\bigtriangledown vdx + \mu\int_{\Omega}c^{2}(1-c-v)dx.$$
(7.106)

Note that

$$\int_{\Omega} c \bigtriangledown c \cdot \bigtriangledown u \, dx = -\frac{1}{2} \int_{\Omega} c^2 \bigtriangleup u \, dx, \qquad (7.107)$$

$$\int_{\Omega} c \bigtriangledown c \cdot \bigtriangledown v \, dx = -\frac{1}{2} \int_{\Omega} c^2 \bigtriangleup v \, dx \tag{7.108}$$

by the boundary conditions (7.11) and (7.97).

Applying the Gagliardo–Nirenberg interpolation inequality (7.42) with l = 1, p = 3, k = q = r = d = 2, and $\theta = 2/3$, one obtains

$$\| \bigtriangleup u \|_{L^{3}(\Omega)} \le A_{0} \| u \|_{H^{3}(\Omega)}^{2/3} \| u \|_{H^{1}(\Omega)}^{1/3} \quad \text{for } u \in H^{3}(\Omega),$$
(7.109)

where $H^q(\Omega) := W^q_2(\Omega)$.

We shall also need the following interpolation inequality proved by Biler, Hebisch, and Nadzieja [13]:

$$\|c\|_{L^{3}(\Omega)}^{3} \leq \varepsilon \|c\|_{H^{1}(\Omega)}^{2} \| (c+1)\log(c+1) \|_{L^{1}(\Omega)} + g(\varepsilon^{-1})\|c\|_{L^{1}(\Omega)}$$
(7.110)

for $c \ge 0$ and $c \in H^1(\Omega)$, where $\varepsilon > 0$ is any number and $g(\cdot)$ is some increasing function.

By (7.45), Eq. (7.9) and the regularity theory for elliptic equations in two dimensions, one has

$$\| \nabla u \|_{L^2(\Omega)} \le A_0 \| c \|_{L^1(\Omega)} \le A_0.$$
(7.111)

Applying the interpolation inequality (7.42) and using the estimates (7.46)and (7.11), one obtains

$$\| u \|_{L^{2}(\Omega)} \le A_{0} \| u \|_{W_{1}^{1}\Omega} \| u \|_{L^{1}(\Omega)} \le A_{0}.$$
(7.112)

Now we estimate $\left|\int_{\Omega} c^2 \bigtriangleup u \ dx\right|$:

$$\begin{split} \left| \int_{\Omega} c^{2} \bigtriangleup u \, dx \right| &\leq A_{0} \|c\|_{L^{3}(\Omega)}^{2} \|\bigtriangleup u\|_{L^{3}(\Omega)} \quad \text{(by Hölder's inequality)} \\ &\leq A_{0} \|c\|_{L^{3}(\Omega)}^{2} \|u\|_{H^{3}(\Omega)}^{2/3} \|u\|_{H^{1}(\Omega)}^{1/3} \quad \text{(by (7.109))} \\ &\leq A_{0} \|c\|_{L^{3}(\Omega)}^{2} \|u\|_{H^{3}(\Omega)}^{2/3} \quad \text{(by (7.111) and (7.112))} \\ &\leq A_{0} \left[\varepsilon \|c\|_{H^{1}(\Omega)}^{2} \|(c+1)\log(c+1)\|_{L^{1}(\Omega)} + g(\varepsilon^{-1})\|c\|_{L^{1}(\Omega)} \right]^{2/3} \|u\|_{H^{3}(\Omega)}^{2/3} \end{split}$$

(by Biler, Hebisch, and Nadzieja's interpolation inequality (7.110))

$$\leq A \Big[\varepsilon \| c \|_{H^{1}(\Omega)}^{2} + g(\varepsilon^{-1}) \Big]^{2/3} \| u \|_{H^{3}(\Omega)}^{2/3} \quad (by (7.45) \text{ and } (7.104)) \\ \leq \varepsilon \| u \|_{H^{3}(\Omega)}^{2} + A \varepsilon^{-1/2} \Big[\varepsilon \| c \|_{H^{1}(\Omega)}^{2} + g(\varepsilon^{-1}) \Big] \quad (by \text{ Young's inequality}) \\ = \varepsilon \| u \|_{H^{3}(\Omega)}^{2} + A \varepsilon^{1/2} \| c \|_{H^{1}(\Omega)}^{2} + A \varepsilon^{-1/2} g(\varepsilon^{-1}).$$
(7.113)

Similarly, one has

$$\left|\int_{\Omega} c^2 \bigtriangleup v \, dx\right| \le \varepsilon \|v\|_{H^3(\Omega)}^2 + A\varepsilon^{1/2} \|c\|_{H^1(\Omega)}^2 + A\varepsilon^{-1/2} g(\varepsilon^{-1}).$$
(7.114)

On the other hand, by Young's inequality,

$$\int_{\Omega} c^2 dx \le \varepsilon \int_{\Omega} c^3 dx + A_0(\varepsilon) |\Omega|.$$
(7.115)

Taking into account above estimates (7.106)-(7.108), (7.113)-(7.115), and using assumption $\mu > 0$, one has

7.5 Global Existence and Boundedness in dimension 2 189

$$\frac{1}{2} \frac{d}{dt} \int_{\Omega} c^2 dx + \int_{\Omega} |\nabla c|^2 dx$$

$$\leq \varepsilon \Big(\|u\|_{H^3(\Omega)}^2 + \|v\|_{H^3(\Omega)}^2 \Big) + A\varepsilon^{-1/2}g(\varepsilon^{-1}) + A_0$$

$$\leq \varepsilon \Big(\|\nabla \Delta u\|_{L^2(\Omega)}^2 + \|\nabla \Delta v\|_{L^2(\Omega)}^2 \Big) + A\varepsilon^{-1/2}g(\varepsilon^{-1}) + A$$

for sufficiently small $\varepsilon > 0$, where we have used the facts that

$$\begin{aligned} \|u\|_{H^{3}(\Omega)} &\leq A_{0} \Big(\|\bigtriangledown \bigtriangleup u\|_{L^{2}(\Omega)} + \|u\|_{H^{1}(\Omega)} \Big) \quad \text{(for } u \in H^{3}(\Omega) \text{ with } \frac{\partial u}{\partial \nu} \Big|_{\partial \Omega} = 0 \\ &\leq A_{0} \|\bigtriangledown \bigtriangleup u\|_{L^{2}(\Omega)} + A \qquad \text{(by (7.111) and (7.112))} \end{aligned}$$

and that

$$\|v\|_{H^{3}(\Omega)} \leq A_{0} \| \bigtriangledown \bigtriangleup v \|_{L^{2}(\Omega)} + A \quad (\text{for } v \in H^{3}(\Omega) \text{ with } \frac{\partial v}{\partial \nu} \Big|_{\partial \Omega} = 0).$$

This completes the proof of Lemma 7.4. \Box

Lemma 7.5. Assume that d = 2 and $\mu > 0$. Then there holds

$$\| c \|_{L^3(Q_T)} \le A. \tag{7.116}$$

Proof. The proof is divided into the following eight steps. In Steps 1-6, we will give attention to the dependency of the bounds of some estimates on the parameters μ, χ and ξ . From (7.45) and its proof, we find that the bound from above of $||c||_{L^1(\Omega)}$ is independent of the parameters μ, χ , and ξ . In the following we may assume that $\mu > 0$ is *small*, say, $0 < \mu < 1$, since we can proceed the estimates for large $\mu > 0$ in two dimensions as in three dimensions in Section 7.4.

Step 1: Estimate $||c||^2_{L^2(Q_T)}$. Integrating Eq. (7.7) in $\Omega \times [0, t]$ $(t \leq T)$, noting the no-flux boundary condition (7.10), and using the assumption that

 $\mu > 0$ and the estimates (7.1) and (7.45), one obtains

$$\int_0^t \int_{\Omega} c^2 dx ds \le A\mu^{-1}.$$
(7.117)

Step 2: Estimate $\| \bigtriangleup u \|_{L^2(Q_T)}^2$. Multiply Eq. (7.9) by $\bigtriangleup u$, integrate the product in Ω , and use the no-flux boundary condition (7.11). Then

$$\int_{\Omega} |\bigtriangleup u|^2 dx + \beta \int_{\Omega} |\bigtriangledown u|^2 dx = -\alpha \int_{\Omega} c \bigtriangleup u dx.$$

Cauchy's inequality allows us to write

$$\frac{1}{2}\int_{\Omega}|\bigtriangleup u|^{2}dx+\beta\int_{\Omega}|\bigtriangledown u|^{2}dx\leq \frac{\alpha^{2}}{2}\int_{\Omega}c^{2}dx,$$

which gives

$$\int_{\Omega} |\bigtriangleup u|^2 dx \le \alpha^2 \int_{\Omega} c^2 dx.$$

This, together with the estimate (7.117), yields

$$\int_0^t \int_{\Omega} |\bigtriangleup u|^2 dx ds \le A\mu^{-1}.$$
(7.118)

Step 3: Estimate $\| \bigtriangledown u \|_{L^2(\Omega)}^2$. We notice by the standard results in the regularity theory for elliptic equations that from Eq. (7.9) and the L^1 -estimate (7.45) of c we have

$$\sup_{t \in [0,T]} \| \nabla u(\cdot,t) \|_{L^q(\Omega)} \le A_0 \sup_{t \in [0,T]} \| c(\cdot,t) \|_{L^1(\Omega)} \le A_0$$
(7.119)

where

$$q = \frac{d}{d-1},$$

which gives

$$\| \nabla u(\cdot, t) \|_{L^2(\Omega)} \le A_0.$$
 (7.120)

7.5 Global Existence and Boundedness in dimension 2 191

Step 4: Estimate $\| \bigtriangledown v \|_{L^2(\Omega)}^2$. Note that Eq. (7.8) can be rewritten as

$$v(x, t) = v_0(x)e^{-\delta \int_0^t u(x, s)ds},$$
(7.121)

and therefore

$$\nabla v = e^{-\delta \int_0^t u(x, s)ds} \nabla v_0 - \delta v_0(x) e^{-\delta \int_0^t u(x, s)ds} \int_0^t \nabla u ds.$$
(7.122)

Using the estimates (7.1) and (7.120), one obtains from (7.122) that

$$\int_{\Omega} |\nabla v|^2 dx \le A. \tag{7.123}$$

Step 5: Estimate $\| \bigtriangleup v \|_{L^2(Q_T)}^2$. One derives from (7.122) that

$$\Delta v = e^{-\delta \int_0^t u(x, s)ds} \Delta v_0 - 2\delta e^{-\delta \int_0^t u(x, s)ds} \int_0^t \nabla u \cdot \nabla v_0 ds$$
$$+ \delta^2 v_0(x) e^{-\delta \int_0^t u(x, s)ds} \Big(\int_0^t \nabla u ds \Big)^2$$
$$-\delta v_0(x) e^{-\delta \int_0^t u(x, s)ds} \int_0^t \Delta u ds.$$
(7.124)

By Eq. (7.9) and the elliptic L^p -theory, one finds

$$\|u\|_{H^2(\Omega)} \le A_0 \|c\|_{L^2(\Omega)}.$$
(7.125)

This, together with the estimate (7.117), yields

$$\int_0^t \|u\|_{H^2(\Omega)}^2 ds \le A_0 \int_0^t \|c\|_{L^2(\Omega)}^2 ds \le A\mu^{-1}.$$
(7.126)

Applying the Gagliardo–Nirenberg interpolation inequality (7.42) with l = 0, p = 4, k = 1, q = r = d = 2, $\theta = 1/2$ and using the estimate (7.120), one obtains

$$\| \nabla u \|_{L^{4}(\Omega)} \le A_{0} \| u \|_{H^{2}(\Omega)}^{1/2} \| \nabla u \|_{L^{2}(\Omega)}^{1/2} \le A_{0} \| u \|_{H^{2}(\Omega)}^{1/2}.$$

This, together with the estimate (7.126), yields

$$\int_{0}^{t} \| \nabla u \|_{L^{4}(\Omega)}^{4} ds \le A_{0} \int_{0}^{t} \| u \|_{H^{2}(\Omega)}^{2} ds \le A \mu^{-1}.$$
 (7.127)

Finally, using the estimates (7.1), (7.118), (7.120) and (7.127), one derives from (7.124) that

$$\int_0^t \int_{\Omega} |\Delta v|^2 dx ds \le A\mu^{-1}.$$
(7.128)

Step 6: Estimate $||(c+1)\log(c+1)||_{L^1(\Omega)}$. Lemma 7.3, together with Gronwall's lemma and the estimates (7.117), (7.118), and (7.128), yields

$$\int_{\Omega} \left[(c+1)\log(c+1) - c \right] dx \le A(\chi + \xi + 1)\mu^{-1}.$$
 (7.129)

Here we should note that $(c+1)\log(c+1) - c \ge 0$ for any $c \ge 0$. Combining (7.45) and (7.129), one has

$$\int_{\Omega} (c+1)\log(c+1)dx \le A(\chi+\xi+1)\mu^{-1}.$$
(7.130)

Clearly, the bound from above of $||(c+1)\log(c+1)||_{L^1(\Omega)}$ in (7.130) strongly depends on μ , and it becomes unbounded as $\mu \to 0^+$. This is why the assumption that $\mu > 0$ is so crucial to our results of global existence in two dimensions.

Step 7: Estimate $\| \bigtriangledown c \|_{L^2(Q_T)}$. By Lemma 7.4 and the estimate (7.130), one has

$$\frac{d}{dt} \int_{\Omega} c^2 dx + \int_{\Omega} |\nabla c|^2 dx$$

$$\leq \epsilon \Big(\|\nabla \Delta u\|_{L^2(\Omega)}^2 + \|\nabla \Delta v\|_{L^2(\Omega)}^2 \Big) + A \epsilon^{-1/2} g(\epsilon^{-1}) + A. \quad (7.131)$$

7.5 Global Existence and Boundedness in dimension 2 193

We first estimate $\| \bigtriangledown \bigtriangleup u \|_{L^2(\Omega)}$. We derive from (7.122), the boundary condition (7.11), and the assumption $\frac{\partial v_0(x)}{\partial \nu} |_{\partial \Omega} = 0$ in (7.2) that

$$\frac{\partial v}{\partial \nu}\Big|_{\Gamma_T} = 0. \tag{7.132}$$

This, together with the boundary conditions (7.10) and (7.11), further yields

$$\frac{\partial c}{\partial \nu}\Big|_{\Gamma_T} = 0. \tag{7.133}$$

Next, multiply Eq. (7.9) by $\triangle^2 u$ and integrate the products. Using integration by parts, one obtains

$$\int_{\Omega} |\nabla \bigtriangleup u|^2 dx + \beta \int_{\Omega} |\bigtriangleup u|^2 dx$$
$$= -\alpha \int_{\Omega} \nabla c \cdot \nabla \bigtriangleup u dx$$
$$\leq \frac{1}{2} \int_{\Omega} |\nabla \bigtriangleup u|^2 dx + \frac{\alpha^2}{2} \int_{\Omega} |\nabla c|^2 dx, \qquad (7.134)$$

where we have used the following compatibility condition

$$\frac{\partial \bigtriangleup u}{\partial \nu}\Big|_{\Gamma_T} = \beta \frac{\partial u}{\partial \nu}\Big|_{\Gamma_T} - \alpha \frac{\partial c}{\partial \nu}\Big|_{\Gamma_T} = 0$$
(7.135)

by (7.9), (7.11), and (7.133). One further derives from (7.134) that

$$\int_{\Omega} |\nabla \bigtriangleup u|^2 dx \le \alpha^2 \int_{\Omega} |\nabla c|^2 dx.$$
(7.136)

We are now in a position to consider $\| \bigtriangledown \bigtriangleup v \|_{L^2(\Omega)}^2$. One can derive from (7.124), the assumption $v_0(x) \in C^3(\overline{\Omega})$ in (7.2), the estimate $u \ge 0$, and the estimates (7.118), (7.120), and (7.127) that

$$\| \bigtriangledown \bigtriangleup v \|_{L^2(\Omega)}^2 \le A + A \int_0^t \int_{\Omega} | \bigtriangledown \bigtriangleup u |^2 dx ds.$$
 (7.137)

Inserting (7.136) and (7.137) into (7.131), one obtains

$$\frac{d}{dt}\int_{\Omega}c^{2}dx + (1-\alpha^{2}\epsilon)\int_{\Omega}|\nabla c|^{2}dx \leq A\epsilon\int_{0}^{t}\int_{\Omega}|\nabla c|^{2}dxds + A(\epsilon).$$

Integrating with respect to the variable t in both sides of the above inequality and taking $\epsilon > 0$ sufficiently small, one finds

$$\int_0^T \int_{\Omega} |\nabla c|^2 dx dt \le A.$$
(7.138)

Step 8: Estimate $||c||_{L^3(Q_T)}$. Applying the Gagliardo–Nirenberg inequality (7.42) with l = 0, k = 1, p = 3, q = d = 2, r = 1, $\theta = 2/3$ and using the estimate (7.45), one obtains

$$\begin{aligned} \|c(t)\|_{L^{3}(\Omega)} &\leq A_{0} \|c(t)\|_{W^{1}_{2}(\Omega)}^{2/3} \|c(t)\|_{L^{1}(\Omega)}^{1/3} \\ &\leq A_{0} \|c(t)\|_{W^{1}_{2}(\Omega)}^{2/3}. \end{aligned}$$

This, together with the estimate (7.117) and (7.138), yields

$$\int_0^t \|c(s)\|_{L^3(\Omega)}^3 ds \le A_0 \int_0^t \|c(s)\|_{W_2^1(\Omega)}^2 ds \le A.$$

This completes the proof of Lemma 7.5. \Box

Lemma 7.6. Assume that d = 2 and $\mu > 0$. Then there holds

$$\| \nabla u \|_{L^6(Q_T)} \le A.$$
 (7.139)

Proof. Applying the interpolation inequality (7.42) with l = 0, p = d = 2, k = q = r = 1, $\theta = 1$ and using the estimates (7.46) and (7.120), one obtains

$$||u||_{L^{2}(\Omega)} \leq A_{0} ||u||_{W_{1}^{1}(\Omega)} ||u||_{L^{1}(\Omega)} \leq A_{0}.$$
(7.140)

Combining (7.136) and (7.138), one also has

$$\int_0^T \int_{\Omega} |\nabla \bigtriangleup u|^2 dx dt \le A.$$
(7.141)

Note the fact that

$$\|u\|_{H^{3}(\Omega)} \leq A_{0} \Big(\|\nabla \Delta u\|_{L^{2}(\Omega)} + \|u\|_{H^{1}(\Omega)} \Big) \quad \text{for } u \in H^{3}(\Omega) \text{ with } \frac{\partial u}{\partial \nu} \Big|_{\partial \Omega} = 0.$$

This, together with (7.120), (7.140) and (7.41), yields

$$\int_{0}^{T} \|u(t)\|_{H^{3}(\Omega)}^{2} \le A.$$
(7.142)

Applying the interpolation inequality (7.42) with l = 0, p = 6, k = q = r = d = 2, $\theta = 1/3$ and using the estimate (7.120), one obtains

$$\|\nabla u\|_{L^{6}(\Omega)} \leq A_{0} \|\nabla u\|_{H^{2}(\Omega)}^{1/3} \|\nabla u\|_{L^{2}(\Omega)}^{2/3} \leq A_{0} \|u\|_{H^{3}(\Omega)}^{1/3}.$$
 (7.143)

Combining (7.142) and (7.143), one finds

$$\int_0^T \| \nabla u(t) \|_{L^6(\Omega)}^6 dt \le A \int_0^T \| u(t) \|_{H^3(\Omega)}^2 dt \le A$$

This completes the proof of Lemma 7.6. \Box

Lemma 7.7. Assume that d = 2 and $\mu > 0$. Then there holds

$$\|a\|_{L^2(\Omega)} \le A. \tag{7.144}$$

Proof. Setting s = 2 in the inequality (7.50), one finds

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^2 dx \leq -2 \int_{\Omega} |\bigtriangledown a|^2 dx + 2\mu e^{\xi} \int_{\Omega} a^2 dx + 2\delta \xi e^{\xi} \int_{\Omega} u a^2 dx + 2\chi e^{\xi} \int_{\Omega} a |\bigtriangledown a \cdot \bigtriangledown u| dx. \quad (7.145)$$

We first estimate the integral $\int_{\Omega} ua^2 dx$. Using the Young inequality and the elliptic L^p -theory as before, one has

$$\int_{\Omega} ua^2 dx \le \frac{1}{3} \int_{\Omega} u^3 dx + \frac{2}{3} \int_{\Omega} a^3 dx \le A_0 \int_{\Omega} a^3 dx.$$
(7.146)

We now consider the integral $\int_{\Omega} a | \bigtriangledown a \cdot \bigtriangledown u | dx$ in (7.145). Applying the Young inequality (7.52), one finds

$$\int_{\Omega} a | \nabla a \cdot \nabla u | dx$$

$$\leq \epsilon \int_{\Omega} (a | \nabla a |)^{\frac{6}{5}} dx + A_0(\epsilon) \int_{\Omega} |\nabla u|^6 dx$$

$$\leq \epsilon \int_{\Omega} |\nabla a|^{\frac{6}{5} \cdot \frac{5}{3}} + A_0(\epsilon) \int_{\Omega} a^{\frac{6}{5} \cdot \frac{5}{2}} dx + A_0(\epsilon) \int_{\Omega} |\nabla u|^6 dx$$

$$= \epsilon \int_{\Omega} |\nabla a|^2 dx + A_0(\epsilon) \int_{\Omega} a^3 dx + A_0(\epsilon) \int_{\Omega} |\nabla u|^6 dx. \quad (7.147)$$

Inserting (7.146) and (7.147) into (7.145), taking ϵ sufficiently small, using $1 \leq e^{\xi v} \leq e^{\xi}$, and noting $\int_{\Omega} a^2 dx \leq A_0 + A_0 \int_{\Omega} a^3 dx$, one obtains

$$\frac{d}{dt}\int_{\Omega}e^{\xi v}a^{2}dx \leq A_{0} + A_{0}\int_{\Omega}a^{3}dx + A_{0}\int_{\Omega}|\nabla u|^{6}dx.$$

This, together with Lemmas 7.5 and 7.6, yields

$$\begin{split} \int_{\Omega} e^{\xi v} a^2 dx &\leq \int_{\Omega} e^{\xi v_0(x)} a_0^2(x) dx + A_0 T \\ &+ A_0 \int_0^T \int_{\Omega} a^3 dx dt + A_0 \int_0^T \int_{\Omega} |\nabla u|^6 dx dt \\ &\leq A. \end{split}$$

This completes the proof of Lemma 7.7. $\hfill\square$

Lemma 7.8. Assume that d = 2 and $\mu > 0$. Then there hold

7.5 Global Existence and Boundedness in dimension 2 197

$$\|c\|_{L^4(Q_T)} \le A,\tag{7.148}$$

$$\| \nabla u \|_{L^q(\Omega)} \le A \quad \text{for any } 1 < q < \infty.$$
(7.149)

Proof. Applying the interpolation inequality (7.42) with l = 0, p = 4, q = r = d = 2, $\theta = 1/2$ and using Lemma 7.7 and the estimate (7.1), one finds

$$\|c\|_{L^4(\Omega)} \le A_0 \|c\|_{H^1(\Omega)}^{1/2} \|c\|_{L^2(\Omega)}^{1/2} \le A \|c\|_{H^1(\Omega)}^{1/2}.$$

Hence, using the estimates (7.138) and (7.144), one obtains

$$\int_{0}^{T} \|c(t)\|_{L^{4}(\Omega)}^{4} dt \leq A \int_{0}^{t} \|c(t)\|_{H^{1}(\Omega)}^{2} dt$$
$$\leq A \int_{0}^{T} \|\nabla c(t)\|_{L^{2}(\Omega)}^{2} dt + A \int_{0}^{T} \|c(t)\|_{L^{2}(\Omega)}^{2} dt$$
$$\leq A.$$

So, the estimate (7.148) holds.

We now turn to prove the estimate (7.149). Note that Eq. (7.9) can be rewritten as follows:

$$-\bigtriangleup u + \beta u = \alpha c \in L^2(\Omega) \tag{7.150}$$

by Lemma 7.7. We conclude from (7.150), (7.11), and the standard elliptic L^p -theory [71] that $u \in W_2^2(\Omega)$. This, together with Sobolev imbedding theorem, yields the estimate (7.149). Hence, Lemma 7.8 is proved. \Box

Lemma 7.9. Assume that d = 2 and $\mu > 0$. Then there holds

$$\|a\|_{L^3(\Omega)} \le A. \tag{7.151}$$

Proof. Setting s = 3 in the inequality (7.50), one finds

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^{3} dx \leq -\frac{8}{3} \int_{\Omega} |\nabla a^{3/2}|^{2} dx + 3\mu e^{\xi} \int_{\Omega} a^{3} dx + 3\xi \delta e^{\xi} \int_{\Omega} u a^{3} dx + 6\chi e^{\xi} \int_{\Omega} a^{2} |\nabla a \cdot \nabla u| dx.$$
(7.152)

We first estimate the integral $\int_{\Omega} ua^3 dx$. Applying the Young inequality and using the the elliptic L^p -theory as before, one finds

$$\int_{\Omega} u a^3 dx \le \frac{1}{4} \int_{\Omega} u^4 dx + \frac{3}{4} \int_{\Omega} a^4 dx \le A_0 \int_{\Omega} a^4 dx.$$
(7.153)

We now turn to the integral $\int_{\Omega} a^2 | \bigtriangledown a \cdot \bigtriangledown u | dx$ in (7.152). Applying the Young inequality and using the estimate (7.149), one finds

$$\begin{split} \int_{\Omega} a^2 | \nabla a \cdot \nabla u | dx &\leq \epsilon \int_{\Omega} (a^2 | \nabla a |)^{\frac{12}{11}} dx + A(\epsilon) \int_{\Omega} | \nabla u |^{12} dx \\ &\leq \epsilon \int_{\Omega} (a^{\frac{3}{2}} \cdot a^{\frac{1}{2}} | \nabla a |)^{\frac{12}{11}} dx + A(\epsilon) \\ &\leq \epsilon \int_{\Omega} (a^{\frac{1}{2}} | \nabla a |)^{\frac{12}{11} \cdot \frac{11}{6}} dx + A(\epsilon) \int_{\Omega} a^{\frac{3}{2} \cdot \frac{12}{11} \cdot \frac{11}{5}} dx + A(\epsilon) \\ &= \frac{4\epsilon}{9} \int_{\Omega} | \nabla a^{\frac{3}{2}} |^2 dx + A(\epsilon) \int_{\Omega} a^{\frac{18}{5}} dx \\ &\leq \epsilon \int_{\Omega} | \nabla a^{\frac{3}{2}} |^2 dx + A(\epsilon) \int_{\Omega} a^4 dx + A(\epsilon). \end{split}$$
(7.154)

Finally, inserting (7.153) and (7.154) into the inequality (7.152), noting $\int_{\Omega} a^3 dx \leq A_0 + A_0 \int_{\Omega} a^4 dx$, and taking ϵ sufficiently small, one obtains

$$\frac{d}{dt} \int_{\Omega} e^{\xi v} a^3 dx \le A + \int_{\Omega} a^4 dx$$

Integrating with respect to t in both sides of above inequality and using the estimate (7.148), one finds

$$\int_{\Omega} e^{\xi v} a^3 dx \le \int_{\Omega} e^{\xi v_0(x)} a_0^3(x) dx + \int_0^T \int_{\Omega} a^4 dx dt \le A.$$

This completes the proof of Lemma 7.9. \Box

7.5 Global Existence and Boundedness in dimension 2 199

Remark 7.3. It is almost impossible to find an explicit bound from above in terms of μ, χ , and ξ in the estimate (7.151), since this bound is involved in all the bounds appeared in the estimates (7.104)-(7.149) and it concerns many complicated computations. However, the bound from above of $||(c + 1) \log(c + 1)||_{L^1(\Omega)}$ in (7.130) has already given us a good understanding of the above-mentioned dependency on the parameters μ, χ and ξ . Roughly speaking, the bound from above in the estimate (7.151) will probably be directly proportional to the following quantity: $(\chi + \xi)\mu^{-1}$. Therefore, we conclude that the assumption that $\mu > 0$ is central to our results of global existence in two dimensions.

Using the estimate (7.151) and proceeding as in the proof of Theorem 7.3 -7.4 and Theorems 7.5 - 7.6, one has the following main results of this section.

Theorem 7.7. Under the assumptions (7.2) and that d = 2 and $\mu > 0$, there exists a unique global solution

$$(a, v, u) \in C^{2+\tilde{\sigma}, 1+\tilde{\sigma}/2}(Q_T) \times C^{2+\tilde{\sigma}, 1+\tilde{\sigma}/2}(Q_T) \times C^{2+\tilde{\sigma}, \tilde{\sigma}/2}(Q_T) \quad (\tilde{\sigma} := \frac{2}{3})$$

of the system (7.14)-(7.18) for any T > 0. Furthermore,

$$a \ge 0, \quad u \ge 0, \quad 0 \le v \le 1,$$
 (7.155)

$$\|u\|_{L^{\infty}(Q_T)} \le A_0, \quad \|\bigtriangledown u\|_{L^{\infty}(Q_T)} \le A_0, \tag{7.156}$$

$$\|a\|_{L^{\infty}(Q_T)} \le A_0 \|a_0\|_{L^{\infty}(\Omega)}.$$
(7.157)

Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion

8.1 Introduction

In Chapter 7, we reviewed a simplified version of the Chaplain and Lolas' model [30]. However, the original model is a 3×3 parabolic-ODE-parabolic chemotaxis-haptotaxis system. The global existence and uniqueness of classical solutions to this model has been proved for any $\mu \ge 0$ (where μ is the logistic growth rate of cancer cells) in one space dimension (see [149]), for any $\mu > 0$ in two space dimensions (see [151]) and for large μ in three space dimensions (see [149]). In addition to global existence and uniqueness, the uniform-in-time boundedness of solutions to a simplified 3×3 parabolic-ODE-elliptic chemotaxis-haptotaxis system has been proved for any $\mu > 0$ in two space dimensions and for large μ in three space dimensions (see [153]), which has been reviewed in last chapter. We should note that the global existence is still open for *small* $\mu > 0$ in three space dimensions for the parabolic-ODE-parabolic chemotaxis-haptotaxis system and the parabolic-ODE-elliptic chemotaxis-haptotaxis system. When $\mu = 0$, the solution of Chaplain and Lolas' model may blow up 202 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion in finite time (see [149, Section 6]). However, it is obvious that the blow-up of cancer cell density in a finite time is biologically irrelevant. Hence, we need to deal with the following problem, how can we reasonably modify the Chaplain and Lolas model [30] to obtain the global existence? This is the concern of the present chapter.

Recently, Tao and Cui [155] extended the Chaplain and Lolas' model to a new one with nonlinear density-dependent chemotaxis and haptotaxis, and studied the global existence and boundedness of solutions to this newly extended model.

This chapter will review the extended model and its mathematical analysis. This chapter is organized into five sections. Section 8.2 presents the mathematical model. Section 8.3 proves the local existence and uniqueness of solutions. Section 8.4 completes the proof of global existence. Finally, Section ?? shows the boundedness of solutions to a chemotaxis-haptotaxis model with volume-filling.

8.2 Mathematical Model

The mathematical model of cancer invasion is involved in the following three physical variables: cancer cell density c(x,t), ECM density v(x,t), and MDE concentration u(x,t).

The migration of cancer cells is assumed to be governed by random motion, chemotaxis and haptotaxis. In the absence of any ECM, cancer cell proliferation is assumed to be typically logistic. The presence of ECM leads to competition for space between the cancer cells and the ECM. Hence, the equation describing the evolution of cancer cell density reads (see [30])

$$\frac{\partial c}{\partial t} = \underbrace{D_c \bigtriangleup c}_{\text{random motion}} - \underbrace{\bigtriangledown \cdot (V_1(c) \bigtriangledown u)}_{\text{chemotaxis}} - \underbrace{\bigtriangledown \cdot (V_2(c) \bigtriangledown v)}_{\text{haptotaxis}} + \underbrace{\mu c(1 - c - v)}_{\text{proliferation}}, (8.1)$$

where D_c is the random diffusion coefficient, $V_1(c)$ and $V_2(c)$ are the density-dependent chemotactic and haptotactic sensitivity functions, respectively, and μ is the logistic proliferation rate of the cells.

Since ECM is "static," we neglect any diffusion and focus solely on its degradation by MDEs upon contact; for simplicity, we assume that no remodeling of the ECM takes palce. Hence, the equation modeling the proteolysis of ECM is therefore given by (see [30])

$$\frac{\partial v}{\partial t} = -\underbrace{\delta u v}_{\text{proteolysis}},\tag{8.2}$$

where $\delta > 0$ is a rate parameter of degradation.

MDE is produced by cancer cells, diffuses throughout ECM, and undergoes decay through simple degradation. Hence, the equation for MDE concentration is (see [30])

$$\frac{\partial u}{\partial t} = \underbrace{D_u \bigtriangleup u}_{\text{diffusion}} + \underbrace{\alpha c}_{\text{production}} - \underbrace{\beta u}_{\text{decay}}, \tag{8.3}$$

where D_u , α and β are assumed to be positive constants.

Throughout this paper we will assume that

$$V_i(c) \in C^1([0, +\infty)), V_i(c) \ge 0, V_i(0) = 0, \text{ and } V'_i(c) \text{ is Lipschitz continuous,}$$

$$(8.4)$$

204 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion where i = 1, 2. Here we should note that it is necessary for the global existence of C^2 -smooth solutions of Eqs. (8.1)-(8.3) to assume that $V'_1(c)$ and $V'_2(c)$ are Lipschitz continuous (see [156]).

In Chaplain and Lolas' original model [30], it is assumed that $V_1(c) = \chi c$ and $V_2(c) = \xi c$ (χ , $\xi > 0$ are some constants). For this choice of $V_1(c)$ and $V_2(c)$, although the assumption (8.4) is satisfied, the solution of the model may blow up in finite time as afore-mentioned. However, the blow-up of cancer density in finite time is biologically irrelevant. Hence, we would like to slightly modify the choice of $V_1(c)$ and $V_2(c)$ such that the modified model has a unique global solution, which excludes the possibility of a blow-up in finite time. To this end, in addition to the assumption (8.4), we will assume that

$$V_1(c)$$
 and $V_2(c)$ are bounded for any $c \ge 0$. (8.5)

For example, we may take $V_1(c) = \frac{\chi c}{1+\varepsilon_1 c}$ and $V_2(c) = \frac{\xi c}{1+\varepsilon_2 c}$ ($\varepsilon_1, \varepsilon_2 > 0$ are small constants; see [81], for instance). Clearly $V_1(c) \to \chi c$ as $\varepsilon_1 \to 0$ and $V_2(c) \to \xi c$ as $\varepsilon_2 \to 0$. For this choice of $V_1(c)$ and $V_2(c)$, the assumptions (8.4) and (8.5) are both satisfied. Another choice of $V_1(c)$ and $V_2(c)$ satisfying (8.5) is that $V_1(c) \equiv 0$ and $V_2(c) \equiv 0$ for $c \ge c_m$, which has a clear biologically relevant interpretation: the cancer cells stop to accumulate at a given point of the tumor tissue after their density attains a maximal density c_m . A similar assumption for a prey-taxis sensitivity function was made in [2].

However, for typical volume-filling chemotactic-haptotactic functions $V_1(c) = \chi c(1-c/\gamma)$ and $V_2(c) = \xi c(1-c/\gamma)$ ($\gamma \ge 1$ denotes the maximal cell density;
see [81]), the assumption (8.5) is not satisfied. However, these specific forms of $V_1(c)$ and $V_2(c)$ will be in favor of the proofs of global existence and boundedness of solutions to the model (see Section ?? below).

The Eqs. (8.1)-(8.3) are considered on some bounded domain $\Omega \subset \mathcal{R}^3$ with boundary $\partial \Omega$. To close the system of equations, we need to impose boundary and initial conditions.

Boundary conditions: Guided by the in vitro experimental protocol in which invasion takes place within an isolated system [124], we assume that there is no-flux of cancer cells or MDEs across the boundary of the domain

$$-D_c \frac{\partial c}{\partial \nu} + V_1(c) \frac{\partial u}{\partial \nu} + V_2(c) \frac{\partial v}{\partial \nu} = 0 \quad \text{on } \partial \Omega \times (0, \infty),$$
(8.6)

$$\frac{\partial u}{\partial \nu} = 0 \quad \text{on } \partial \Omega \times (0, \infty), \tag{8.7}$$

where ν is the outward normal vector to $\partial \Omega$.

Initial conditions: We prescribe the initial data

$$c(x,0) = c_0(x), \quad v(x,0) = v_0(x), \quad u(x,0) = u_0(x), \quad x \in \Omega.$$
 (8.8)

For any $0 < T < \infty$ we set

$$\Omega_T = \Omega \times \{ 0 < t < T \}, \qquad \partial \Omega_T = \partial \Omega \times \{ 0 < t < T \},$$

To simplify the formulae, throughout this chapter we suppose that

$$D_c = \delta = D_u = \alpha = \beta = 1. \tag{8.9}$$

However, we will keep the key model parameter μ , since our result on global existence will depend on the presence of logistic damping.

Throughout this chapter we assume that

$$\begin{cases} c_0(x) \ge 0, \quad 0 \le v_0(x) \le 1, \quad u_0(x) \ge 0, \\\\ \partial \Omega \in C^{2+\sigma}, \quad 0 < \sigma < 1, \\\\ c_0(x), v_0(x), u_0(x) \in C^{2+\sigma}(\overline{\Omega}), \\\\ \frac{\partial c_0(x)}{\partial \nu} = \frac{\partial v_0(x)}{\partial \nu} = \frac{\partial u_0(x)}{\partial \nu} = 0 \quad \text{on } \partial \Omega. \end{cases}$$

$$(8.10)$$

Note that Eq. (8.2) can be rewritten as

$$v = v_0(x)e^{-\int_0^t u(x,s)ds}$$

and therefore

$$\nabla v = e^{-\int_0^t u(x,s)ds} \nabla v_0 - v_0(x)e^{-\int_0^t u(x,s)ds} \int_0^t \nabla u \ ds.$$

This, along with (8.7) and $\frac{\partial v_0(x)}{\partial \nu} = 0$ on $\partial \Omega$ in (8.10), yields

$$\frac{\partial v}{\partial \nu} = 0 \qquad \text{on } \partial \Omega_T. \tag{8.11}$$

We then conclude from (8.6), (8.7), and (8.11) that

$$\frac{\partial c}{\partial \nu} = 0 \qquad \text{on } \partial \Omega_T.$$
 (8.12)

8.3 Local Existence and Uniqueness

Using the assumptions (8.9) and the boundary conditions (8.11) and (8.12), the problem (8.1)-(8.3) and (8.6)-(8.2) takes the following form:

$$c_t = \Delta c - \nabla \cdot (V_1(c) \nabla u) - \nabla \cdot (V_2(c) \nabla v) + \mu c(1 - c - v) \quad \text{in } \Omega_T, \quad (8.13)$$

$$v_t = -uv \qquad \qquad \text{in } \Omega_T, \quad (8.14)$$

$$u_t = \Delta u + c - u \qquad \qquad \text{in } \Omega_T, \quad (8.15)$$

$$\frac{\partial c}{\partial \nu} = \frac{\partial v}{\partial \nu} = \frac{\partial u}{\partial \nu} = 0 \qquad \text{on } \partial \Omega_T, \qquad (8.16)$$

$$c(x,0) = c_0(x), v(x,0) = v_0(x), u(x,0) = u_0(x)$$
 in Ω . (8.17)

For notational convenience, in what follows we denote various constants which are independent of T by A_0 , whereas we denote various constants which depend on T by A.

In the following, under the assumptions (8.4) and (8.10), we shall prove that the system (8.13)-(8.17) has a unique local (in time) smooth solution for any $\mu \ge 0$.

Theorem 8.1. For any $\mu \geq 0$, under the assumptions (8.4) and (8.10), there exists a unique solution $(c, v, u) \in \left(C^{2+\sigma, 1+\sigma/2}(\Omega_T)\right)^3$ of the system (8.13)-(8.17) for some small T > 0 which depends on $\| (c_0(x), v_0(x), u_0(x)) \|_{C^{2+\sigma}(\overline{\Omega})}$. Proof. We shall prove the local existence by a fixed point argument. We

introduce the Banach space X of function c with norm

$$||c||_X = ||c||_{C^{1+\sigma,\sigma/2}(\bar{\Omega}_T)} \qquad (0 < T < 1)$$

and a subset

$$X_M = \left\{ c \in X : \|c\|_{C^{1+\sigma,\sigma/2}(\bar{\Omega}_T)} \le M \right\},$$

where

$$M := \parallel c_0(x) \parallel_{C^{2+\sigma}(\overline{\Omega})} + \parallel v_0(x) \parallel_{C^{2+\sigma}(\overline{\Omega})} + \parallel u_0(x) \parallel_{C^{2+\sigma}(\overline{\Omega})} + 1.$$

Given any $c \in X_M$, we define a corresponding function $\tilde{c} = Fc$, where \tilde{c} , together with u and v, satisfies the following system:

$$u_t - \Delta u + u = c \qquad \qquad \text{in } \Omega_T, \qquad (8.18)$$

$$\frac{\partial u}{\partial \nu} = 0 \qquad \qquad \text{on } \partial \Omega_T, \qquad (8.19)$$

$$u(x,0) = u_0(x) \qquad \qquad \text{in } \Omega, \tag{8.20}$$

$$v = v_0(x)e^{-\int_0^t u(x,s)ds} \qquad \text{in } \Omega_T, \qquad (8.21)$$

$$\tilde{c}_t - \Delta \tilde{c} - \mu (1 - c - v) \tilde{c} = f(c, u, v) \qquad \text{in } \Omega_T, \qquad (8.22)$$

$$\frac{\partial \tilde{c}}{\partial \nu} = 0 \qquad \qquad \text{on } \partial \Omega_T, \qquad (8.23)$$

$$\tilde{c}(x,0) = c_0(x)$$
 in Ω . (8.24)

where

$$f(c, u, v) := - \bigtriangledown \cdot (V_1(c) \bigtriangledown u) - \bigtriangledown \cdot (V_2(c) \bigtriangledown v)$$
$$= -V_1(c) \bigtriangleup u - V_2(c) \bigtriangleup v$$
$$-(V_1'(c) \bigtriangledown u + V_2'(c) \bigtriangledown v) \cdot \bigtriangledown c.$$
(8.25)

We first consider the linear parabolic problem (8.18)-(8.20). By $c \in X_M, 0 < T < 1$, (8.10), the maximum principle, and the Schauder theory (see [108], for instance), the problem (8.18)-(8.20) has a unique solution u satisfying

8.3 Local Existence and Uniqueness 209

$$\begin{aligned} \|u\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_{T})} &\leq A_0 \Big(\|u\|_{C^0(\bar{\Omega}_{T})} + \|u_0\|_{C^{2+\sigma}(\bar{\Omega})} + \|c\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_{T})} \Big) \\ &\leq A_0 \Big(\|u_0\|_{C^0(\bar{\Omega})} + A_0 T \|c\|_{C^0(\bar{\Omega}_{T})} \\ &+ \|u_0\|_{C^{2+\sigma}(\bar{\Omega})} + \|c\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_{T})} \Big) \\ &\leq A_0 M. \end{aligned}$$

$$(8.26)$$

We easily derive from (8.21) that

$$\nabla v = e^{-\int_{0}^{t} u(x,s)ds} \nabla v_{0} - v_{0}(x)e^{-\int_{0}^{t} u(x,s)ds} \int_{0}^{t} \nabla u \, ds, \qquad (8.27)$$
$$\Delta v = e^{-\int_{0}^{t} u(x,s)ds} \Delta v_{0}$$
$$-2e^{-\int_{0}^{t} u(x,s)ds} \int_{0}^{t} \nabla u \cdot \nabla v_{0}ds + v_{0}(x)e^{-\int_{0}^{t} u(x,s)ds} \left(\int_{0}^{t} \nabla u ds\right)^{2}$$
$$-v_{0}(x)e^{-\int_{0}^{t} u(x,s)ds} \int_{0}^{t} \Delta u \, ds. \qquad (8.28)$$

Using $v_0(x) \in C^{2+\sigma}(\overline{\Omega})$, (8.26), and 0 < T < 1, we obtain from (8.21), (8.27), and (8.28) that

$$\|v\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_T)} \le A_0 M. \tag{8.29}$$

We now turn to the linear parabolic problem (8.22)-(8.24). Using $c \in X_M$, (8.25), (8.26), and (8.29) and noting $V'_1(c)$ and $V'_2(c)$ are Lipschitz continuous, we have

$$||f||_{C^{\sigma,\sigma/2}(\bar{\Omega}_T)} \le A_0 M,$$
 (8.30)

$$\|1 - c - v\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_T)} \le A_0 M.$$
(8.31)

Hence, by 0 < T < 1 and the parabolic Schauder theory as before, the problem (8.22)-(8.24) admits a unique solution \tilde{c} satisfying

$$\|\tilde{c}\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_T)} \le A_0 \Big(\|c_0\|_{C^{2+\sigma}(\bar{\Omega})} + \|f\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_T)} \Big) \le A_0 M.$$
(8.32)

By direct calculations, we find that for any function \tilde{c} ,

$$\|\tilde{c}(x,t) - \tilde{c}(x,0)\|_{C^{1+\sigma,\sigma/2}(\bar{\Omega}_T)} \le A_0 \max(T^{\sigma/2}, T^{1/2}) \|\tilde{c}\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_T)}.$$

If we further take T = T(M) sufficiently small, then by (8.32)

$$\|\tilde{c}(x,t)\|_{C^{1+\sigma,\sigma/2}(\bar{\Omega}_T)} \le \|\tilde{c}(x,0)\|_{C^{1+\sigma}(\bar{\Omega})} + A_0 \max(T^{\sigma/2}, T^{1/2})M$$
$$\le \|\tilde{c}(x,0)\|_{C^{1+\sigma}(\bar{\Omega})} + 1$$
$$\le M.$$

Hence, $\tilde{c} \in X_M$, i.e. F maps X_M into itself.

We are now in a position to show that F is contractive. Take c_1, c_2 in X_M and set $\tilde{c}_1 \equiv Fc_1, \tilde{c}_2 \equiv Fc_2$. We derive from (8.18)-(8.20) that

$$\partial_t (u_1 - u_2) - \triangle (u_1 - u_2) + (u_1 - u_2) = c_1 - c_2 \quad \text{in } \Omega_T, \quad (8.33)$$

$$\frac{\partial(u_1 - u_2)}{\partial \nu} = 0 \qquad \qquad \text{on } \partial \Omega_T, \qquad (8.34)$$

$$(u_1 - u_2)(x, 0) = 0$$
 in Ω . (8.35)

Proceeding as in the proof of (8.26), we have

$$\|u_1 - u_2\|_{C^{2+\sigma, 1+\sigma/2}(\bar{\Omega}_T)} \le A_0 \|c_1 - c_2\|_{C^{\sigma, \sigma/2}(\bar{\Omega}_T)}.$$
(8.36)

This, along with (8.21), (8.27), (8.28), and 0 < T < 1, yields

$$\|v_1 - v_2\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_T)} \le A_0 \|u_1 - u_2\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_T)} \le A_0 \|c_1 - c_2\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_T)}.$$
(8.37)

Next, we derive from (8.22)-(8.25) that

$$\partial_t (\tilde{c}_1 - \tilde{c}_2) - \triangle (\tilde{c}_1 - \tilde{c}_2) - \mu (1 - c_1 - v_1) (\tilde{c}_1 - \tilde{c}_2) = h \quad \text{in } \Omega_T, \qquad (8.38)$$

$$\frac{\partial(\tilde{c}_1 - \tilde{c}_2)}{\partial\nu} = 0 \qquad \text{on } \partial\Omega_T, \quad (8.39)$$

$$(\tilde{c}_1 - \tilde{c}_2)(x, 0) = 0$$
 in Ω , (8.40)

where

$$\begin{split} h &:= -\mu \tilde{c}_2[(c_1 - c_2) + (v_1 - v_2)] \\ &- V_1(c_1) \bigtriangleup (u_1 - u_2) + (V_1(c_2) - V_1(c_1)) \bigtriangleup u_2 \\ &- V_2(c_1) \bigtriangleup (v_1 - v_2) + (V_2(c_2) - V_2(c_1)) \bigtriangleup v_2 \\ &- V_1'(c_1) \bigtriangledown c_1 \cdot \bigtriangledown (u_1 - u_2) + (V_1'(c_2) \bigtriangledown c_2 - V_1'(c_1) \bigtriangledown c_1) \cdot \bigtriangledown u_2 \\ &- V_2'(c_1) \bigtriangledown c_1 \cdot \bigtriangledown (v_1 - v_2) + (V_2'(c_2) \bigtriangledown c_2 - V_1'(c_1) \bigtriangledown c_1) \cdot \bigtriangledown v_2. \end{split}$$

Noting $V'_1(c)$ and $V'_2(c)$ are Lipschitz continuous and using (8.4), (8.26), (8.29), (8.32), (8.36), and (8.37), we find that

$$\|h\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_{T})} \leq A_{0} \Big(\|c_{1}-c_{2}\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_{T})} + \|u_{1}-u_{2}\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_{T})} + \|v_{1}-v_{2}\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_{T})}\Big)$$

$$\leq A_{0}\|c_{1}-c_{2}\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_{T})}.$$
(8.41)

This, along with (8.38)-(8.40) and the parabolic Schauder theory, yields

$$\|\tilde{c}_1 - \tilde{c}_2\|_{C^{2+\sigma, 1+\sigma/2}(\bar{\Omega}_T)} \le A_0 \|c_1 - c_2\|_{C^{\sigma, \sigma/2}(\bar{\Omega}_T)}.$$
(8.42)

Noting $(\tilde{c}_1 - \tilde{c}_2)(x, 0) \equiv 0$ and proceeding as before, we have

$$\begin{aligned} \|\tilde{c}_{1} - \tilde{c}_{2}\|_{C^{1+\sigma,\sigma/2}(\bar{\Omega}_{T})} &\equiv \|(\tilde{c}_{1} - \tilde{c}_{2})(x,t) - (\tilde{c}_{1} - \tilde{c}_{2})(x,0)\|_{C^{1+\sigma,\sigma/2}(\bar{\Omega}_{T})} \\ &\leq A_{0} \max(T^{\sigma/2}, T^{1/2}) \|\tilde{c}_{1} - \tilde{c}_{2}\|_{C^{2+\sigma,1+\sigma/2}(\bar{\Omega}_{T})} \\ &\leq A_{0} \max(T^{\sigma/2}, T^{1/2}) \|c_{1} - c_{2}\|_{C^{\sigma,\sigma/2}(\bar{\Omega}_{T})}. \end{aligned}$$
(8.43)

Finally, taking T sufficiently small such that

$$A_0 \max(T^{\sigma/2}, T^{1/2}) \le 1/2,$$

we conclude from (8.43) that F is contractive in X_M . By the contraction mapping theorem, F has a unique fixed point c in X_M . This completes the proof of Theorem 8.1. \Box

8.4 A Priori Estimates and Global Existence

To continue the local solution in Theorem 8.1 to all t > 0, we need to establish some *a priori* estimates. Throughout this section, in addition to the assumptions (8.4) and (8.10), we assume that the assumption (8.5) holds.

Noting $V_1(0) = V_2(0) = 0$, $c_0(x) \ge 0$, $0 \le v_0(x) \le 1$, and $u_0(x) \ge 0$, and using the maximum principle, we easily prove the following lemma.

Lemma 8.1. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17), then there holds

$$c \ge 0, \quad 0 \le v \le 1, \quad u \ge 0.$$
 (8.44)

Lemma 8.2. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that 8.4 A Priori Estimates and Global Existence 213

$$\mu > 0, \tag{8.45}$$

then there hold

$$\|c\|_{L^1(\Omega)} \le A_0, \tag{8.46}$$

$$\|u\|_{L^1(\Omega)} \le A_0, \tag{8.47}$$

$$\|\nabla u\|_{L^2(\Omega)} \le A_0,\tag{8.48}$$

$$\|\nabla v\|_{L^2(\Omega)} \le A. \tag{8.49}$$

Proof. Integrating Eqs. (8.13) and (8.11) over Ω and proceeding as in the proof of Lemma 7.2 in Chapter 7, we easily prove the estimates (8.46) and (8.47).

We next turn to prove the estimate (8.48). Integrating Eq. (8.13) over Ω and using (8.16) and (8.44), we have

$$\frac{d}{dt} \|c\|_{L^{1}(\Omega)} \le \mu \|c\|_{L^{1}(\Omega)} - \mu \int_{\Omega} c^{2} dx.$$
(8.50)

Multiplying Eq. (8.11) by $- \bigtriangleup u$ and integrating over Ω , we find that

$$\begin{split} &\frac{1}{2}\frac{d}{dt}\int_{\Omega}|\bigtriangledown u|^{2}dx+\int_{\Omega}|\bigtriangleup u|^{2}dx+\int_{\Omega}|\bigtriangledown u|^{2}dx\\ &=-\int_{\Omega}c\bigtriangleup udx\leq\frac{1}{4}\int_{\Omega}c^{2}dx+\int_{\Omega}|\bigtriangleup u|^{2}dx. \end{split}$$

So,

$$\frac{1}{2}\frac{d}{dt}\int_{\Omega}|\nabla u|^{2}dx+\int_{\Omega}|\nabla u|^{2}dx\leq\frac{1}{4}\int_{\Omega}c^{2}dx.$$

Combining this with (8.50), we get

$$\begin{aligned} &\frac{d}{dt} \Big(\frac{1}{2} \int_{\Omega} |\bigtriangledown u|^2 dx + \frac{1}{4\mu} \int_{\Omega} c \ dx \Big) + 2 \Big(\frac{1}{2} \int_{\Omega} |\bigtriangledown u|^2 dx + \frac{1}{4\mu} \int_{\Omega} c \ dx \Big) \\ &\leq \Big(\frac{1}{4} + \frac{1}{2\mu}\Big) \|c\|_{L^1(\Omega)}. \end{aligned}$$

This, along with Gronwall's lemma and the estimate (8.46), yields

$$\frac{1}{2}\int_{\Omega} |\nabla u|^2 dx + \frac{1}{4\mu}\int_{\Omega} c \, dx \le A_0.$$

Hence, the estimate (8.48) holds.

Finally, we prove the estimate (8.49). We derive from (8.10), (8.27), (8.48),

 $u \geq 0$ and Hölder's inequality that

$$\int_{\Omega} |\nabla v|^2 dx \le A_0 + A_0 T \int_0^T \int_{\Omega} |\nabla u|^2 dx ds \le A.$$

This completes the proof of Lemma 8.2. $\hfill\square$

Lemma 8.3. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that the assumption (8.45) holds, then there hold

$$||c||_{L^2(\Omega)} \le A.$$
 (8.51)

Proof. For any $s \ge 2$, we derive from (8.13), (8.16), (8.44), and the assumption (8.5) that

$$\frac{d}{dt} \int_{\Omega} c^{s} dx = s \int_{\Omega} c^{s-1} c_{t} dx$$

$$= s \int_{\Omega} c^{s-1} \left[\bigtriangleup c - \bigtriangledown \cdot (V_{1}(c) \bigtriangledown u) - \bigtriangledown \cdot (V_{2}(c) \bigtriangledown v) + \mu c(1-c-v) \right] dx$$

$$\leq -\frac{4(s-1)}{s} \int_{\Omega} |\bigtriangledown c^{s/2}|^{2} dx + \mu s \int_{\Omega} c^{s} dx$$

$$+A_{0}s(s-1) \int_{\Omega} c^{s-2} \left(|\bigtriangledown c| \cdot |\bigtriangledown u| + |\bigtriangledown c| \cdot |\bigtriangledown v| \right) dx.$$
(8.52)

Taking s = 2 in (8.52) and using Cauchy's inequality and the estimates

(8.48) and (8.49), we obtain

8.4 A Priori Estimates and Global Existence 215

$$\begin{split} \frac{d}{dt} \int_{\Omega} c^2 dx &\leq -2 \int_{\Omega} |\bigtriangledown c|^2 dx + 2\mu \int_{\Omega} c^2 dx \\ &+ A_0 \int_{\Omega} \left(|\bigtriangledown c| \cdot |\bigtriangledown u| + |\bigtriangledown c| \cdot |\bigtriangledown v| \right) dx \\ &\leq -2 \int_{\Omega} |\bigtriangledown c|^2 dx + 2\mu \int_{\Omega} c^2 dx \\ &+ 2\varepsilon \int_{\Omega} |\bigtriangledown c|^2 dx + A_0(\varepsilon) \int_{\Omega} \left(|\bigtriangledown u|^2 + |\bigtriangledown v|^2 \right) dx \\ &\leq A + 2\mu \int_{\Omega} c^2 dx. \end{split}$$

This, together with Gronwall's lemma, yields the estimate (8.51).

Up to now we have had the $L^2(\Omega)$ -estimate. To raise the a priori estimate to $L^s(\Omega)$ -estimate (s > 3), we need the following lemma [104, Lemma 1], which is an extension of Lemma 4.1 in [86].

Lemma 8.4. Consider the following linear parabolic problem

$$u_t - \Delta u + u = c \qquad \qquad in \ \Omega_T, \tag{8.53}$$

$$\frac{\partial u}{\partial \nu} = 0 \qquad \qquad on \ \partial \Omega_T, \qquad (8.54)$$

$$u(x,0) = u_0(x) \qquad \qquad \text{in } \Omega. \tag{8.55}$$

Assume that $u_0 \in W^1_{\infty}(\Omega)$ and that (u, c) satisfies (8.53)-(8.55). Moreover,

$$\| c \|_{L^{\rho}(\Omega)} \leq A_0$$

for all $t \in (0,T)$. Then for every $1 \le \rho < d$ (where d := the space dimension) we have

$$\| u(t) \|_{W_q^1(\Omega)} \le A_0(q),$$
 (8.56)

where

$$q < \frac{d\rho}{d-\rho}.\tag{8.57}$$

If $\rho = d$, then (8.56) is true with every $q < +\infty$ and if $\rho > d$, then (8.56) is true with $q = +\infty$.

Lemma 8.5. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that the assumption (8.45) holds, then there holds

$$||c||_{L^4(\Omega)} \le A.$$
 (8.58)

Proof. Going back to (8.52) and taking s = 4, we have

$$\frac{d}{dt} \int_{\Omega} c^4 dx \leq -3 \int_{\Omega} |\nabla c^2|^2 dx + 4\mu \int_{\Omega} c^4 dx + A_0 \int_{\Omega} c^2 (|\nabla c| \cdot |\nabla u| + |\nabla c| \cdot |\nabla v|) dx.$$
(8.59)

By the estimate (8.51) and Lemma 8.4, we have

$$\parallel u(t) \parallel_{W_q^1(\Omega)} \le A \qquad \text{for any } 1 < q < 6;$$

in particular,

$$\| \bigtriangledown u(t) \|_{L^5(\Omega)} \le A. \tag{8.60}$$

This, along with (8.27) and Hölder's inequality, yields

$$\int_{\Omega} |\nabla v|^5 dx \le A_0 + A_0 T^4 \int_0^T \int_{\Omega} |\nabla u|^5 dx dt \le A.$$
(8.61)

By Young's inequality and the estimate (8.60), we have that for any sufficiently small $\varepsilon > 0$,

$$\int_{\Omega} c^{2} |\nabla c| \cdot |\nabla u| dx$$

$$\leq \varepsilon \int_{\Omega} c^{2} |\nabla c|^{2} dx + A_{0}(\varepsilon) \int_{\Omega} c^{2} |\nabla u|^{2} dx$$

$$\leq \frac{\varepsilon}{4} \int_{\Omega} |\nabla c^{2}|^{2} dx + A_{0}(\varepsilon) \int_{\Omega} (c^{2})^{\frac{5}{3}} dx + A_{0}(\varepsilon) \int_{\Omega} (|\nabla u|^{2})^{\frac{5}{2}} dx$$

$$\leq \frac{\varepsilon}{4} \int_{\Omega} |\nabla c^{2}|^{2} dx + A_{0}(\varepsilon) \int_{\Omega} c^{\frac{10}{3}} dx + A(\varepsilon)$$

$$\leq \frac{\varepsilon}{4} \int_{\Omega} |\nabla c^{2}|^{2} dx + A_{0}(\varepsilon) \int_{\Omega} c^{4} dx + A(\varepsilon).$$
(8.62)

Similarly,

$$\int_{\Omega} c^2 |\nabla c| \cdot |\nabla v| dx \le \frac{\varepsilon}{4} \int_{\Omega} |\nabla c^2|^2 dx + A_0(\varepsilon) \int_{\Omega} c^4 dx + A(\varepsilon).$$
(8.63)

Inserting (8.62) and (8.63) into (8.59) and taking ε sufficiently small, we get

$$\frac{d}{dt} \int_{\Omega} c^4 dx \le A_0 \int_{\Omega} c^4 dx + A.$$

This, together with Gronwall's lemma, yields the estimate (??).

Lemma 8.6. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that the assumption (8.45) holds, then there hold

$$\|u\|_{W^1_{\infty}(\Omega)} \le A,\tag{8.64}$$

$$\| \bigtriangledown v \|_{L^{\infty}(\Omega)} \le A, \tag{8.65}$$

$$||c||_{L^p(\Omega)} \le A \qquad for \ any \ p > 5.$$
 (8.66)

Proof. By the estimate (??) and Lemma 8.4, we find that the estimate (8.64) holds. This, along with (8.27), yields the estimate (8.65).

We now turn to prove the estimate (8.66). Going back to (8.52), taking s = p > 5 and using (8.64), (8.65), and Young's inequality, we have

$$\begin{split} \frac{d}{dt} \int_{\Omega} c^{p} dx &\leq -\frac{4(p-1)}{p} \int_{\Omega} |\bigtriangledown c^{p/2}|^{2} dx + p\mu \int_{\Omega} c^{p} dx \\ &+ A(p) \int_{\Omega} c^{p-2} |\bigtriangledown c| dx \\ &\leq -\frac{4(p-1)}{p} \int_{\Omega} |\bigtriangledown c^{p/2}|^{2} dx + p\mu \int_{\Omega} c^{p} dx \\ &+ \varepsilon \int_{\Omega} |\bigtriangledown c^{p/2}|^{2} dx + A(p,\varepsilon) \int_{\Omega} c^{p-2} dx \\ &\leq A \int_{\Omega} c^{p} dx + A. \end{split}$$

So,

$$\frac{d}{dt} \int_{\Omega} c^p dx \le A \int_{\Omega} c^p dx + A.$$

This, along with Gronwall's lemma, yields the estimate (8.66).

Lemma 8.7. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that the assumption (8.45) holds, then, for any p > 5 there hold

$$\|u\|_{W_{p}^{2,1}(\Omega_{T})} \le A, \tag{8.67}$$

$$\| \bigtriangleup v \|_{L^p(\Omega)} \le A. \tag{8.68}$$

Proof. By (8.53)-(8.55), (8.66), and the parabolic L^p -theory, we get the estimate (8.67).

By (8.28), (8.44), (8.64), (8.67), and Hölder's inequality, we have

$$\int_{\Omega} |\Delta v|^p dx \le A + A_0 T^{p-1} \int_0^T \int_{\Omega} |\Delta u|^p dx dt \le A.$$

So, the estimate (8.68) holds. \Box

In the following we will establish a priori $W_p^{2,1}(\Omega_T)$ -estimate on c. We derive from (8.13), (8.16), and (8.17) that

8.4 A Priori Estimates and Global Existence 219

$$c_t - \triangle c + \mathbf{b} \cdot \bigtriangledown c = g \qquad \text{in } \Omega_T, \qquad (8.69)$$

$$\frac{\partial c}{\partial \mu} = 0 \qquad \qquad \text{on } \partial \Omega_T, \qquad (8.70)$$

$$c(x,0) = c_0(x) \qquad \qquad \text{in } \Omega, \tag{8.71}$$

where

$$\mathbf{b} := V_1'(c) \bigtriangledown u + V_2'(c) \bigtriangledown v, \tag{8.72}$$

$$g := \mu c (1 - c - v) - V_1(c) \bigtriangleup u - V_2(c) \bigtriangleup v.$$
(8.73)

To apply the L^p -theory to the problem (8.69)-(8.71), we need to prove $\|\mathbf{b}\|_{L^{\infty}(\Omega)} \leq A$. To this end, by (8.64) and (8.65), we need only $\|V'_1(c)\|_{L^{\infty}(\Omega)} \leq A$ and $\|V'_2(c)\|_{L^{\infty}(\Omega)} \leq A$, which in turn need to prove $\|c\|_{L^{\infty}(\Omega)} \leq A$.

In Chapter 7 we used the iterative technique of Alikakos [4] to establish the uniform-in-time boundedness of solutions. However, in the present chapter we cannot get the unifrom-in-time boundedness of c since the $L^2(\Omega)$ -bound of ∇v depends on the time T (see (8.49)). Unlike Chapter 7, this chapter will employ Horstmann and Winkler's method (see [48, 86]), along with the estimates (8.64)-(8.66) and the assumption (8.5), to establish a $L^{\infty}(\Omega)$ -bound of c, which depends on the time T.

Let p > 1 and define

$$B := -\bigtriangleup + I$$

with domain

$$D(B) := \Big\{ c \in W_p^2(\Omega) : \frac{\partial c}{\partial \nu} = 0 \quad \text{on } \partial \Omega \Big\}.$$

220 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion For each $\eta \ge 0$, define the sectorial operator B^{η} (see [78]) and

$$X_{\eta} := D(B^{\eta})$$
 with the norm $\|c\|_{X_{\eta}} := \|B^{\eta}c\|_{L^{p}(\Omega)}$.

Lemma 8.8. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that the assumption (8.45) holds, then there hold

$$\|c(t)\|_{X_{\eta}} \le A(t_0) \quad \text{for } 2\eta < 1 \text{ and } t \in [t_0, T) \ (0 < t_0 < T), \quad (8.74)$$

$$\|c(t)\|_{L^{\infty}(\Omega)} \le A \qquad for \ all \ t \in [0,T).$$

$$(8.75)$$

Proof. We first prove the estimate (8.74). By (8.13) and $c(x,0) = c_0(x)$, we have

$$c(t) = e^{-tB}c_0 + \int_0^t e^{-(t-\tau)B} \left[-\nabla \cdot (V_1(c) \nabla u) - \nabla \cdot (V_2(c) \nabla v) + (\mu+1)c - \mu c^2 - \mu cv \right] d\tau$$

and therefore

$$\|c(t)\|_{X_{\eta}} \leq \|e^{-tB}c_{0}\|_{X_{\eta}} + \int_{0}^{t} \|e^{-(t-\tau)B}\left[-\bigtriangledown \cdot (V_{1}(c)\bigtriangledown u) - \bigtriangledown \cdot (V_{2}(c)\bigtriangledown v) + (\mu+1)c - \mu c^{2} - \mu cv\right]\|_{X_{\eta}}d\tau.$$
(8.76)

By [78, Theorem 1.4.3] and (8.10)

$$\|e^{-tB}c_0\|_{X_{\eta}} \le A_0 t^{-\eta} e^{-\delta t} \|c_0\|_{L^p(\Omega)} \le A_0 t^{-\eta} e^{-\delta t}$$
(8.77)

and, by $0 \le v \le 1$ and (8.66),

$$\|e^{-(t-\tau)B}((\mu+1)c - \mu c^{2} - \mu cv)\|_{X_{\eta}}$$

$$\leq (t-\tau)^{-\eta}e^{-\delta(t-\tau)}\left((2\mu+1)\|c\|_{L^{p}(\Omega)} + \mu\|c^{2}\|_{L^{p}(\Omega)}\right)$$

$$\leq A(t-\tau)^{-\eta}e^{-\delta(t-\tau)}$$
(8.78)

8.4 A Priori Estimates and Global Existence 221

where $\delta \in (0, 1)$ and p > 5. By [86, Lemma 2.1], (8.5), (8.64), and (8.65), we have

$$\|e^{-(t-\tau)B}\left[-\bigtriangledown (V_{1}(c) \bigtriangledown u) - \bigtriangledown (V_{2}(c) \bigtriangledown v)\right]\|_{X_{\eta}}$$

$$\leq A_{0}\|e^{-(t-\tau)\Delta}\left[-\bigtriangledown (V_{1}(c) \bigtriangledown u) - \bigtriangledown (V_{2}(c) \bigtriangledown v)\right]\|_{X_{\eta}}$$

$$\leq A_{0}(\varepsilon)(t-\tau)^{-1/2-\eta-\varepsilon}e^{-\delta(t-\tau)}\|V_{1}(c) \bigtriangledown u - V_{2}(c) \bigtriangledown v\|_{L^{p}(\Omega)}$$

$$\leq A(t-\tau)^{-1/2-\eta-\varepsilon}e^{-\delta(t-\tau)} \qquad (8.79)$$

where $\varepsilon > 0$ such that $-1/2 - \eta - \varepsilon > -1$.

Inserting (8.77)-(8.79) into (8.76) and noting $1/2+\eta+\varepsilon<1$ and $\eta<1,$ we obtain

$$\begin{aligned} \|c(t)\|_{X_{\eta}} &\leq A_0 t^{-\eta} e^{-\delta t} + A(\varepsilon) \int_0^t \left[(t-\tau)^{-1/2-\eta-\varepsilon} e^{-\delta(t-\tau)} + (t-\tau)^{-\eta} e^{-\delta(t-\tau)} \right] d\tau \\ &\leq A(t_0) \end{aligned}$$

for all $t \in [t_0, T)$ $(0 < t_0 < T)$. Hence, the estimate (8.74) is proved.

We are now in a position to prove the estimate (8.75). Note p > 5 and let $2\eta \in (\frac{3}{p}, 1)$. Since $2\eta > d/p$ (d := the space dimension), by [78, Theorem 1.6.1] we have that

$$X_{\eta} \hookrightarrow C(\overline{\Omega}).$$

Hence, by (8.74) we have that

$$||c(t)||_{L^{\infty}(\Omega)} \le A(t_0) \quad \text{for } t > t_0 > 0.$$

Furthermore, the local existence Theorem 8.1 yields that there exists some $t_0 \in (0, 1)$ such that

$$||c(t)||_{L^{\infty}(\Omega)} \le A_0 \qquad \text{for } t \le t_0.$$

Therefore,

$$||c(t)||_{L^{\infty}(\Omega)} \le A \qquad \text{for all } t \in [0, T).$$

This completes the proof of the estimate (8.75). \Box

Lemma 8.9. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that the assumption (8.45) holds, then, for any p > 5 there hold

$$\|c\|_{W_{p}^{2,1}(\Omega_{T})} \le A. \tag{8.80}$$

Proof. Returning to (8.69)-(8.73), noting (8.4) and (8.44), and using Lemmas 8.6 - 8.8, we have

$$\|\mathbf{b}\|_{L^{\infty}(\Omega)} \le A,\tag{8.81}$$

$$\|g\|_{L^p(\Omega_T)} \le A. \tag{8.82}$$

These, along with (8.10) and the parabolic L^p -theory, yields the estimate (8.80). \Box

Lemma 8.10. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that the assumption (8.45) holds, then there hold

$$\|u\|_{C^{2+\sigma,1+\sigma/2}(\Omega_T)} \le A,\tag{8.83}$$

$$\|v\|_{C^{2+\sigma,1+\sigma/2}(\Omega_T)} \le A, \tag{8.84}$$

$$\|c\|_{C^{2+\sigma,1+\sigma/2}(\Omega_T)} \le A.$$
(8.85)

Proof. By (8.80) and the Sobolev embedding theorem (see [108], taking p sufficiently large),

8.4 A Priori Estimates and Global Existence 223 $\|c\|_{C^{\sigma,\sigma/2}(\varOmega_T)} \leq A.$

This, together with (8.53)-(8.55) and the parabolic Schauder theory, yields (8.83).

Moreover, by (8.10), (8.21), (8.27), (8.28), and (8.83), we get the estimate (8.84).

We next turn to prove the estimate (8.85). Returning to (8.69)-(8.73) and noting (8.4), (8.83), and (8.84), we have

$$\|\mathbf{b}\|_{C^{\sigma,\sigma/2}(\Omega_T)} \le A, \qquad \|g\|_{C^{\sigma,\sigma/2}(\Omega_T)} \le A.$$

Hence, by the Schauder theory again, we obtain the estimate (8.85).

With a priori estimates (8.83)-(8.85), we can extend the local classical solution established in Theorem 8.1 to all t > 0, as done in Chapters 6 and 7. Namely, we have

Theorem 8.2. In addition to the assumption (8.4), (8.5), and (8.10), we assume that

 $\mu > 0.$

Then, there exists a unique solution $(c, v, u) \in (C^{2+\sigma, 1+\sigma/2}(\Omega_T))^3$ of the system (8.13)-(8.17) for any given T > 0. Furthermore

$$c \ge 0, \quad 0 \le v \le 1, \quad u \ge 0.$$

Remark 8.1. The uniform-in-time boundedness of c remains open due to the bound of $\| \bigtriangledown v \|_{L^2(\Omega)}$ depends on the time T (see the estimate (8.49) and its 224 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion proof). Moreover, our global existence result strongly depends on the presence of the logistic damping (i.e. $\mu > 0$). In other word, the global existence remains open for $\mu = 0$.

8.5 Boundedness for A Volume-Filling Model

Assuming that cancer cells carry a nonzero finite volume and that occupation of an area limits other cells from penetrating it, typical density-dependent chemotactic and haptotactic sensitivity functions reads as follows (see[70, 81]): for i = 1, 2,

$$V_i(c) = \chi_i c \left(1 - \frac{c}{\gamma}\right), \quad \text{where } \gamma \ge 1 \text{ denotes the maximum cell density.}$$

$$(8.86)$$

In (8.86), χ_1 and χ_2 are assumed to be two positive constants. Clearly, $V_i(c) \rightarrow \chi_i c$ as $\gamma \rightarrow +\infty$. For this choice of $V_1(c)$ and $V_2(c)$, the assumption (8.4) holds, but the assumption (8.5) is not satisfied (since $V_1(c), V_2(c) \rightarrow -\infty$ as $c \rightarrow +\infty$). However, these specific forms of $V_1(c)$ and $V_2(c)$ are in favor of the proofs of global existence and boundedness. In fact, we have

Theorem 8.3. In addition to the assumption (8.10) and (8.86), we assume that

$$0 \le c_0(x) \le \gamma. \tag{8.87}$$

Then, there exists a unique solution $(c, v, u) \in \left(C^{2+\sigma, 1+\sigma/2}(\Omega_T)\right)^3$ of the system (8.13)-(8.17) for any given T > 0. Furthermore

8.5 Boundedness for A Volume-Filling Model 225

$$0 \le c \le \gamma, \tag{8.88}$$

$$0 \le v \le 1,\tag{8.89}$$

$$u \ge 0. \tag{8.90}$$

Proceeding as in the proof of Theorem 8.1, we can first establish the local existence and uniqueness of solutions. To extend the local solution to all t > 0, as done in Section ??8.4, we need to establish a priori $C^{2+\sigma, 1+\sigma/2}(\Omega_T)$ -estimate of (c, v, u), which strongly depends on a priori $L^{\infty}\Omega$)-estimate of c.

Lemma 8.11. Assume that $(c, v, u) \in C^{2,1}(\Omega_T)$ is a solution to (8.13)-(8.17) and that the assumptions (8.86) and (8.87) hold, then there holds

$$0 \le c \le \gamma. \tag{8.91}$$

Proof. By $V_1(0) = V_2(0) = 0$ and $c_0(x) \ge 0$, we easily find that $\underline{c} := 0$ is a sub-solution of the problem (8.69)-(8.71). On the other hand, by (8.87), $v \ge 0$, and $V_1(\gamma) = V_2(\gamma) = 0$, we easily find that $\overline{c} := \gamma$ is a sup-solution of the problem (8.69)-(8.71). Hence, by the maximum principle, we have the estimate (8.91). \Box

With the a priori estimate (8.91), we can prove Theorem 8.3 in the same way of the proof of Theorem 8.2.

References

- Adam, J.: A simplified mathematical model of tumor growth, Math. Biosci., 81, 224-229, 1986.
- Ainseba, B.E., Bendahmane, M., Noussair, A.: A reaction-diffusion system modeling predator-prey with prey-taxis, Nonlinear Anal. RWA, 9, 2086-2105, 2008.
- Albert, F.K., et al.: Early postoperative magnetic resonance imaging after resection of malignant glioma: objective evaluation of residual tumor and its influence on regrowth and prognosis, Neurosurgery, 34, 45-61, 1994.
- Alikakos, N.D.: L^p bounds of solutions of reaction-diffusion equations, Comm. Partial Differential Equations, 4, 827-868, 1979.
- Abbas AK, Lichtman AH. Cellular and Molecular Immunology. Philadelphia: Saunders; 2003.
- Bazaliy, B.V., Friedman, A.: A free boundary problem for an elliptic-parabolic system: Application to a model of tumor growth, Commun. Partial Diff. Eqns., 28, 517-560, 2003.
- Bellomo, N., Forni, G.: Looking for new paradigms towards a biological-mathematical theory of complex multicellular systems, Math. Mod. Meth. Appl. Sci., 16, 1001-1029, 2006.
- Bellomo, N., Delitala, M.: From the mathematical kinetic, and stochastic game theory to modelling mutations, onset, progression and immune competition of cancer cells, Physics of Life Reviews, 5, 183-206, 2008.
- Bellomo, N., Li, N.K., Maini, P.K.: On the foundations of cancer modelling: selected topics, speculations, and perspectives, Math. Mod. Meth. Appl. Sci., 18, 593-646, 2008.

- Bergelson, J.M., Cunningham, J.A., Droguett, G., et al.: Isolation of a common receptor for Coxsackie B viruses and adenoviruses 2 and 5, Science, 275, 1320-1323, 1997.
- Bertuzzi, A., Fasano, A., Gandolfi, A.: A mathematical model for tumor cords incorporating the flow of interstitial fluids, Math. Mod. Meth. Appl. Sci., 15, 1735-1778, 2005.
- Bhandari, M. S., Crook, J., Hussain, M.: Should intermittent androgen deprivation be used in routine clinical practice? J. Clin. Oncol., 23, 8212-8218, 2005.
- Biler, P., Hebisch, W., Nadzieja, T.: The debye system: Existence and large time behavior of solutions, Nonlinear Anal., 23, 1189-1209, 1994.
- Bischoff, J.R., et al.: An adenovirus mutant that replicates selectively in p53-deficient human tumor cells, Science, 274, 373-376, 1996.
- Blanchet, A., Dolbeault, J., Perthame, B.: Two-dimensional Keller-Segel model: optimal critical mass and qualitative properties of the solutions, Electron. J. Differential Equations, 44, 32pp, 2006 (electronic).
- Bruchovsky, N., et al. : Intermittent androgen suppression for prostate cancer: Canadian prospective trial and related observations, Mol. Urol., 4 (3), 191-199, 2000.
- Bruchovsky, N., Klotz, L., Crook, J., Malone, S., Ludgate, C., Morris, W.J., Gleave, M.E., Goldenberg, S. L.: Final results of the Canadian prospective phase II trial of intermitten androgen suppression for men in biochemical recurrence after radiotherapy for locally advanced prostate cancer, Cancer, 107, 389-395, 2006.

- 228 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
- Bruchovsky, N., Klotz, L., Crook, J., Larry, S., Goldenberg, S.L.: Locally advanced prostate cancer – biochemical results from a prospective phase II study of intermitten androgen suppression for men with evidence of psa relapse after radiotherapy, Cancer, 109, 858-867, 2007.
- Byrne, H.M., Chaplain, M.A.J.: Growth of nonnecrotic tumours in the presence and absence of inhibitors, Math. Biosci., 130, 130-151, 1995.
- Byrne, H.M., Chaplain, M.A.J.: Growth of necrotic tumors in the presence and absence of inhibitors, Math. Biosci., 135, 187-216, 1996a.
- Byrne, H.M., Chaplain, M.A.J.: Modelling the role of cell-cell adhesion in the growth and development of carcinoma, Math. Comput. Model, 12, 1-17, 1996b.
- Byrne, H.M.: A weakly nonlinear analysis of a model of vascular solid tumour growth, J. Math. Biol., 39, 59-89, 1999.
- Byrne, H.M., Alarcón, T.A., Murphy J., Maini, P.K.: Modelling the response of vascular tumours to chemotherapy: A multiscale approach, Math. Mod. Meth. Appl. Sci., 16, 1219-1241, 2006.
- Calvez, V., Carrillo, J.A.: Volume effects in the Keller-Segel model: energy estimates preventing blow-up, J. Math. Pures Appl., 86, 155-175, 2006.
- Chang, E.L., et al.: Hypofractionated radiotherapy for elderly or younger low-performance status glioblastoma patients: outcome and prognostic factors, Int. J. Radiat. Oncol. Biol. Phys., 56, 519-528, 2003.
- Chaplain, M.A.J.: Avascular growth, angiogenesis, and vascular growth in solid tumors: The mathematical modelling of the stages of tumor development, Math. Comput. Model, 23, 47-87, 1996.
- 27. Chaplain, M.A.J., Kuznetsov, V.A., James, Z.H., Stepanova, L.A.: Spatio-temporal dynamics of the immune system response to cancer, Proceedings

of the mathematical Models in Medical and Health Sciences Conference (eds. M. A. Horn, G. Simonett, G. Webb), Vanderbilt University Press, ISBN 0-8265-1310-7, 1998.

- Chaplain, M.A.J., Ganesh, M., Graham, I.G.: Spatio-temporal pattern formation on spherical surfaces: numerical simulation and application to solid tumour growth, J. Math. Biol., 42, 387-423, 2001.
- Chaplain, M.A.J., Anderson, A.R.A.: Mathematical modelling of tissue invasion, Cancer Modelling and Simulation (Ed. L. Preziosi), Chapman & Hall/CRT, 267-297, 2003.
- Chaplain, M.A.J., Lolas, G.: Mathematical modelling of cancer invasion of tissue: dynamic heterogeneity, Net. Hetero. Med., 1, 399-439, 2006.
- Chaplain, M.A.J., Lachowicz, M., Szymańska, Z., Wrzosek, D. Mathematical modelling of cancer invasion: the importance of cell-cell adhension and cell-matrix adhension, preprint, 2010.
- Chen, Y.: Parabolic Partial Differential Equations of Second Order, Beijing University Press, Beijing, 2003, (in Chinese).
- 33. Chiocca, E.A.: Oncolytic viruses, Nat. Rev., Cancer, 2, 938-950, 2002.
- 34. Chiocca, E.A., Abbed, K.M., Tatter, S., et al.: A phase I open-label, dose-escalation, multi-institutional trial of injection with an E1B-attenuated adenovirus, ONYX-015, into the peritumoral region of recurrent malignant gliomas, in the adjuvant setting, Mol. Ther., 10, 958-966, 2004.
- 35. Cieślak, T., Morales-Rodrigo, C.: Quasilinear nonlinear non-uniformly parabolic-elliptic system modelling chemotaxis with volume filling effect: Existence and uniqueness of global-in-time solutions, Topological Methods in Nonlinear Analysis, 29, 361-382, 2007.

- 230 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
- Cieślak, T., Winkler, M.: Finite-time blow-up in a quasilinear system of chemotaxis, Nonlinearity, 21, 1057-1076, 2008.
- Cieślak, T., Laurençot, P.: Finite time blow-up for radially symmetric solutions to a critical quasilinear Smoluchowski-Poisson system, C. R. Acad. Sci. Paris, Ser. I, 347, 237-242, 2009.
- Coffey, M.C., Strong, J.E., Forsyth, P.A., Lee, P.W.K.: Reovirus therapy of tumors with activated Ras pathways, Science, 282, 1332-1334, 1998.
- Corrias, L., Perthame, B., Zaag, H.: Global solutions of some chemotaxis and angiogenesis systems in high space dimensions, Milan J. Math., 72, 1-28, 2004.
- 40. Corrias, L., Perthame, B., Zaag, H.: L^p and L[∞] a priori estimates for some chemotaxis models and applications to the Cauchy problem (Dynamics of spatio-temporal patterns for the system of reaction-diffusion equations), Kyoto University, RIMS **1416**, 105-119, 2005.
- Crampin, E.J., Mani, P.K.: Modelling biological pattern formation: the role of domain growth, Comments on Theoretical Biology, 3, 229-249, 2001.
- Critini, V., Lowengrub, J.S., Nie, Q.: Nonlinear simulation of tumor growth, J. Math. Biol., 46, 191-224, 2003.
- Cui, S., Friedman, A.: A free boundary problem for a singular system of differential equations: An application to a model of tumor growth, Trans. Amer. Math. Soc., 355, 3537-3590, 2003.
- Cui, S.: Global existence of solutions for a free boundary problem modeling the growth of necrotic tumors, Interfaces and Free Boundaries, 7, 147-159, 2005.
- Cui, S., Escher, J.: Well-posedness and stability of a multi-dimensional tumor growth model, Arch. Ration. Mech. Anal., 191, 173-193, 2009.

- De Angelis, E., Preziosi, L.: Advection diffusion models for solid tumours in vivo and related free-boundary problems, Math. Mod. Meth. Appl. Sci., 10, 379-408, 2000.
- De Angelis, E., Jabin, P.E.: Qualitative analysis of a mean field model of tumor-immune system competition, Math. Mod. Meth. Appl. Sci., 13, 187-206, 2003.
- Delgado, M., Gayte, I., Morales-Rodrigo, C., Suárez, A.: An angiogenesis model with nonlinear chemotactic response and flux at the tumor boundary, Nonlinear Anal., 72, 330-347, 2010.
- Dingli, D., et al.: Dynamic iodide trapping by tumor cells expressing the thyroidal sodium iodide symporter, Biochem. Biophys. Res. Commun., 325, 157-166, 2004a.
- Dingli, D., et al.: Image-guided radiovirotherapy for multiple myeloma using a recombinant measles virus expressing the thyroidal sodium iodide symporter, Blood, 103, 1641-1646, 2004b.
- Dingli, D., et al.: Mathematical modeling of cancer radiovirotherapy, Math. Biosci., 199, 80-103, 2006.
- Ellis, W.J., et al.: Characterization of a novel androgen-sensitive, prostate-specific antigen-producing prostatic carcinoma xenograft: LuCap 23¹, Clin. Cancer Res., 2, 1039-1048, 1996.
- Ferlay, J., et al: GLOBOCAN 2002: Cancer Incidense, Mortality and prevalence Worldwide, IARC CnacerBase No. 5, Version 2.0. IARCPress, Lyon, 2002.
- Floyd, N.S., Woo, S.Y., Teh, B.S.: Hypofractionated intensity-modulated radiotherapy for primary glioblastoma multiforme, Int. J. Radiat. Oncol., 58, 721-726, 2004.

- 232 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
- Fontelos, M.A., Friedman, A., Hu, B.: Mathematical analysis of a model for the iniation of angiogenesis, SIAM J. Math. Anal., 33, 1330-1355, 2002.
- 56. Freytag, S.O., et al.: A novel three-pronged approach to kill cancer cells selectively: concomitant viral, double suicide gene, and radiotherapy (see comments), Hum. Gene Ther., 9, 1323-1333, 1998.
- Friedman, A.: Partial Differential Equations, Holt, Rinehart & Winston, New York, 1969.
- Friedman, A., Reitich, F.: Analysis of a mathematical model for the growth of tumors, J. Math. Biol., 38, 262-284, 1999.
- Friedman, A., Reitich, F.: On the existence of spatially patterned dormant malignancies in a model for the growth of non-necrotic vascular tumor, Math. Mod. Meth. Appl. Sci., 77, 1-25, 2001.
- Friedman, A., Tello, J.I.: Stability of solutions of chemotaxis equations in reinforced random walks, J. Math. Anal. Appl., 272, 138-163, 2002.
- Friedman, A., Tao, Y.: Analysis of a model of a virus that selectively in tumor cells, J. Math. Biol., 47, 391-423, 2003.
- Friedman, A., Hu, B.: Bifurcation from stability to instability for a free boundary problem arising in a tumor model, Arch. Rat. Mech. Anal., 172, 267-294, 2004.
- Friedman, A., Lolas, G.: Analysis of a mathematical model of tumor lymphangiogenesis, Math. Mod. Meth. Appl. Sci., 1, 95-107, 2005.
- Friedman, A., Hu, B.: Asymptotic stability for a free boundary problem arising in a tumor model, J. Diff. Eqs., 227, 598-639, 2006.
- Friedman, A., Tian, J.P., Fulci, G., Chiocca, E.A., Wang, J.: Glioma virotherapy: Effects of innate immune suppression and increased viral replication capacity, Cancer Res, 66, 2314-2319, 2006.

- Friedman, A.: Mathematical analysis and challenges arising from models of tumour growth, Math. Mod. Meth. Appl. Sci., 17, 1751-1772, 2007.
- Fowler, J.F.: The phantom of tumor treatment-continually rapid proliferation unmasked, Radiother. Oncol., 22, 156-158, 1991.
- Fulci, G., et al.: Cyclophospharnide enhances glioma virotherapy by inhibiting innate immune response, Pro. Nat. Acad. Sci., 103, 12873-12878, 2006.
- Gatenby, R.A., Gawlinski, E.T.: A reaction-diffusion model of cancer invasion, Cancer Res., 56, 5745-5753, 1996.
- Gerisch, A., Chaplain, M.A.J.: Mathematical modelling of cancer cell invasion of tissue: Local and non-local models and the effect of adhesion, J. Theor. Biol., 250, 684-704, 2008.
- Gilbarg, D., Trudinger, N. S.: Elliptic Partial Differential Equations of Second Order, Springer-Verlag Berlin Heidelberg, 2001.
- 72. Gorlia T, van den bent M.J., Hegi M.E. et al: Nomogram for predicting survival of patients with newly diagnosed glioblastoma: prognostic factor analysis of EORTC and NCIC trial 26981/CE.3 http://Oncology.Lancet.com, 9:29-38
- Greenspan, H.: Models for the growth of a solid tumor by diffusion, Stud. Appl. Math., 52, 317-340, 1972.
- Greenspan, H.: On the growth and stability of cell cultures and solid tumors, J. Theor. Biol., 56, 229-235, 1976.
- Guo, B.: Viscosity Methods and the Viscosity of Difference Schemes, Science Press, Beijing, 1993, (in Chinese).
- 76. Guo, Q., Tao, Y., Aihara, K.: Mathematical modelling of prostate tumor growth under intermittent androgen suppression with partial differential equations, Internat. J. Bifur. Chaos, 18, 3789-3797, 2008.

- 234 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
- 77. Heise, C., Sampson-Johannes, A., Williams, A., McCormick, F., Von Hoff, D.D., Kirn, D.H.: ONYX-015, an E1B gene-attenuated adenovirus, causes tumor-specific cytolysis and antitumoral efficacy that can be augmented by standard chemotherapeutic agents, Nature Med., 3, 639-645, 1997.
- Henry, D.: Geometric Theorey of Semilinear Parabolic Equations, Lecture Notes in Mathematics, 840, Berlin: Springer, 1981.
- Herrero, M.A., Velázquez, J.L.L.: A blow-up mechanism for a chemotaxis model, Ann. Sc. Norm. Super. Pisa Cl. Sci., 24(4), 633-683, 1997.
- Hillen, T., Painter, K.J.: Global existence for a parabolic chemotaxis model with prevention of overcrowding, Adv. Appl. Math., 26, 280-301, 2001.
- Hillen, T., Painter, K.J.: A user's guide to PDE models for chemotaxis, J. Math. Biol., 58, 183-217, 2009.
- Hobbie, R.K.: Intermediate Physics for Medicine and Biology, Springer, New York, 1997.
- Holland, J.F., Frei, E.: Cancer Medicine, 5th edition, Harcourt asia Pte Ltd., 2001.
- Horstmann, D., Wang, G.: Blow-up in a chemotaxis model without symmetry assumptions, Eur. J. Appl. Math., 12, 159-177, 2001.
- Horstmann, D.: On the existence of radially symmetric blow-up solutions for the Keller-Segel model, J. Math. Biol., 44, 463-478, 2002.
- 86. Horstmann, D., Winkler, M.: Boundedness vs. blow-up in a chemotaxis system,
 J. Diff. Eqns., 215, 52-107, 2005.
- Hsing, A., Tsao, L., Devesa, S.: International trends and patterns of prostate cancer incidence and mortality, International Journal of Cancer (Pred. Oncol), 85, 60-67, 2000.

- Huggins, C., Hodges, C.: Studies of prostate cancer: I. the effects of castration, oestrogen and androgen injections on serum phosphates in metastatic carcinoma of the prostate, Cancer Res., 1, 293-207, 1941.
- Ideta, A., Tanaka, G., Takeuchi, T., Aihara, K.: A mathematical model of intermitten androgen suppression for prostate cancer, J. Nonlinear Sci., 18, 593-614, 2008.
- Igawa, T., Lin, F.F., Lee, M.S., Karan, D., Batra, S.K., Lin, M.F.: Establishment and characterization of androgen-independent human prostate cancer LNCaP cell model, Prostate, 50, 222-235, 2002.
- Isaacs, J.T., Coffey, D.S.: Adaptation versus selection as mechanism response for the relapse of prostatic cancer to androgen therapy as studied in the dunning *R*-3327-H adenocarcinoma, Cancer Res., 41, 5070-5075, 1981.
- 92. Jain, R.: Barriers to drug delivery in solid tumors, Sci. Am., 271, 58-65, 1994.
- Jackson, T.L., Byrne, H.M.: A mathematical model to study the effects of drug resistance and vasculature on the response of solid tumors to chemotherapy, Math. Biosci., 164, 17-38, 2000.
- Jackson, T.L.: Vascular tumor growth and treatment: Consequence of polyclonality, competition and dynamic vascular support, J. Math. Biol., 44, 201-226, 2002.
- 95. Jackson, T.L.: A mathematical model of prostate tumor growth and androgen-independent relapse, Discrete Contin. Dyn. Syst. B, 4, 187-201, 2004a.
- 96. Jackson, T.L.: A mathematical investigation of multiple pathways to recurrent prostate cancer: comparion with experimental data, Neoplasia, 6, 697-704, 2004b.
- 97. Jäger, W., Luckhaus, S.: On explosions of solutions to a system of partial differential equations modelling chemotaxis, Trans. Amer. Math. Soc., **329**,

- 236 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
 819-824, 1992.
- Kamradt, J.M., Pienta, K.J.: Novel molecular targets for prostate cancer therapy, Semin. Oncol., 26, 234-243, 1999.
- Keller, E.F., Segel, L.A.: Initiation of slime mold aggregation viewed as an instaility, J. Theor. Biol., 26, 399-415, 1970.
- 100. Kokontis, J. M., Takakura, J., Hay, N., Liao, S.: Increased androgen receptor activity and altered c-myc expression in prostate cancer cells after long-term androgen deprivation, Cancer Res., 54, 1566-1573, 1994.
- 101. Kokontis, J. M., Hay, N., Liao, S.: Progression of LNCaP prostate tumor cells during androgen deprivation: Hormone-independent growth, repression of proliferation by androgen, and role for p27^{Kip1} in androgen-induced cell cycle arrest, Mol. Endocrinol., **12**, 941-953, 1998.
- Kokontis, J.M., Hsu, S., Chuu, C., Dang, M., Fukuchi, J., Hiipakka, R.A., Liao,
 S.: Role of androgen receptor in the progression of human prostate tumor cells to androgen independence and insensitivity, The Prostate, 65, 287-298, 2005.
- 103. Kowalczyk, R.: Preventing blow-up in a chemotaxis model, J. Math. Anal. Appl., 305, 566-588, 2005.
- 104. Kowalczyk, R., Szymańska, Z.: On the global existence of solutions to an aggregation model, J. Math. Anal. Appl., 343, 379-398, 2008.
- 105. Lachowicz, M.: Micro and meso scales of description corresponding to a model of tissue invasion by solid tumours, Math. Mod. Meth. Appl. Sci., 15, 1667-1683, 2005.
- 106. Lachowicz, M.: Towards microscopic and nonlocal models of tumour invasion of tissue, Selected Topics in Cancer Modeling (Eds. N. Bellomo, M. Chaplain, E. De Angelis), Birkhauser, Boston, 49-63, 2008.

- 107. Lacroix, M., et al.: A multivariate analysis of 416 patients with glioblastoma mltiforme: prognosis, extent of resection, and survival, J. Neurosurg., 95, 190-198, 2001.
- 108. Ladyzenskaja, O.A., Solonnikov, V.A., Ural'ceva, N. N.: Linear and Quasi-Linear Equations of Parabolic Type, Amer. Math. Soc. Transl., Vol. 23, American Mathematics Society, Providence, RI, 1968.
- 109. Lieberman, G.M.: Second Order Parabolic Differential Equations, World Scientific Publishing Co. Pte. Ltd., 1996.
- Liţcanu, G., Morales-Rodrigo, C.: Global solutions and asymptotic behavior for a parabolic degenerate coupled system arising from biology, Nonlinear Anal., 72, 77-98, 2010.
- 111. Liu, A.Y., Corey, E., Bladou, F., Lange, P.H., Vessella, R.L.: Prostatic cell lineage markers: emergence of BCL2⁺ cells of human prostate cancer xenograft LuCaP 23 following castration, Int. J. Cancer, 65, 85-89, 1996.
- 112. Lowengrub, J.S., Frieboes, H.B., Jin, F., Chuang, Y-L., Li, X., Macklin, P., Wise, S.M. Cristini, V.: Nonlinear modelling of cancer: bridging the gap between cells and tumours, Nonlinearity, 23, R1-R91, 2010.
- Macri, E., Loda, M.: Role of p27 in prostate carcinogenesis, Cancer metastasis Rev., 17, 337-344, 1998.
- Marciniak-Czochra, A., Ptashnyk, M.: Boundness of solutions of a haptotaxis model, Math. Mod. Meth. Appl. Sci., 20, 449-476, 2010.
- Nagai, T.: Blow-up of radially symmetric solutions to a chemotaxis system, Adv. Math. Sci. Appl., 5, 581–601, 1995.
- Nagai, T.: Blowup of nonradial solutions to parabolic-elliptic systems modeling chemotaxis in two-dimensional domains, J. Inequal. Appl., 6, 37–55, 2001.

- 238 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
- 117. Nemunaitis, J., et al.: Selective replication and oncolysis in p53 mutant tumors with ONYX-015, an E1B-55kD gene-deleted adenovirus, in patients with advanced head and neck cancer: a phase II trial, Cancer Res. 60, 6359-6366, 2000.
- Nemunaitis, J., et al.: Intravenous infusion of a replication-selective adenovirus (ONYX-015) in cancer patients: safety, feasibility and biological activity, Gene Therapy, 8, 746-759, 2001.
- 119. O'Donoghue, J.A., Bardies, M., Wheldon, T.E.: Relationships between tumor size and curability for uniformly targeted therapy with beta-emitting radionuclides, J. Nucl. Med., 36, 1902-1909, 1995.
- 120. Osaki, K., Tsujikawa, T., Yagi, A., Mimura, M.: Exponential attractor for a chemotaxis-growth system of equations, Nonlinear Anal., 51, 119-144, 2002.
- 121. O'Connell J, Bennett M, O'Sullivan G, Collins J, Shanahan F. The fas counterattack: cancer as a site of immune privilege. Immunol Today 1999;20:46-52.
- 122. O'Donoghue JA, Bardies M, Wheldon TE. Relationships between tumor size and curability for uniformly targeted therapy with beta-emitting radionuclides. J Nucl Med 1995;36:1902-9.
- 123. Perthame, B.: Transport Equations in Biology, Switzerland, Birkhäuser Verlag, 2007.
- 124. Perumpanani, A.J., Byrne, H.M.: Extracellular matrix concentration exerts selection pressure on invasive cells, European Journal of cancer, 8, 1274-1280, 1999.
- 125. Pettet, G., Please, C.P., Tindall, M.J., McElwain, D.: The migration of cells in multicell tumor spheroids, Bull. Math. Biol., 63, 231-257, 2001.

- Rascle, M., Ziti, C.: finite time blow-up in some models of chemotaxis, J. Math. Biol., 33, 388-414, 1995.
- 127. Rew, D.A., Wilso, G.D.: Cell production rate in human tissues and tumours and their significance,. Part II: clinical data, Eur. J. Surg. Oncol., 26, 405-417, 2000.
- 128. Rodriguez, R., et al.: Prostate attenuated replication competent adenovirus (ARCA) CN706: a selective cytotoxic for prostate-specific antigen-positive prostate cancer cells, Cancer Res., 57, 2559-2563, 1997.
- 129. Rogulski, K.R., et al.: In vivo antitumor activity of ONYX-015 is influenced by p53 status and is augmented by radiotherapy, Cancer Res., 60, 1193-1196, 2000.
- Routes, J.M., et al.: Adenovirus E1A oncogene expression in tumor cells enhances killing by TNF-related apoptosis-inducing ligand (TRAIL), J. Immunol. 165, 4522-4527, 2000.
- 131. Savageau, M.A.: Allometric morphogenesis of complex systems: a derivation of the basic equations from the first principles, Proc. Natl. Acad. Sci., 76, 6023-6025, 1979.
- 132. Shah, R.B., et al.: Androgen-independendent prostate cancer is a heterogeneous group of diseases: lessons from a rapid autopsy program, Cancer Res., 64, 9209-9216, 2004.
- 133. Sherrat, J., Chaplain, M.: A new mathematical model for avascular tumor growth, J. Math. Biol., 43, 291-312, 2001.
- 134. Shimada, T., Aihara, K.: A nonlinear model with competition between tumor cells and its application to intermittent androgen suppression therapy of prostate cancer, Math. Biosci., 214, 134-139, 2008.

- 240 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
- 135. Simon, C., Everitt, H., Birtwistle, J., Stevenson, B. Oxford Handbook of General Practice, Oxford: Oxford University Press, 2002.
- 136. Stummer, W., et al.: Extend of resection and survival in glioblastoma multiforme: identification and adjustment for bias, Neurosurgey, 62, 1-11, 2008.
- 137. Stupp, S., et al.: Radiotherapy plus concomitant and adjuvant temozolomide for glioblastoma, N. Engl. J. Med., 352, 987-996, 2005.
- 138. Sugiyama, Y.: Time global existence and asymptotic behavior of solutions to degenerate quasi-linear parabolic systems of chemotaxis, Differential and Integral Equations, 20, 133-180, 2007.
- 139. Sultanem, K., et al.: The use of hypofractionated intensity-modulated irradiation in the trearment of glioblastoma multiforme: preliminary results of a prospective trial, Int. J. Radiat. Oncol. Biol. Phys., 58, 247-252, 2004.
- 140. Swanson, K.R., True, L. D., Lin, D.W., Buhler, K.R., Vessella R., Murray, J. D.: A quantitative model for the dynamics of serum prostate-specific antigen as a marker for cancerous growth: An explanation for medical anomaly, Am. J. Pathol., 158, 2195-2199, 2001.
- 141. Swabb, E.A., Wei, J., Gullino, P.M.: Diffusion and convection in normal and neoplastic tissues, Cancer Res., 34, 2814-2822, 1974.
- 142. Szymańska, Z., Morales-Rodrigo, C., Lachowicz, M., Chaplain, M.: Mathematical modelling of cancer invasion of tissue: The role and effect of nonlocal interactions, Math. Mod. Meth. Appl. Sci., 19, 257-281, 2009.
- 143. Tao, Y., Yoshida, N., Guo, Q.: Nonlinear analysis of a model of vascular tumour growth and treatment, Nonlinearity, 17, 867-895, 2004.
- 144. Tao, Y., Guo, Q.: The competitive dynamics between tumor cells, a replication-competent virus and an immune response, J. Math. Biol., **51**, 37-74,
2005.

- 145. Tao, Y., Chen, M.: An elliptic-hyperbolic free boundary problem modelling cancer therapy, Nonlinearity, 19, 419-440, 2006.
- 146. Tao, Y., Guo, Q.: A free boundary problem modelling cancer radiovirotherapy, Math. Mod. Meth. Appl. Sci., 17, 1241-1259, 2007.
- 147. Tao, Y., Zhang, H.: A parabolic-hyperbolic free boundary problem modelling tumor treatment with virus, Math. Mod. Meth. Appl. Sci., **17**, 63-80, 2007.
- 148. Tao, Y., Guo, Q.: A mathematical model of combined therapies against cancer using viruses and inhibitors, Sciences in China Ser. A, 51, 2315-2329, 2008.
- 149. Tao, Y., Wang, M.: Global solution for a chemotactic-haptotactic model of cancer invasion, Nonlinearity, 21, 2221-2238, 2008.
- Tao, Y.: A free boundary problem modeling the cell cycle and cell movement in multicellular tumor spheroids, J. Differential Equations, 247, 49-68, 2009a.
- 151. Tao, Y.: Global existence of classical solutions to a combined chemotaxis-haptotaxis model with logistic source, J. Math. Anal. Appl., 354, 60-69, 2009b.
- 152. Tao, Y., Guo, Q., Aihara, K.: A model at the macroscopic scale of prostate tumor growth under intermittent androgen suppression, Math. Mod. Meth. Appl. Sci., 19, 2177-2201, 2009.
- 153. Tao, Y., Wang, M.: A combined chemotaxis-haptotaxis system: The role of logistic source, SIAM J. Math. Anal., 41, 1533-1558, 2009.
- 154. Tao, Y., Guo, Q., Aihara, K.: A mathematical model of prostate tumor growth under hormone therapy with mutation inhibitor, Journal of Nonlinear Sciences, 20, 219-240, 2010.

- 242 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
- 155. Tao, Y., Cui, C.: A density-dependent chemotaxis-haptotaxis system modeling cancer invasion, J. Math. Anal. Appl., 367, 612-624, 2010.
- 156. Tao, Y.: Global existence of classical solutions to a predator-prey model with nonlinear prey-taxis, Nonlinear Anal. RWA, 11, 2056-2064, 2010.
- 157. Tian, J.P.: The replicability of oncolytic virus: defining conditions on tumor virotherapy, Math. Biosci. and Eng., Vol. 8 (2011), no. 3, pp 841 - 860.
- 158. Tian, J.P., Y. Kuang, and H. Yang: Intracellular viral life-cycle induced rich dynamics in tumor virotherapy, accepted by The Canadian Applied Mathematics Quarterly, 2012.
- 159. Wang, Y., J.P. Tian, J. Wei: Lytic cycle: a defining process in oncolytic virotherapy, accepted by Applied Mathematical Modelling, Dec 2012.
- Tian, J.P., Friedman, A., Wang, J., Chiocca, E.A.: Modeling the effects of resection, radiation and chemotherapy in glioblastoma, J. Neurooncol., 91, 287-293, 2009.
- 161. Tello, J.I.: Mathematical analysis and stability of a chemotaxis model with logistic term, Math. Methods Appl. Sci., 27, 1865-1880, 2004.
- 162. Tello, J.I., Winkler, M.: A chemotaxis system with logistic source, Commun. PDE, **32**, 849-877, 2007.
- 163. Tindall, M., Please, C.: Modelling the cell cycle and cell movement in multicellular tumour spheroids, Bull. Math. Biol., 69, 1147-1165, 2007.
- 164. Vladar, H.P.de, Gonzalez, J.A.: Dynamic response of cancer under the influence of immunological activity and therapy, J. Theor. Biol., 227, 335-348, 2004.
- 165. Walker, C., Webb, G.F.: Global existence of classical solutions for a haptotaxis model, SIAM J. Math. Anal., 38, 1694-1713, 2007.

- 166. Ward, J.P., King, J.R.: Mathematical modelling of avascular tumor growth, IMA J. Math. Appl. Med. Biol., 14, 39-69, 1997.
- 167. Ward, J.P., King, J.R.: Mathematical modelling of avascular-tumor growth II: Modelling growth saturation, IMA J. Math. Appl. Med. Biol., 15, 1-42, 1998.
- 168. Ward, J. P., King, J.R.: Mathematical modelling of drug transport in tumour multicell spheroids and monolayer cultures, Math. Biosci., 181, 177-207, 2003.
- 169. Wei, X., Cui, S.: Global well-posedness for a drug transport model in tumor multicell spheroids, Math. Comp. Modelling, 45, 553-563, 2007.
- 170. Wein, L.M., Wu, J.T., Kirn, D.H.: Validation and analysis of a mathematical model of a replication-competent oncolytic virus for cancer treatment: implications for virus design and delivery, Cancer Res., 63, 1317-1324, 2003.
- 171. Werner-Wasik, M., et al.: Final report of a phaseI/II trial of hyperfractionated and accelerated hyperfractionated radiation therapy with camustine for adults with supratentorial malignant gliomas: radiation therapy oncology group study 83-02, Cancer, 77, 1535-1543, 1996.
- 172. Winkler, M.: Chemotaxis with logistic source: Very weak global solutions and their boundedness properties, J. Math. Anal. Appl., 348, 708-729, 2008.
- Winkler, M.: Aggregation vs. global diffusive behavior in the higher-dimensional Keller-Segel model, J. Differential Equations, 248, 2889-2905, 2010
- 174. Winkler, M.: Boundedness in the higher-dimensional parabolic-parabolic chemotaxis system with logistic source, Commun. Partial Differential Equations, 2010, to appear.
- 175. Wodarz, D.: Viruses as antitumor weapons: defining conditions for tumor remission, Cancer Res., 61, 3501-3507, 2001.

- 244 8 Density-Dependent Chemotaxis-Haptotaxis Model of Cancer Invasion
- 176. Wodarz, D.: Gene therapy for killing p53-negative cancer cells: use of replicating versus nonreplicating agents, Hum. Gene Ther., 14, 153-159, 2003.
- 177. Wrzosek, D.: Global attractor for a chemotaxis model with prevention of overcrowding, Nonlinear Anal. TMA, **59**, 1293-1310, 2004.
- 178. Wu, J.T., Byrne, H.M., Kirn, D.H., Wein, L. M.: Modeling and analysis of a virus that replicates selectively in tumor cells, Bull. Math. Biol., 63, 731-768, 2001.
- 179. J. T. Wu, D. H. Kirn and L. M. Wein, Analysis of a three-way race between tumor growth, a replication-competent virus and an immune response, Bull. Math. Biol., 66, 605-625, 2004.
- 180. Yang, D, J.P. Tian, J. Wang: A solvable hyperbolic free boundary problem modelling tumour regrowth, Applicable Analysis, 92, 1541-1558, 2013
- 181. Yoon, S.S., Carroll, N.M., Chiocca, E.A., Tanabe, K. K.: Cancer gene therapy using a replication-competent herpes simplex virus type I vector, Ann. Surg., 228, 366-374, 1998.
- 182. Yoshida, K., et al.: Expression of epidermal growth factor, transforming growth factor-α and their receptor genes in human carcinomas: implication for autocrine growth, Cancer Res., 81, 43-51, 1990.
- Zurakowski, R., Wodarz, D.: Model-driven approaches for in vitro combination therapy using ONYX-015 replicating oncolytic adenovirus, J. Theoret. Biol., 245, 1-8, 2007

Index

androgen deprivation therapy, 4

Total androgen blockage, 4